### SECURITIES AND EXCHANGE COMMISSION

## **FORM 10-Q**

Quarterly report pursuant to sections 13 or 15(d)

Filing Date: **2021-02-12** | Period of Report: **2020-12-31** SEC Accession No. 0001437749-21-002735

(HTML Version on secdatabase.com)

### **FILER**

### **CONSUMERS BANCORP INC /OH/**

CIK:1006830| IRS No.: 341771400 | State of Incorp.:OH | Fiscal Year End: 0630

Type: 10-Q | Act: 34 | File No.: 033-79130 | Film No.: 21625160

SIC: 6021 National commercial banks

Mailing Address 614 E LINCOLN WAY PO BOX 256 MINERVA OH 44657-2095 Business Address 614 E LINCOLN WAY PO BOX 256 MINERVA OH 44657-2096 3308687701

### UNITED STATES SECURITIES AND EXCHANGE COMMISSION WASHINGTON, D.C. 20549

### FORM 10-Q

Quar	terly Report Pursuant to Section 13 or 15 (d) of the Secur	rities Exchange Act of 1934
□ Trans	sition Report Under Section 13 or 15(d) of the Securities	Exchange Act of 1934
	For the quarterly period ended	d December 31, 2020
	Commission File No	. 033-79130
	CONSUMERS BA (Exact name of registrant as s	,
	OHIO (State or other jurisdiction of incorporation or organization)	34-1771400 (I.R.S. Employer Identification No.)
	t Lincoln Way, P.O. Box 256, Minerva, Ohio Address of principal executive offices)	44657 (Zip Code)
	(330) 868-7 (Registrant's telepho	
	Not applicate (Former name, former address and former fised)	
Exchange Act	of 1934 during the preceding 12 months (or for such short	s required to be filed by Section 13 or 15 (d) of the Securities ter period that the Registrant was required to file such reports),
and (2) has be	een subject to such filing requirements for the past 90 day	s. Yes ℤ No □
pursuant to Ru	<u> </u>	onically every Interactive Data File required to be submitted ag the preceding 12 months (or for such shorter period that the
registrant was	required to sublifit such mes).	Yes ℤ No □
reporting com	č č	filer, an accelerated filer, a non-accelerated filer, a smaller ions of "large accelerated filer," "accelerated filer," "smaller f the Exchange Act.
Large accelerat		Accelerated filer □ Smaller reporting company ☑ Emerging growth company □
_	g growth company, indicate by check mark if the regist th any new or revised financial accounting standards pro-	trant has elected not to use the extended transition period for vided pursuant to Section 13(a) of the Exchange Act. □
Indicate by ch	neck mark whether the registrant is a shell company (as do	efined in Rule 12b-2 of the Exchange Act).

Yes □ No 🗷

Securities registered pursuant to Section 12(b) of the Act:

Title of each class	Trading Symbol(s)	Name of each exchange on which registered
None		

There were 3,028,100 shares of Registrant's common stock, no par value, outstanding as of February 10, 2021.

### CONSUMERS BANCORP, INC. FORM 10-Q QUARTER ENDED December 31, 2020

### Table of Contents

	Page Number (s)
Part I – Financial Information	Ì
Item 1 – Financial Statements (Unaudited)	
Consolidated Balance Sheets at December 31, 2020 and June 30, 2020	1
Consolidated Statements of Income for the three and six months ended December 31, 2020 and 2019	2
Consolidated Statements of Comprehensive Income for the three and six months ended December 31, 2020 and 2019	3
Condensed Consolidated Statements of Changes in Shareholders' Equity for the three and six months ended December 31, 2020 and 2019	4
	_
Condensed Consolidated Statements of Cash Flows for the six months ended December 31, 2020 and 2019	5
Notes to the Consolidated Financial Statements	6-22
Notes to the Consolidated Financial Statements	0-22
Item 2 – Management's Discussion and Analysis of Financial Condition and Results of Operations	23-31
Item 3 – Not Applicable for Smaller Reporting Companies	
Item 4 – Controls and Procedures	32
Part II – Other Information	
Item 1 – Legal Proceedings	33
Itama 1 A Nat Applicable for Consiling Departure Communica	22
Item 1A – Not Applicable for Smaller Reporting Companies	33
Item 2 – Unregistered Sales of Equity Securities and Use of Proceeds	33
Item 3 – Defaults Upon Senior Securities	33
Item 4 – Mine Safety Disclosure	33
Itana 6 Odban Información	22
Item 5 – Other Information	33
Item 6 – Exhibits	33
Signatures	34

### **Item 1 – Financial Statements**

## CONSUMERS BANCORP, INC. CONSOLIDATED BALANCE SHEETS (Unaudited)

	De	cember 31,		June 30,
(Dollars in thousands, except per share data)		2020		2020
ASSETS			_	
Cash on hand and noninterest-bearing deposits in financial institutions	\$	8,996	\$	8,429
Federal funds sold and interest-bearing deposits in financial institutions		49		1,230
Total cash and cash equivalents		9,045		9,659
Certificates of deposit in other financial institutions		8,599		11,635
Securities, available-for-sale		145,665		143,918
Securities, held-to-maturity (fair value of \$3,605 at December 31, 2020 and \$3,868 at June 30, 2020)		3,336		3,541
Equity securities, at fair value		400		
Federal bank and other restricted stocks, at cost		2,472		2,472
Loans held for sale		4,770		3,507
Total loans		557,257		542,861
Less allowance for loan losses		(5,912)		(5,678)
Net loans		551,345		537,183
Cash surrender value of life insurance		9,573		9,442
Premises and equipment, net		14,972		14,901
Goodwill		836		836
Core deposit intangible, net		242		256
Accrued interest receivable and other assets		2,805		3,470
Total assets	\$	754,060	\$	740,820
LIABILITIES				
Deposits				
Noninterest-bearing demand	\$	194,180	\$	190,233
Interest bearing demand		104,777		99,173
Savings		241,854		228,567
Time		107,551		115,382
Total deposits		648,362		633,355
Short-term borrowings		13,275		6,943
Federal Home Loan Bank advances		18,083		31,161
Accrued interest and other liabilities		6,632		6,121
Total liabilities			_	
Commitments and contingent liabilities		686,352		677,580
SHAREHOLDERS' EQUITY				
Preferred stock (no par value, 350,000 shares authorized, none outstanding)		_		
Common stock (no par value, 8,500,000 shares authorized; 3,124,053 shares issued as of December 31, 2020 and June 30, 2020)		20,011		19,974
Retained earnings		44,492		40,460
Treasury stock, at cost (95,953 and 108,475 common shares as of December 31, 2020 and				
June 30, 2020, respectively)		(1,324)		(1,454)
Accumulated other comprehensive income		4,529		4,260
Total shareholders' equity		67,708		63,240
Total liabilities and shareholders' equity	\$	754,060	\$	740,820

$\sim$	•		1'1'1	C 1	
V-06	accompanying	notes to	concolidated	tinancial	ctatemente
-	e accompanying	HOLOS LO	Consonuated	HIHAHCIAI	statements.

## CONSUMERS BANCORP, INC. CONSOLIDATED STATEMENTS OF INCOME (Unaudited)

(Dollars in thousands, except per share amounts)		Three Months ended December 31, 2020 2019					Six Months ended December 31, 2020 2019			
(Dollars in thousands, except per share amounts)		2020		)19		2020		2019		
Interest and dividend income										
Loans, including fees	\$	6,583	\$	4,862	\$	13,072	\$	9,623		
Securities, taxable		344		480		716		990		
Securities, tax-exempt		427		400		847		799		
Federal bank and other restricted stocks		21		20		39		40		
Federal funds sold and other interest-bearing deposits		42		16		89		42		
Total interest and dividend income		7,417		5,778		14,763		11,494		
Interest expense				-						
Deposits		487		910		1,064		1,855		
Short-term borrowings		2		13		6		24		
Federal Home Loan Bank advances		70		59		141		138		
Total interest expense		559		982		1,211		2,017		
Net interest income		6,858		4,796		13,552		9,477		
Provision for loan losses		130		185		260		315		
Net interest income after provision for loan losses		6,728		4,611		13,292		9,162		
•		,		,		ĺ		Í		
Noninterest income										
Service charges on deposit accounts		314		360		621		733		
Debit card interchange income		445		384		901		775		
Gain on sale of mortgage loans		246		142		482		276		
Bank owned life insurance death benefit		_		_		_		324		
Bank owned life insurance income		65		66		131		134		
Securities gains, net		8		4		8		110		
Other		75		69		151		142		
Total noninterest income		1,153		1,025		2,294		2,494		
Noninterest expenses										
Salaries and employee benefits		2,760		2,207		5,451		4,380		
Occupancy and equipment		636		595		1,276		1,127		
Data processing expenses		176		141		362		526		
Debit card processing expenses		216		194		454		395		
Professional and director fees		249		133		486		390		
FDIC assessments		87		5		148		(2)		
Franchise taxes		108		95		217		190		
Marketing and advertising		116		93		250		274		
Telephone and network communications		79		70		163		144		
Amortization of intangible		7		_		14		_		
Other		397		402		809		816		
Total noninterest expenses		4,831		3,935		9,630		8,240		
Income before income taxes		3,050		1,701		5,956		3,416		
Income tax expense		543		261		1,048		473		
Net income	\$	2,507	\$	1,440	\$	4,908	\$	2,943		
		<u> </u>				<u> </u>	_			
Basic and diluted earnings per share	\$	0.83	\$	0.53	\$	1.63	\$	1.07		

See accompanying notes to consolidated financial statements.

# CONSUMERS BANCORP, INC. CONSOLIDATED STATEMENTS OF COMPREHENSIVE INCOME (Unaudited)

(Dollars in thousands)	Three Months ended December 31,					Six Mont Decem	 	
		2020 2019				2020	 2019	
Net income	\$	2,507	\$	1,440	\$	4,908	\$ 2,943	
Other comprehensive income, net of tax:								
Net change in unrealized gains (losses) on securities available-for-								
sale:								
Unrealized gains (losses) arising during the period		278		(28)		348	790	
Reclassification adjustment for gains included in income		(8)		(4)		(8)	(110)	
Net unrealized gains (losses)		270		(32)		340	680	
Income tax effect		(57)		6		(71)	(144)	
Other comprehensive income (loss)		213		(26)		269	536	
						_		
Total comprehensive income	\$	2,720	\$	1,414	\$	5,177	\$ 3,479	
See accompanying notes to cons	ماناه	ted financia	Letat	ements				

See accompanying notes to consolidated financial statements.

# CONSUMERS BANCORP, INC. CONDENSED CONSOLIDATED STATEMENTS OF CHANGES IN SHAREHOLDERS' EQUITY (Unaudited)

Common

Stock

20,011

(Dollars in thousands, except per share data)

Balance, September 30, 2020

Retained

**Earnings** 

42,424

Treasury

Stock

(1,324) \$

Accumulated

Other

Comprehensive

Income (Loss)

4,316

Total

Shareholders'

**Equity** 

65,427

Net income	-		Ψ	2,507	-	(1,021)	•	1,0 1 0	Ψ	2,507
Other comprehensive income								213		213
Cash dividends declared (\$0.145 per share)				(439)						(439)
Balance, December 31, 2020	\$	20,011	\$	44,492	\$	(1,324)	\$	4,529	\$	67,708
,										
Balance, September 30, 2019	\$	14,697	\$	37,622	\$	(1,454)	\$	2,128	\$	52,993
Net income				1,440		, ,				1,440
Other comprehensive loss								(26)		(26)
Cash dividends declared (\$0.135 per share)				(371)				, , ,		(371)
Balance, December 31, 2019	\$	14,697	\$	38,691	\$	(1,454)	\$	2,102	\$	54,036
,					_				_	
							Ac	cumulated		75 4 1
(D.11 : 4 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	C	ommon	R	etained	T	reasury		Other	CI.	Total
(Dollars in thousands, except per share data)		Stock	$\mathbf{E}$	arnings		Stock	Cor	nprehensive	Sn	areholders'
							T	- OT \		Equity
							inc	come (Loss)		1 ,
Balance, June 30, 2020	\$	19,974	\$	40,460	\$	(1,454)	\$	4,260	\$	63,240
Balance, June 30, 2020 Net income	\$	19,974	\$	40,460 4,908	\$	(1,454)			\$	
	\$	19,974	\$	,	\$	(1,454)			\$	63,240
Net income	\$	·	\$	,	\$			4,260	\$	63,240 4,908 269
Net income Other comprehensive income	\$	19,974	\$	4,908	\$	(1,454)		4,260	\$	63,240 4,908 269 167
Net income Other comprehensive income 12,522 shares associated with vested and expired	\$	37		,	\$	130	\$	4,260	\$	63,240 4,908 269
Net income Other comprehensive income 12,522 shares associated with vested and expired stock awards	\$	·	\$ \$	4,908	\$		\$	4,260	\$	63,240 4,908 269 167
Net income Other comprehensive income 12,522 shares associated with vested and expired stock awards Cash dividends declared (\$0.29 per share)		37		4,908 (876)		130	\$	4,260		63,240 4,908 269 167 (876)
Net income Other comprehensive income 12,522 shares associated with vested and expired stock awards Cash dividends declared (\$0.29 per share)		37		4,908 (876)		130	\$	4,260		63,240 4,908 269 167 (876)
Net income Other comprehensive income 12,522 shares associated with vested and expired stock awards Cash dividends declared (\$0.29 per share)		37		4,908 (876)		130	\$	4,260		63,240 4,908 269 167 (876)
Net income Other comprehensive income 12,522 shares associated with vested and expired stock awards Cash dividends declared (\$0.29 per share) Balance, December 31, 2020	\$	20,011	\$	4,908 (876) 44,492	\$	130	\$	4,260 269 4,529	\$	63,240 4,908 269 167 (876) 67,708
Net income Other comprehensive income 12,522 shares associated with vested and expired stock awards Cash dividends declared (\$0.29 per share) Balance, December 31, 2020  Balance, June 30, 2019 Net income Other comprehensive income	\$	20,011 14,656	\$	4,908 (876) 44,492 36,487	\$	(1,324) (1,543)	\$	4,260 269 4,529	\$	63,240 4,908 269 167 (876) 67,708 51,166 2,943 536
Net income Other comprehensive income 12,522 shares associated with vested and expired stock awards Cash dividends declared (\$0.29 per share) Balance, December 31, 2020  Balance, June 30, 2019 Net income Other comprehensive income 11,813 shares associated with vested stock awards	\$	20,011	\$	4,908 (876) 44,492 36,487 2,943	\$	130	\$	4,260 269 4,529	\$	63,240 4,908 269 167 (876) 67,708 51,166 2,943 536 130
Net income Other comprehensive income 12,522 shares associated with vested and expired stock awards Cash dividends declared (\$0.29 per share) Balance, December 31, 2020  Balance, June 30, 2019 Net income Other comprehensive income	\$	20,011 14,656 41	\$ \$	4,908 (876) 44,492 36,487 2,943 (739)	\$	(1,324) (1,543) 89	\$	4,260 269 4,529 1,566 536	\$	63,240 4,908 269 167 (876) 67,708 51,166 2,943 536 130 (739)
Net income Other comprehensive income 12,522 shares associated with vested and expired stock awards Cash dividends declared (\$0.29 per share) Balance, December 31, 2020  Balance, June 30, 2019 Net income Other comprehensive income 11,813 shares associated with vested stock awards	\$	20,011 14,656	\$	4,908 (876) 44,492 36,487 2,943	\$	(1,324) (1,543)	\$	4,260 269 4,529	\$	63,240 4,908 269 167 (876) 67,708 51,166 2,943 536 130

See accompanying notes to consolidated financial statements.

# CONSUMERS BANCORP, INC. CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (Unaudited)

(Dollars in thousands)		Six Month Decemb		
	<u> </u>	2020		2019
Cash flows from operating activities				
Net cash from operating activities	\$	5,706	\$	2,214
Cash flow from investing activities				
Purchases of securities, available-for-sale		(24,975)		(6,678)
Maturities, calls and principal pay downs of securities, available-for-sale		20,304		11,902
Sale of securities, available-for-sale		2,733		6,110
Principal pay downs of securities, held-to-maturity		205		206
Purchase of equity securities		(400)		_
Net decrease in certificate of deposit in other financial institutions		3,036		989
Net increase in loans		(14,431)		(27,226)
Proceeds from BOLI death benefit		_		753
Premises and equipment purchases		(194)		(219)
Sale of other repossessed assets		17		<u> </u>
Net cash from investing activities		(13,705)		(14,163)
Cash flow from financing activities				
Net increase in deposit accounts		15,007		15,471
Net change in short-term borrowings		6,332		184
Proceeds from Federal Home Loan Bank advances		1,300		9,500
Repayments of Federal Home Loan Bank advances		(14,378)		(7,900)
Dividends paid		(876)		(739)
Net cash from financing activities		7,385		16,516
Increase (decrease) in cash or cash equivalents		(614)		4,567
Cash and cash equivalents, beginning of period		9,659		9,461
Cash and cash equivalents, end of period	\$	9,045	\$	14,028
Supplemental disclosure of cash flow information:				
Cash paid during the period:				
Interest	\$	1,230	\$	2,022
Federal income taxes	<u> </u>	1,055	4	350
Non-cash items:		1,000		
Transfer of loans to other repossessed assets		9		_
Issuance of treasury stock for stock awards		167		89
Right of use assets obtained in exchange for lease liabilities		_		582
<u> </u>				
See accompanying notes to consolidated financial st	tatements.			

(Dollars in thousands, except per share amounts)

### Note 1 – Summary of Significant Accounting Policies:

**Nature of Operations:** Consumers Bancorp, Inc. (the Corporation) is a bank holding company headquartered in Minerva, Ohio that provides, through its banking subsidiary, Consumers National Bank (the Bank), a broad array of products and services throughout its primary market area of Carroll, Columbiana, Jefferson, Stark, Summit, Wayne and contiguous counties in Ohio. The Bank's business involves attracting deposits from businesses and individual customers and using such deposits to originate commercial, mortgage and consumer loans in its primary market area.

Basis of Presentation: The consolidated financial statements for interim periods are unaudited and reflect all adjustments (consisting of only normal recurring adjustments), which, in the opinion of management, are necessary to present fairly the financial position and results of operations and cash flows for the periods presented. The unaudited financial statements are presented in accordance with the requirements of Form 10-Q and do not include all disclosures normally required by accounting principles generally accepted in the United States of America. The financial statements should be read in conjunction with the consolidated financial statements and notes thereto included in the Corporation's Form 10-K for the year ended June 30, 2020. The results of operations for the interim period disclosed herein are not necessarily indicative of the results that may be expected for a full year.

The consolidated financial statements include the accounts of the Corporation and the Bank. All significant inter-company transactions and accounts have been eliminated in consolidation.

**Segment Information:** The Corporation is a bank holding company engaged in the business of commercial and retail banking, which accounts for substantially all the revenues, operating income, and assets. Accordingly, all of the Corporation's operations are recorded in one segment, banking.

**Reclassifications:** Certain items in prior financial statements have been reclassified to conform to the current presentation. Any reclassifications had no impact on prior year net income or shareholders' equity.

Recently Issued Accounting Pronouncements Not Yet Effective: In June 2016, the Financial Accounting Standards Board (FASB) issued ASU 2016-13, Financial Instruments—Credit Losses (Topic 326): Measurement of Credit Losses on Financial Instruments. This ASU adds a new Topic 326 to the codification and removes the thresholds that companies apply to measure credit losses on financial instruments measured at amortized cost, such as loans, receivables, and held-to-maturity debt securities. Under current U.S. generally accepted accounting principles, companies generally recognize credit losses when it is probable that the loss has been incurred. The revised guidance will remove all current loss recognition thresholds and will require companies to recognize an allowance for credit losses for the difference between the amortized cost basis of a financial instrument and the amount of amortized cost that the corporation expects to collect over the instrument's contractual life. ASU 2016-13 also amends the credit loss measurement guidance for available-for-sale debt securities and beneficial interests in securitized financial assets. The guidance in ASU 2016-13 is effective for "public business entities," as defined in the guidance, that are SEC filers for fiscal years and for interim periods within those fiscal years beginning after December 15, 2018, including interim periods within those fiscal years. However, during July 2019, FASB unanimously voted for a proposal to delay this ASU to January 2023 for smaller reporting companies. On October 16, 2019, FASB approved a final ASU delaying the effective date. The new guidance is effective for annual and interim periods beginning after December 15, 2022 for certain entities, including smaller reporting companies. The Corporation is a smaller reporting company.

### CONSUMERS BANCORP, INC.

### Notes to the Consolidated Financial Statements (Unaudited) (continued)

(Dollars in thousands, except per share amounts)

In March 2020, the FASB issued ASU 2020-04, "Facilitation of the Effects of Reference Rate Reform on Financial Reporting". The ASU is intended to provide relief for companies preparing for discontinuation of interest rates based on LIBOR, or other reference rates that may be discontinued, and provides optional expedients and exceptions for applying GAAP to contract modifications and hedging relationships, subject to meeting certain criteria. The ASU also provides for a one-time sale and/or transfer to AFS or trading to be made for HTM debt securities that both reference an eligible reference rate and were classified as HTM before January 1, 2020. ASU 2020-04 is effective March 12, 2020 through December 31, 2022. The guidance requires companies to apply the guidance prospectively to contract modifications and hedging relationships while the one-time election to sell and/or transfer debt securities classified as HTM may be made any time after March 12, 2020. The Corporation does not expect ASU 2020-04 to have a material impact on its financial statements and disclosures.

### Note 2 – Acquisition

On December 29, 2020, the Bank entered into a Branch Purchase and Assumption Agreement (P&A Agreement) with CFBank National Association (CFBank) to acquire two branches of CFBank in Columbiana County, Ohio. The P&A Agreement provides for the sale and transfer by CFBank to the Bank the land, buildings and other associated assets of CFBank's drive-up branch location in Wellsville, Ohio and CFBank's branch location in Calcutta, Ohio (the Branches); approximately \$100 million in deposits attributable to the Branches; \$15 million in aggregate principal amount of subordinated debt securities issued by unrelated financial institutions; all performing loans attributable to the Branches which are outstanding at closing (totaling approximately \$3.1 million in aggregate principal amount as of November 30, 2020); and up to \$13.5 million in aggregate principal amount of single family residential mortgage loans and home equity lines of credit to be identified by the parties prior to the closing principally from CFBank's Northeast Ohio loan portfolio. In addition, CFBank will provide the opportunity for the Corporation to purchase at par at least \$15 million in aggregate principal amount of participation interests in commercial and commercial real estate loans originated by and held in CFBank's portfolio. In exchange, Consumers will pay to CFBank the net book value of the land, building and associated assets of the Branches, a deposit premium equal to 1.75% of the average daily deposits of the Branches for the 30 days preceding the closing, and the par value of the subordinated debt securities and loans acquired by Consumers.

The closing of the transactions contemplated by the P&A Agreement is subject to regulatory approval and satisfaction of other customary closing conditions. The parties expect the closing of the transactions to occur early in the third calendar quarter of 2021.

On June 14, 2019, the Corporation entered into an Agreement and Plan of Merger with Peoples Bancorp of Mt. Pleasant, Inc. (Peoples) and its wholly owned subsidiary, The Peoples National Bank of Mount Pleasant (Peoples Bank). On January 1, 2020, Consumers completed the acquisition by merger of Peoples in a stock and cash transaction for an aggregate consideration of approximately \$10,405. In connection with the acquisition, the Corporation issued 269,920 shares of common stock and paid \$5,128 in cash to the former shareholders of Peoples. Immediately following the merger, Peoples Bank, was merged into the Corporation's banking subsidiary, Consumers National Bank.

On December 31, 2019, Peoples had approximately \$72,016 in total assets, \$55,273 in loans and \$60,826 in deposits at its three banking centers located in Mt. Pleasant, Adena, and Dillonvale, Ohio. The assets and liabilities of Peoples were recorded on the Corporation's Balance Sheet at their estimated fair values as of January 1, 2020, the acquisition date, and Peoples' results of operations are included in the Corporation's Consolidated Statements of Income beginning on that date.

The following table summarizes the estimated fair values of the assets acquired and liabilities assumed at the date of acquisition of Peoples. The core deposit intangible will be amortized over ten years on a straight-line basis. Goodwill will not be amortized, but instead will be evaluated for impairment.

Consideration Paid	\$	10,405
Net assets acquired:		
Cash and cash equivalents	\$ 833	
Certificates of deposit in other financial institutions	11,839	
Securities, available-for-sale	4,051	
Federal bank and other restricted stocks, at cost	154	
Loans, net	55,320	

Premises and equipment	818
Core deposit intangible	270
Accrued interest receivable and other assets	140
Noninterest-bearing deposits	(11,979)
Interest-bearing deposits	(48,872)
Federal funds purchased	(2,348)
Federal Home Loan Bank advances	(491)
Other liabilities	(166)
Total net assets acquired	9,569
Goodwill	\$ 836

The acquired assets and liabilities were measured at estimated fair values. Management made certain estimates and exercised judgement in accounting for the acquisition. The fair value of loans was estimated using discounted contractual cash flows. The book balance of the loans at the time of the acquisition was \$55,273 before considering Peoples' allowance for loan losses, which was not carried over. The fair value disclosed above reflects a credit-related adjustment of \$(890) and an adjustment for other factors of \$937. Loans evidencing credit deterioration since origination, purchased credit impaired loans, included in loans receivable, were immaterial. Acquisition costs of \$827 pre-tax, or \$680 after-tax, were recorded during the twelve-month period ended June 30, 2020.

(Dollars in thousands, except per share amounts)

### Note 3 – Securities

Available –for-Sale	Aı	mortized Cost	l 	Gross Unrealized Gains	l 	Gross Unrealized Losses	_	Fair Value
December 31, 2020	¢	40	١٥	¢.	2	¢	ф	501
U.S. Treasury	\$	49		\$		\$ —	\$	501
Obligations of U.S. government-sponsored entities and agencies Obligations of state and political subdivisions		7,79 63,35		31 3,67		_		8,109
U.S. Government-sponsored mortgage-backed securities—residenti	o.1	56,75		1,49		(5)		67,034 58,248
U.S. Government-sponsored mortgage-backed securities—	11	30,73	) /	1,49	O	(5)		36,246
commercial		7,00	)4	7	5	(1)		7,078
U.S. Government-sponsored collateralized mortgage obligations—								
residential		4,52	20	17	5	_		4,695
Total available-for-sale securities	\$	139,93	32	\$ 5,73	9	\$ (6)	\$	145,665
Total available-for-sale securities	Ψ	137,73		ψ <i>3,13</i>	_	<u>ψ (0)</u>	Ψ	143,003
Held-to-Maturity		rtized ost	Ur	Gross recognized Gains	U	Gross nrecognized Losses		Fair Value
December 31, 2020							_	
Obligations of state and political subdivisions	\$	3,336	\$	269	\$	_	\$	3,605
Total held-to-maturity securities	\$	3,336	\$	269	\$	_	\$	
Available-for-Sale	A	mortized Cost	l	Gross Unrealized Gains	l	Gross Unrealized Losses		Fair Value
June 30, 2020								
U.S. Treasury	\$	1,24	18	\$	8	\$ —	\$	1,256
Obligations of U.S. government-sponsored entities and agencies		10,13	33	39	9	_		10,532
Obligations of state and political subdivisions		60,34	13	3,14	.9	_		63,492
U.S. government-sponsored mortgage-backed securities - residenti	al	48,64	15	1,51	5	(4)		50,156
U.S. government-sponsored mortgage-backed securities - commercial		8,44	14	5	5	(2)		8,497
U.S. government-sponsored collateralized mortgage obligations - residential		9,71	12	28	5	(12)		9,985
Total available-for-sale securities	\$	138,52	25	\$ 5,41	1	\$ (18)	\$	143,918
	Φ	130,32	Ě	ψ 3,11	=			
Held-to-Maturity	Amo	rtized ost	_	Gross nrecognized Gains	=	Gross nrecognized Losses	_	Fair Value
June 30, 2020	Amo Co	rtized ost	Ur	Gross nrecognized Gains	U	nrecognized		Value
June 30, 2020 Obligations of state and political subdivisions	Amo Co	rtized ost	Uı	Gross nrecognized Gains		nrecognized	\$	<b>Value</b> 3,868
June 30, 2020	Amo Co	rtized ost	Ur	Gross nrecognized Gains	U	nrecognized	\$ \$	<b>Value</b> 3,868

(Dollars in thousands, except per share amounts)

Proceeds from the sale and call of available-for-sale securities were as follows:

	Three Mor	nths E	Ended	Six Month	ns Ended
	Decem	iber 3	1,	Decemb	per 31,
	2020		2019	2020	2019
Proceeds from sales and calls	\$ 2,733	\$	1,650	2,733	6,110
Gross realized gains	31		4	31	110
Gross realized losses	23			23	_

The income tax provision related to the net realized gain amounted to \$2 for the three- and six-month periods ended December 31, 2020. The income tax provision related to the net realized gains amounted to \$1 and \$22 for the three- and six-month periods ended December 31, 2019, respectively.

The amortized cost and fair values of debt securities at December 31, 2020, by expected maturity, are shown below. Expected maturities will differ from contractual maturities because borrowers may have the right to call or prepay obligations with or without call or prepayment penalties. Securities not due at a single maturity date, primarily mortgage-backed securities, are shown separately.

	A	Est	timated Fair	
Available-for-Sale		Cost		Value
Due in one year or less	\$	4,930	\$	5,006
Due after one year through five years		14,015		14,545
Due after five years through ten years		13,700		14,327
Due after ten years		39,006		41,766
Total		71,651		75,644
U.S. Government-sponsored mortgage-backed and related securities		69,281		70,021
Total available-for-sale securities	\$	139,932	\$	145,665
Held-to-Maturity				
Due after one year through five years	\$	334	\$	356
Due after five years through ten years		667		735
Due after ten years		2,335		2,514
Total held-to-maturity securities	\$	3,336	\$	3,605

The following table summarizes the securities with unrealized losses at December 31, 2020 and June 30, 2020, aggregated by investment category and length of time that individual securities have been in a continuous unrealized loss position:

	Less than	12 Months	12 Month	s or more	To	otal
Available-for-sale	Fair Value			Unrealized Loss	Fair Value	Unrealized Loss
December 31, 2020						
U.S. Government-sponsored mortgage-backed securities – residential	1,198	(5)	_	_	1,198	(5)
U.S. Government-sponsored mortgage-backed securities-commercial	1,422	(1)	_	_	1,422	(1)
Total temporarily impaired	\$ 2,620	<u>\$ (6)</u>	<u> </u>	<u>\$</u>	\$ 2,620	\$ (6)
		9				

### CONSUMERS BANCORP, INC.

## Notes to the Consolidated Financial Statements (Unaudited) (continued)

(Dollars in thousands, except per share amounts)

	Less than	12 Months	12 Month	s or more	Total				
Available-for-sale	Fair Value	Unrealized Loss	Fair Value	Unrealized Loss	Fair Value	Unrealized Loss			
June 30, 2020									
U.S. Government-sponsored mortgage-backed securities – residential		_	625	(4)	625	(4)			
U.S. Government-sponsored mortgage-backed securities – commercial	1,806	(2)	_	_	1,806	(2)			
U.S. Government-sponsored collateralized mortgage obligations - residential	1,700	(12)	_	_	1,700	(12)			
Total temporarily impaired	\$ 3,506	\$ (14)	\$ 625	\$ (4)	\$ 4,131	\$ (18)			

Management evaluates securities for other-than-temporary impairment (OTTI) on a quarterly basis, and more frequently when economic or market conditions warrant such an evaluation. The securities portfolio is evaluated for OTTI by segregating the portfolio into two general segments and applying the appropriate OTTI model. Investment securities are generally evaluated for OTTI under FASB ASC Topic 320, Accounting for Certain Investments in Debt and Equity Securities.

In determining OTTI under the ASC Topic 320 model, management considers many factors, including: (1) the length of time and the extent to which the fair value has been less than cost, (2) the financial condition and near-term prospects of the issuer, (3) whether the market decline was affected by macroeconomic conditions, and (4) whether the entity has the intent to sell the debt security or more likely than not will be required to sell the debt security before its anticipated recovery. The assessment of whether an other-than-temporary decline exists involves a high degree of subjectivity and judgment and is based on the information available to management at a point in time.

The unrealized losses within the securities portfolio as of December 31, 2020 have not been recognized into income because the decline in fair value is not attributed to credit quality and management does not intend to sell, and it is not likely that management will be required to sell, the securities prior to their anticipated recovery. The decline in fair value within the securities portfolio is largely due to increases in mortgage backed and collateralized mortgage obligations prepayment speeds impacting the yield on bonds that were purchased at a premium. The mortgage-backed securities and collateralized mortgage obligations were primarily issued by Fannie Mae, Freddie Mac and Ginnie Mae, institutions which the government has affirmed its commitment to support. The Corporation does not own any private label mortgage-backed securities.

### Note 4 - Loans

Major classifications of loans were as follows:

	Dec	2020 cember 31,	une 30, 2020
Commercial	\$	158,911	\$ 158,667
Commercial real estate:			
Construction		3,802	16,235
Other		250,529	229,029
1 – 4 Family residential real estate:			
Owner occupied		93,726	90,494
Non-owner occupied		20,210	19,370
Construction		6,026	9,344
Consumer		24,230	 21,334

Subtotal	55	57,434	544,473
Net Deferred loan fees and costs		(177)	(1,612)
Allowance for loan losses	(	(5,912)	(5,678)
Net Loans	\$ 55	\$1,345	537,183

The commercial loan category in the above table includes PPP loans of \$52,539 as of December 31, 2020 and \$66,606 as of June 30, 2020 and a mortgage loan warehouse line of credit to another financial institution with an outstanding balance of \$47,268 as of December 31, 2020 and \$32,869 as of June 30, 2020. The outstanding balance of the warehouse line of credit can fluctuate significantly based on the other financial institution's funding needs.

(Dollars in thousands, except per share amounts)

The following table presents the activity in the allowance for loan losses by portfolio segment for the three months ended December 31, 2020:

	Com	Commercial		Commercial Real Estate		-4 Family Lesidential Real Estate	_(	Consumer	 Total
Allowance for loan losses:									
Beginning balance	\$	940	\$	3,694	\$	997	\$	136	\$ 5,767
Provision for loan losses		(82)		122		31		59	130
Loans charged-off				_				(12)	(12)
Recoveries		_		1		_		26	27
Total ending allowance balance	\$	858	\$	3,817	\$	1,028	\$	209	\$ 5,912

The following table presents the activity in the allowance for loan losses by portfolio segment for the six months ended December 31, 2020:

			С	ommercial	-4 Family desidential		
	Com	mercial		Real Estate	 Real Estate	 Consumer	 Total
Allowance for loan losses:							
Beginning balance	\$	947	\$	3,623	\$ 989	\$ 119	\$ 5,678
Provision for loan losses		(67)		192	39	96	260
Loans charged-off		(22)		_	_	(56)	(78)
Recoveries		_		2	_	50	52
Total ending allowance balance	\$	858	\$	3,817	\$ 1,028	\$ 209	\$ 5,912

The following table presents the activity in the allowance for loan losses by portfolio segment for the three months ended December 31, 2019:

	Com	nmercial	Co	ommercial Real Estate	-4 Family esidential Real Estate	 Consumer	Total
Allowance for loan losses:							
Beginning balance	\$	649	\$	2,645	\$ 550	\$ 65 \$	3,909
Provision for loan losses		117		6	64	(2)	185
Loans charged-off		_		_	_	(7)	(7)
Recoveries		_		1	1	6	8
Total ending allowance balance	\$	766	\$	2,652	\$ 615	\$ 62 \$	4,095

(Dollars in thousands, except per share amounts)

The following table presents the activity in the allowance for loan losses by portfolio segment for the six months ended December 31, 2019:

	Com	mercial	Co	ommercial Real Estate	-4 Family esidential Real Estate	 Consumer	 Total
Allowance for loan losses:							
Beginning balance	\$	660	\$	2,575	\$ 494	\$ 59	\$ 3,788
Provision for loan losses		106		75	120	14	315
Loans charged-off					_	(23)	(23)
Recoveries		_		2	1	12	15
Total ending allowance balance	\$	766	\$	2,652	\$ 615	\$ 62	\$ 4,095

The following table presents the balance in the allowance for loan losses and the recorded investment in loans by portfolio segment and based on impairment method as of December 31, 2020. Included in the recorded investment in loans is \$1,607 of accrued interest receivable.

			С	ommercial	4 Family esidential				
				Real	Real				
	Со	mmercial		Estate	Estate	C	onsumer		Total
Allowance for loan losses:									
Ending allowance balance attributable to loans:									
Individually evaluated for impairment	\$	3	\$	_	\$ 6	\$	_	\$	9
Acquired loans collectively evaluated for				101	84				185
impairment		_		101	04				103
Originated loans collectively evaluated for		855		3,716	938		209		5,718
impairment		055		3,710	 750		209		3,710
Total ending allowance balance	\$	858	\$	3,817	\$ 1,028	\$	209	\$	5,912
Recorded investment in loans:									
Loans individually evaluated for impairment	\$	148	\$	1,151	\$ 1,044	\$	_	\$	2,343
Acquired loans collectively evaluated for		661		7 924	22 220		9.025		40.640
impairment		001		7,824	23,220		8,935		40,640
Originated loans collectively evaluated for		158,331		245,489	96,733		15,328		515,881
impairment		130,331		243,409	90,733		13,326		313,861
Total ending loans balance	\$	159,140	\$	254,464	\$ 120,997	\$	24,263	\$	558,864
5								===	
		12							

(Dollars in thousands, except per share amounts)

The following table presents the balance in the allowance for loan losses and the recorded investment in loans by portfolio segment and based on impairment method as of June 30, 2020. Included in the recorded investment in loans is \$1,936 of accrued interest receivable.

Allowance for loan losses:	Со	ommercial	Сс	ommercial Real Estate		4 Family esidential Real Estate	C	onsumer_		Total
Ending allowance balance attributable to loans:	Φ.	• •	Φ.		Φ.		Φ.		Φ.	•
Individually evaluated for impairment	\$	28	\$	6	\$	_	\$	_	\$	34
Acquired loans collectively evaluated for impairment		_		103		94		_		197
Originated loans collectively evaluated for impairment		919		3,514		895		119		5,447
Total ending allowance balance	\$	947	\$	3,623	\$	989	\$	119	\$	5,678
Recorded investment in loans:										
Loans individually evaluated for impairment	\$	179	\$	1,045	\$	699	\$	_	\$	1,923
Acquired loans collectively evaluated for impairment		1,095		8,072		27,252		12,550		48,969
Originated loans collectively evaluated for impairment		156,054		236,840		92,168		8,843		493,905
Total ending loans balance	\$	157,328	\$	245,957	\$	120,119	\$	21,393	\$	544,797

The following table presents information related to unpaid principal balance, recorded investment and interest income associated with loans individually evaluated for impairment by class of loans as of December 31, 2020 and for the six months ended December 31, 2020:

		As of	mber 31,	0	Six Months ended December 31, 2020							
	U	Inpaid			A	Allowance for	A	verage	Interest		Ca	sh Basis
	Pr	incipal	Re	corded		Loan Losses	Re	corded		Income	I	nterest
	В	alance	Investment Allocated I		Investment		Recognized		Re	cognized		
With no related allowance recorded:												
Commercial real estate:												
Other	\$	1,286	\$	1,151	\$	_	\$	867	\$	4	\$	4
1-4 Family residential real estate:												
Owner occupied		947		797		_		629		11		11
Non-owner occupied		274		217		_		225		_		_
With an allowance recorded:												
Commercial		147		148		3		160		4		4
Commercial real estate:												
Other		_		_		_		205		6		6
1-4 Family residential real estate:												

Owner occupied	 30	 30	 6	 5	 	 
Total	\$ 2,684	\$ 2,343	\$ 9	\$ 2,091	\$ 25	\$ 25

13

(Dollars in thousands, except per share amounts)

The following table presents information related to average recorded investment and interest income associated with loans individually evaluated for impairment by class of loans for the three months ended December 31, 2020:

	Average			Interest	Cash Basis		
	Rec	orded		Income	Interest		
	Inve	Investment		ecognized	Re	cognized	
With no related allowance							
recorded:							
Commercial real estate:							
Other	\$	907	\$	3	\$	3	
1-4 Family residential real							
estate:							
Owner occupied		720		5		5	
Non-owner occupied		220				_	
With an allowance recorded:							
Commercial		150		2		2	
Commercial real estate:							
Other		204		3		3	
1-4 Family residential real							
estate:							
Owner occupied		10		_			
Total	\$	2,211	\$	13	\$	13	

The following table presents information related to unpaid principal balance, recorded investment and interest income associated with loans individually evaluated for impairment by class of loans as of June 30, 2020 and for the six months ended December 31, 2019:

		As	s of June	e 30, 202	20		Six Months ended December 31, 2019							
	Unpai	id			Al	lowance for	Averaş	ge	Interest		Cas	h Basis		
	Princip	oal	Reco	rded	Loa	n Losses	Record	led	Income		In	terest		
	Balan	ce	Inves	tment	Al	located	Investm	ent	Recognized		Rec	ognized		
With no related allowance recorded:														
Commercial real estate:														
Other	\$	922	\$	836	\$	_	\$	303	\$	87	\$	87		
1-4 Family residential real														
estate:														
Owner occupied		604		463		_		25		7		7		
Non-owner occupied		284		236		_		254		_		_		
With an allowance recorded:														
Commercial		176		179		28		167		5		5		
Commercial real estate:														
Other		209		209		6		219		6		6		
Total	\$ 2	,195	\$	1,923	\$	34	\$	968	\$	105	\$	105		
				14										

## CONSUMERS BANCORP, INC. Notes to the Consolidated Financial Statements

(Unaudited) (continued)

(Dollars in thousands, except per share amounts)

The following table presents information related to average recorded investment and interest income associated with loans individually evaluated for impairment by class of loans for the three months ended December 31, 2019:

	Average Recorded Investment		In	terest come ognized	ash Basis Interest ecognized
With no related allowance	11110	Suncin	RCC	gilized	 cognized
recorded:					
Commercial real estate:					
Other	\$	244	\$	1	\$ 1
1-4 Family residential real					
estate:					
Owner occupied		10		_	
Non-owner occupied		250		_	_
With an allowance recorded:					
Commercial		165		2	2
Commercial real estate:					
Other		218		3	3
Total	\$	887	\$	6	\$ 6

The following table presents the recorded investment in non-accrual and loans past due over 90 days still on accrual by class of loans as of December 31, 2020 and June 30, 2020:

	Dece	mber	31, 2020	June 3	30, 2020	
			Loans Past		Loans	s Past
			Due		Dι	ıe
			Over 90		Ove	r 90
			Days		Da	ys
			Still		St	ill
	Non-accr	ual	Accruing	Non-accrual	Accr	uing
Commercial	\$	_	\$ —	\$ 21	\$	
Commercial real estate:						
Other		902	_	785		_
1 – 4 Family residential:						
Owner occupied		821	_	143		29
Non-owner occupied		217	_	236		
Consumer		_				12
Total	\$ 1,	940	<u>\$</u>	\$ 1,185	\$	41

Non-accrual loans and loans past due 90 days still on accrual include both smaller balance homogeneous loans that are collectively evaluated for impairment and individually classified impaired loans.

### CONSUMERS BANCORP, INC.

## Notes to the Consolidated Financial Statements (Unaudited) (continued)

(Dollars in thousands, except per share amounts)

The following table presents the aging of the recorded investment in past due loans as of December 31, 2020 by class of loans:

		Days Past Due				
	30 - 59	60 - 89	90 Days or	Total	Loans Not	
	Days	Days	Greater	Past Due	Past Due	Total
Commercial	\$ —	\$ —	\$ —	\$ —	\$ 159,140	\$ 159,140
Commercial real estate:						
Construction	_	_	_	_	3,791	3,791
Other	_	_	629	629	250,044	250,673
1-4 Family residential:						
Owner occupied	450	4	269	723	93,926	94,649
Non-owner occupied	_	_	_	_	20,223	20,223
Construction	4	_	_	4	6,121	6,125
Consumer	261	49	_	310	23,953	24,263
Total	\$ 715	\$ 53	\$ 898	\$ 1,666	\$ 557,198	\$ 558,864

The above table of past due loans includes the recorded investment in non-accrual loans of \$898 in the 90 days or greater category and \$1,042 in the loans not past due category.

The following table presents the aging of the recorded investment in past due loans as of June 30, 2020 by class of loans:

			Day	s Past Due							
	3	30 - 59 60 - 89 90 1			Days or	r Total			oans Not		
		Days		Days		Greater	Past Due		Past Due		 Total
Commercial	\$		\$	_	\$	21	\$	21	\$	157,307	\$ 157,328
Commercial real estate:											
Construction		_		_		_		_		16,241	16,241
Other		_		2		628		630		229,086	229,716
1-4 Family residential:											
Owner occupied		_		_		172		172		91,102	91,274
Non-owner occupied		_		_		_		_		19,410	19,410
Construction		_		_		_		_		9,435	9,435
Consumer		127		49		12		188		21,205	 21,393
Total	\$	127	\$	51	\$	833	\$	1,011	\$	543,786	\$ 544,797

The above table of past due loans includes the recorded investment in non-accrual loans of \$2 in the 60-89 days, \$792 in the 90 days or greater category and \$391 in the loans not past due category.

### **Troubled Debt Restructurings (TDR):**

The Corporation has certain loans that have been modified in order to maximize collection of loan balances that are classified as TDRs. A modified loan is usually classified as a TDR if, for economic reasons, management grants a concession to the original terms and conditions of the loan to a borrower who is experiencing financial difficulties that it would not have otherwise considered. In response to COVID-19, on March 22, 2020 the Corporation adopted a loan modification program to assist borrowers impacted by the virus. The program is available to most borrowers whose loan was not past due on March 22, 2020, the date this loan modification program was adopted. The program offers principal and interest payment deferrals for up to 90 days or interest only payments for up to 90 days. Borrowers are eligible for an additional 90 days of payment deferrals if situations warrant a need for an extension. Interest will be deferred but will continue to accrue during the deferment period and the maturity date on amortizing loans will be extended by the number of months the payment was deferred. Consistent with issued regulatory guidance, modifications made under this program in response to COVID-19 will not be classified as TDRs. As of December 31, 2020, 18 borrowers with an aggregate outstanding balance of \$667 are in payment deferral status under this loan modification program.

16

and June 30, 2020, the Corporation had \$3 and \$12, respectively, of specific reserve allocated to these loans.

As of December 31, 2020 and June 30, 2020, the Corporation had \$934 and \$974, respectively, of loans classified as TDRs which are included in impaired loans above. As of December 31, 2020 and June 30, 2020, the Corporation had not committed to lend any additional funds to customers with outstanding loans that were classified as troubled debt restructurings. As of December 31, 2020

### CONSUMERS BANCORP, INC.

### Notes to the Consolidated Financial Statements (Unaudited) (continued)

(Dollars in thousands, except per share amounts)

During the three- and six-month periods ended December 31, 2020 and 2019, there were no loan modifications completed that were classified as troubled debt restructurings. There were no charge offs from troubled debt restructurings that were completed during the three- and six-month periods ended December 31, 2020 and 2019.

There were no loans classified as troubled debt restructurings for which there was a payment default within 12 months following the modification during the three- and six-month periods ended December 31, 2020 and 2019. A loan is considered in payment default once it is 90 days contractually past due under the modified terms.

### **Credit Quality Indicators:**

The Corporation categorizes loans into risk categories based on relevant information about the ability of borrowers to service their debt such as: current financial information, historical payment experience, credit documentation, public information, current economic trends and other relevant information. The Corporation analyzes loans individually by classifying the loans as to credit risk. This analysis includes loans with a total outstanding loan relationship greater than \$100 and non-homogeneous loans, such as commercial and commercial real estate loans. Management monitors the loans on an ongoing basis for any changes in the borrower's ability to service their debt and affirms the risk ratings for the loans and leases in their respective portfolio on an annual basis. The Corporation uses the following definitions for risk ratings:

**Special Mention.** Loans classified as special mention have a potential weakness that deserves management's close attention. If left uncorrected, these potential weaknesses may result in deterioration of the repayment prospects for the loan or of the institution's credit position at some future date.

**Substandard.** Loans classified as substandard are inadequately protected by the current net worth and paying capacity of the obligor or of the collateral pledged, if any. Loans so classified have a well-defined weakness or weaknesses that jeopardize the liquidation of the debt. They are characterized by the distinct possibility that the institution will sustain some loss if the deficiencies are not corrected.

**Doubtful.** Loans classified as doubtful have all the weaknesses inherent in those classified as substandard, with the added characteristic that the weaknesses make collection or liquidation in full, on the basis of currently existing facts, conditions, and values, highly questionable and improbable.

Loans not meeting the criteria above that are analyzed individually as part of the above described process are considered pass rated loans. Loans listed as not rated are either less than \$100 or are included in groups of homogeneous loans. Generally, 1-4 Family Residential and Consumer loans are not risk rated, except when collateral is used for a business purpose. These loans are evaluated based on delinquency status, which are disclosed in the previous table within this footnote. Based on the most recent analysis performed, the recorded investment by risk category of loans by class of loans was as follows:

	As of December 31, 2020											
			Sı	pecial						Not		
		Pass	M	ention	Subst	tandard	Do	oubtful		Rated		
Commercial	\$	158,054	\$	453	\$	359	\$	_	\$	274		
Commercial real estate:												
Construction		3,791		_		_		_				
Other		241,702		1,933		5,075		902		1,061		
1-4 Family residential real estate:												
Owner occupied		1,988		_		18		603		92,040		
Non-owner occupied		19,265		173		210		217		358		
Construction		1,306		_				_		4,819		
Consumer		687		_		_		_		23,576		
Total	\$	426,793	\$	2,559	\$	5,662	\$	1,722	\$	122,128		
		17										

(Dollars in thousands, except per share amounts)

As of June 30, 2020, and based on the most recent analysis performed, the recorded investment by risk category of loans by class of loans is as follows:

	As of June 30, 2020												
				Special						Not			
		Pass	1	Mention	Subs	tandard	Do	ubtful		Rated			
Commercial	\$	152,911	\$	143	\$	3,979	\$	21	\$	274			
Commercial real estate:													
Construction		16,241		_		_		_		_			
Other		220,311		1,469		5,378		785		1,773			
1-4 Family residential real estate:													
Owner occupied		2,419		_		334				88,521			
Non-owner occupied		18,435		186		223		236		330			
Construction		3,234		_		_				6,201			
Consumer		153		_		_		_		21,240			
Total	\$	413,704	\$	1,798	\$	9,914	\$	1,042	\$	118,339			

### Note 5 - Fair Value

Fair value is the exchange price that would be received for an asset or paid to transfer a liability (an exit price) in the principal or most advantageous market for the asset or liability in an orderly transaction between market participants on the measurement date. There are three levels of inputs that may be used to measure fair values:

Level 1: Quoted prices (unadjusted) for identical assets or liabilities in active markets that the entity has the ability to access as of the measurement date.

Level 2: Significant other observable inputs other than Level 1 prices such as quoted prices for similar assets or liabilities; quoted prices in markets that are not active; or other inputs that are observable or can be corroborated by observable market data.

Level 3: Significant unobservable inputs that reflect a company's own assumptions about the assumptions that market participants would use in pricing an asset or liability.

Financial assets and financial liabilities measured at fair value on a recurring basis include the following:

Securities available-for-sale: When available, the fair values of available-for-sale securities are determined by obtaining quoted prices on nationally recognized securities exchanges (Level 1 inputs). For securities where quoted market prices are not available, fair values are calculated based on market prices of similar securities (Level 2 inputs). For securities where quoted prices or market prices of similar securities are not available, fair values are calculated using discounted cash flows or other unobservable inputs (Level 3 inputs).

(Dollars in thousands, except per share amounts)

Assets and liabilities measured at fair value on a recurring basis are summarized below, segregated by the level of the valuation inputs within the fair value hierarchy utilized to measure fair value:

				ir Value Measurements at December 31, 2020				
	Balan Decer 31 202	mber ,	Level 1	Lev	rel 2	]	Level 3	
Assets:								
U.S. Treasury	\$	501	\$ 	\$	501	\$	_	
Obligations of U.S. government-sponsored entities and agencies		8,109	_		8,109		_	
Obligations of states and political subdivisions	(	57,034	_		67,034		_	
U.S. Government-sponsored mortgage-backed securities – residential	4	58,248	_		58,248		_	
U.S. Government-sponsored mortgage-backed securities – commercial		7,078	_		7,078		_	
U.S. Government-sponsored collateralized mortgage obligations - residential		4,695	_		4,695		_	
Equity securities		400	_		400		_	
			 Fair V		easureme 0, 2020	ents a	at	
	Balan June 202	30,	Level 1	Lev	rel 2	]	Level 3	
Assets:								
Securities available-for-sale:								
Obligations of U.S. Treasury	\$	1,256	\$ _	\$	1,256	\$	_	
Obligations of government-sponsored entities		0,532	_		10,532		_	
Obligations of states and political subdivisions		53,492	_		63,492		_	
U.S. Government-sponsored mortgage-backed securities - residential	4	50,156	_		50,156		_	
U.S. Government-sponsored mortgage-backed securities – commercial		8,497	_		8,497		_	
U.S. Government-sponsored collateralized mortgage obligations		9,985	_		9,985		_	

There were no transfers between Level 1 and Level 2 during the three-month and six-month periods ended December 31, 2020 or 2019.

Certain assets and liabilities are measured at fair value on a non-recurring basis; that is, the instruments are not measured at fair value on an ongoing basis but are subject to fair value adjustments in certain circumstances. Assets and liabilities measured at fair value on a non-recurring basis include the following:

Impaired Loans: At the time a loan is considered impaired, it is valued at the lower of cost or fair value. Impaired loans carried at fair value generally receive specific allocations of the allowance for loan losses or are charged down to their fair value. For collateral dependent loans, fair value is commonly based on recent real estate appraisals. These appraisals may utilize a single valuation approach or a combination of approaches including comparable sales and the income approach. Adjustments are routinely made in the appraisal process by the appraisers to adjust for differences between the comparable sales and income data available. Such adjustments are usually significant and typically result in a Level 3 classification of the inputs for determining fair value. There were no impaired loans measured at fair value on a non-recurring basis at December 31, 2020 or June 30, 2020 and there was no impact to the provision for loan losses for the three- or six-month periods ended December 31, 2020 or 2019.

(Dollars in thousands, except per share amounts)

Other Real Estate and Repossessed Assets Owned: Assets acquired through or instead of loan foreclosure are initially recorded at fair value less costs to sell when acquired, establishing a new cost basis. These assets are subsequently accounted for at lower of cost or fair value less estimated costs to sell. Real estate owned properties and other repossessed assets, which are primarily vehicles, are evaluated on a quarterly basis for additional impairment and adjusted accordingly. There was no other real estate owned or other repossessed assets being carried at fair value as of December 31, 2020 or June 30, 2020.

The following table shows the estimated fair values of financial instruments that are reported at amortized cost in the Corporation's consolidated balance sheets, segregated by the level of the valuation inputs within the fair value hierarchy utilized to measure fair value:

Decemb	er 3	31, 2020		June	30,	2020
Carrying Amount		Estimated Fair Value		Carrying Amount		Estimated Fair Value
\$ 9,045	\$	9,045	\$	9,659	\$	9,659
8,599		8,819		11,635		11,889
4,770		4,907		3,507		3,566
2,327		2,327		2,646		2,646
3,336		3,605		3,541		3,868
551,345		565,397		537,183		548,247
540,811		540,811		517,973		517,973
107,551		108,158		115,382		116,238
13,275		13,275		6,943		6,943
18,083		18,423		31,161		31,571
88		88		107		107
	\$ 9,045  \$ 9,045  \$ 8,599  4,770  2,327  3,336  551,345  540,811  107,551  13,275  18,083	\$ 9,045 \$ 8,599 4,770 2,327 3,336 551,345 540,811 107,551 13,275 18,083	Carrying Amount     Fair Value       \$ 9,045     \$ 9,045       8,599     8,819       4,770     4,907       2,327     2,327       3,336     3,605       551,345     565,397       540,811     540,811       107,551     108,158       13,275     13,275       18,083     18,423	Carrying Amount         Estimated Fair Value           \$ 9,045         \$ 9,045           \$ 8,599         8,819           4,770         4,907           2,327         2,327           3,336         3,605           551,345         565,397           540,811         540,811           107,551         108,158           13,275         13,275           18,083         18,423	Carrying Amount         Estimated Fair Value         Carrying Amount           \$ 9,045         \$ 9,045         \$ 9,659           8,599         8,819         11,635           4,770         4,907         3,507           2,327         2,327         2,646           3,336         3,605         3,541           551,345         565,397         537,183           540,811         540,811         517,973           107,551         108,158         115,382           13,275         13,275         6,943           18,083         18,423         31,161	Carrying Amount         Estimated Fair Value         Carrying Amount           \$ 9,045         \$ 9,045         \$ 9,659         \$           8,599         8,819         11,635

### Note 6 - Earnings Per Share

Basic earnings per share is the amount of earnings available to each share of common stock outstanding during the reporting period and is equal to net income divided by the weighted average number of shares outstanding during the period. Diluted earnings per share is the amount of earnings available to each share of common stock outstanding during the reporting period adjusted to include the effect of potentially dilutive common shares that may be issued upon the vesting of restricted stock awards. There were 569 shares of restricted stock that were anti-dilutive for the six-month period ended December 31, 2020. There were 1,645 and 2,204 shares of restricted stock that were anti-dilutive for the three- and six-month periods ended December 30, 2019. The following table details the calculation of basic and diluted earnings per share:

	For the Three Months Ended December 31,			For the Six Months Ended December 31,			
	 2020 2019		2020		2019		
Basic:							
Net income available to common shareholders	\$ 2,507	\$	1,440	\$	4,908	\$	2,943
Weighted average common shares outstanding	3,017,268		2,737,800		3,015,741		2,737,755
Basic income per share	\$ 0.83	\$	0.53	\$	1.63	\$	1.07

(Dollars in thousands, except per share amounts)

Diluted:				
Net income available to common shareholders	\$ 2,507	\$ 1,440	\$ 4,908	\$ 2,943
Weighted average common shares outstanding	3,017,268	2,737,800	3,015,741	2,737,755
Dilutive effect of restricted stock	 21		_	
Total common shares and dilutive potential common shares	3,017,289	2,737,800	3,015,741	2,737,755
Dilutive income per share	\$ 0.83	\$ 0.53	\$ 1.63	\$ 1.07

### Note 7 - Accumulated Other Comprehensive Income (Loss)

The components of other comprehensive income related to unrealized gains and losses on available-for-sale securities for the three and six-month periods ended December 31, 2020 and 2019, were as follows:

	Pretax	Tax Effect	After-tax	Affected Line Item in Consolidated Statements of Income
Balance as of September 30, 2020	\$ 5,463	\$ (1,147)	\$ 4,316	
Unrealized holding loss on available-for-sale securities arising during the period	278	(59)	219	
Amounts reclassified from accumulated other comprehensive income	 (8)	 2	 (6)	(a)(b)
Net current period other comprehensive loss	270	(57)	213	
Balance as of December 31, 2020	\$ 5,733	\$ (1,204)	\$ 4,529	
Balance as of September 30, 2019	\$ 2,694	\$ (566)	\$ 2,128	
Unrealized holding gain on available-for-sale securities arising during the period	(28)	5	(23)	
Amounts reclassified from accumulated other comprehensive income	 (4)	1	(3)	(a)(b)
Net current period other comprehensive income	(32)	6	(26)	
Balance after reclassification as of December 31, 2019	\$ 2,662	\$ (560)	\$ 2,102	

<sup>(</sup>a) Securities (gains) losses, net

<sup>(</sup>b) Income tax expense

(Dollars in thousands, except per share amounts)

	Pretax	Tax Effect			After-tax	Affected Line Item in Consolidated Statements of Income
Balance as of June 30, 2020	\$ 5,393	\$	(1,133)	\$	4,260	
Unrealized holding gain on available-for-sale securities arising during the period	348		(73)		275	
Amounts reclassified from accumulated other comprehensive income	(8)		2		(6)	(a)(b)
Net current period other comprehensive income	340		(71)		269	
Balance as of December 31, 2020	\$ 5,733	\$	(1,204)	\$	4,529	
Balance as of June 30, 2019	\$ 1,982	\$	(416)	\$	1,566	
Unrealized holding gain on available-for-sale securities arising during the period	790		(166)		624	
Amounts reclassified from accumulated other comprehensive income	 (110)		22		(88)	(a)(b)
Net current period other comprehensive income	680		(144)		536	
Balance after reclassification as of December 31, 2019	\$ 2,662	\$	(560)	\$	2,102	

- (a) Securities (gains) losses, net
- (b) Income tax expense

### Note 8 - COVID-19

In December 2019, a novel strain of coronavirus surfaced in Wuhan, China, and has spread around the world, resulting in business and social disruption. The coronavirus was declared a Pandemic by the World Health Organization on March 11, 2020. The operations and business results of the Corporation could be materially adversely affected. The extent to which the coronavirus may impact business activity or investment results will depend on future developments, which are highly uncertain and cannot be predicted, including new information which may emerge concerning the severity of the coronavirus and the actions required to contain the coronavirus or treat its impact, among others. As a result of the economic shutdown engineered to slow down the spread of COVID-19, the ability of our customers to make payments on loans could be adversely impacted, resulting in elevated loan losses and an increase in the Corporation's allowance for loan losses. Additionally, it is reasonably possible future evaluations of the carrying amount of goodwill could result in a conclusion that goodwill is impaired.

### CONSUMERS BANCORP, INC.

### Item 2 - Management's Discussion and Analysis of Financial Condition and Results of Operations

(Dollars in thousands, except per share data)

#### General

The following is management's analysis of the Corporation's results of operations for the three- and six-month periods ended December 31, 2020, compared to the same period in 2019, and the consolidated balance sheet at December 31, 2020, compared to June 30, 2020. This discussion is designed to provide a more comprehensive review of the operating results and financial condition than could be obtained from an examination of the financial statements alone. This analysis should be read in conjunction with the consolidated financial statements and related footnotes and the selected financial data included elsewhere in this report.

### Overview

Consumers Bancorp, Inc., a bank holding company incorporated under the laws of the State of Ohio (the Corporation), owns all of the issued and outstanding common shares of Consumers National Bank, a bank chartered under the laws of the United States of America (the Bank). The Corporation's activities have been limited primarily to holding the common shares of the Bank. The Bank's business involves attracting deposits from businesses and individual customers and using such deposits to originate commercial, mortgage and consumer loans in its market area, consisting primarily of Carroll, Columbiana, Jefferson, Stark, Summit, Wayne and contiguous counties in Ohio. The Bank also invests in securities consisting primarily of U.S. government sponsored entities, municipal obligations, mortgage-backed and collateralized mortgage obligations issued by Fannie Mae, Freddie Mac and Ginnie Mae.

On December 29, 2020, the Bank entered into a P&A Agreement with CFBank to acquire CFBank's drive-up branch location in Wellsville, Ohio and CFBank's branch location in Calcutta, Ohio. As part of this transaction, the Bank will acquire approximately \$100 million in deposits attributable to the Branches; \$15 million in aggregate principal amount of subordinated debt securities issued by unrelated financial institutions; all performing loans attributable to the Branches which are outstanding at closing (totaling approximately \$3.1 million in aggregate principal amount as of November 30, 2020); and up to \$13.5 million in aggregate principal amount of single family residential mortgage loans and home equity lines of credit to be identified by the parties prior to the closing principally from CFBank's Northeast Ohio loan portfolio. In addition, CFBank will provide the opportunity for the Corporation to purchase at par at least \$15 million in aggregate principal amount of participation interests in commercial and commercial real estate loans originated by and held in CFBank's portfolio. The closing of the transactions is subject to regulatory approval and satisfaction of other customary closing conditions and is expected to occur early in the third calendar quarter of 2021.

On January 1, 2020, the Corporation completed the acquisition by merger of Peoples Bancorp of Mt. Pleasant, Inc. (Peoples) in a stock and cash transaction for an aggregate consideration of approximately \$10,405. In connection with the acquisition, the Corporation issued 269,920 shares of common stock and paid \$5,128 in cash to the former shareholders of Peoples. On December 31, 2019, Peoples had approximately \$72,016 in total assets, \$55,273 in loans and \$60,826 in deposits at its three banking centers located in Mt. Pleasant, Adena, and Dillonvale, Ohio.

### **COVID-19 Pandemic**

In response to COVID-19, management is actively pursuing multiple avenues to assist customers during these uncertain times. For commercial borrowers, the Coronavirus Aid, Relief and Economic Security Act (the CARES Act) includes key SBA initiatives to assist small businesses. The Paycheck Protection Program (PPP) was designed to provide a direct incentive for small businesses to keep their workers on the payroll. The SBA will forgive loans obtained under this program if the borrower meets payroll and other requirements. A total of \$52,539 of PPP loans from the first round of assistance were outstanding as of December 31, 2020. Demand for the second round of the PPP program has been strong and submission of applications began on January 15, 2021. As of February 2, 2021, there were a total of \$25,831 of loans funded and an additional \$8,263 million submitted and waiting the SBA's approval as part of the second round of the PPP loan program.

Additionally, on March 22, 2020 the Corporation adopted a loan modification program to assist borrowers impacted by COVID-19. The program is available to most borrowers whose loan was not past due on March 22, 2020, the date this loan modification program was adopted. The program offers principal and interest payment deferrals for up to 90 days or interest only payments for up to 90 days. Borrowers are eligible for an additional 90 days of payment deferrals if situations warrant a need for an extension. Interest will be deferred but will continue to accrue during the deferment period and the maturity date on amortizing loans will be extended by the number of months the payment was deferred. Consistent with issued regulatory guidance, modifications made under this program in response to COVID-19 will not be classified as troubled debt restructurings. As of December 31, 2020, 18 borrowers with an outstanding balance of \$667 are in payment deferral status under this loan modification program.

#### CONSUMERS BANCORP, INC.

## Management's Discussion and Analysis of Financial Condition and Results of Operations (continued)

(Dollars in thousands, except per share amounts)

We have assisted and may continue to assist customers who are experiencing financial hardship due to COVID-19 by waiving late charges, refunding NSF and overdraft fees, and waiving CD prepayment penalties. The consumer reserve personal line of credit, an on this unsecured line of credit that is linked to a personal checking account, has been redesigned to provide easier access and a lower initial rate. Commercial customers have been encouraged to access available funds on their lines of credit, and we've been ready to provide emergency commercial lines of credit to qualified borrowers in order to assist in meeting payroll and other recurring fixed expenses. In response to COVID-19, we provided four emergency lines of credit; however, the lines of credit have since been closed as the borrowers did not need to access the funds.

The Corporation has modified its business practices with a portion of employees working remotely from their homes to limit interruptions to operations as much as possible and to help reduce the risk of COVID-19 infecting entire departments. With the holiday surge in COVID-19 cases, the branch lobbies were again closed to the public for walk-in transactions but available to customers to complete paperwork or complex transactions. If the infection rate improves, we expect to reopen branch lobbies during the first calendar quarter of 2021. The Corporation is encouraging virtual meetings and conference calls in place of in-person meetings. Additionally, travel for business has been restricted. The Corporation is promoting social distancing, frequent hand washing and thorough disinfection of all surfaces.

### **Results of Operations**

### Three- and Six-Month Periods Ended December 31, 2020 and 2019

Net income for the second quarter of fiscal year 2021 was \$2,507, or \$0.83 per common share, compared to \$1,440, or \$0.53 per common share for the three months ended December 31, 2019. The following are key highlights of our results of operations for the three months ended December 31, 2020 compared to the prior fiscal year comparable period:

- net interest income increased by \$2,062 to \$6,858, or 43.0%, in the second quarter of fiscal year 2021 from the same prior year period primarily as a result of an increase in average interest earning assets and a reduction in the cost of funds;
- a \$130 provision for loans loss expense was recorded for the second quarter of fiscal year 2021 compared with \$185 for the prior year period;
- noninterest income increased by \$128, or 12.5%, in the second quarter of fiscal year 2021 from the same prior year period primarily as a result of a \$104, or 73.2%, increase on gains from the sale of mortgage loans and a \$61, or 15.9%, increase in debit card interchange income; and
- noninterest expenses increased by \$896, or 22.8%, in the second quarter of fiscal year 2021 from the same prior year period primarily due to an increase in salaries and employee benefit expenses due to the inclusion of a full quarter of expenses associated with the three new office locations and additional staff gained as a result of the merger with Peoples, increased incentive accruals, and higher mortgage commissions due to the increase in volume.

In the first six months of fiscal year 2021, net income was \$4,908, or \$1.63 per common share, compared to \$2,943, or \$1.07 per common share, for the six months ended December 31, 2019. The following are key highlights of our results of operations for the six months ended December 31, 2020:

- net interest income increased by \$4,075 to \$13,552, or by 43.0%, in the first six months of fiscal year 2021 from the same prior year period;
- a provision for loan loss expense of \$260 was recorded in the first six months of fiscal year 2021 compared with a provision for loan loss expense of \$315 during the same prior year period; noninterest income decreased by \$200 in the first six months of fiscal year 2021 primarily since the prior year period
- included \$324 of income recognized as a result of proceeds received from a bank owned life insurance policy claim and a \$110 gain on the sale of securities. These reductions were partially offset by a \$205, or 74.0%, increase on gains from the sale of mortgage loans and a \$126, or 16.3%, increase in debit card interchange income; and
- noninterest expenses increased by \$1,390, or 16.9%, in the first six months of fiscal year 2021 from the same prior year period primarily due to an increase in salaries and employee benefit expenses due to the inclusion of six months of expenses associated with the three new office locations and additional staff gained as a result of the mercer with Peoples, increased
- associated with the three new office locations and additional staff gained as a result of the merger with Peoples, increased incentive accruals, and higher mortgage commissions due to the increase in volume.

## Management's Discussion and Analysis of Financial Condition and Results of Operations (continued)

(Dollars in thousands, except per share amounts)

Return on average equity and return on average assets were 14.83% and 1.31%, respectively, for the six months ended December 31, 2020 compared to 11.03% and 1.04%, respectively, for the same prior year period.

### **Net Interest Income**

Net interest income, the difference between interest income earned on interest-earning assets and interest expense incurred on interest-bearing liabilities, is the largest component of the Corporation's earnings. Net interest income is affected by changes in the volumes, rates and composition of interest-earning assets and interest-bearing liabilities. In addition, prevailing economic conditions, fiscal and monetary policies and the policies of various regulatory agencies all affect market rates of interest and the availability and cost of credit, which, in turn, can significantly affect net interest income. As a result of the Federal Open Market Committee established a near-zero target range for the federal funds rate, earnings could be negatively affected if the interest we receive on loans and securities falls more quickly than interest we pay on deposits and borrowings. Net interest margin is calculated by dividing net interest income on a fully tax equivalent basis (FTE) by total average interest-earning assets. FTE income includes tax-exempt income, restated to a pre-tax equivalent, based on the statutory federal income tax rate. The federal income tax rate in effect for the 2021 and 2020 fiscal years was 21.0%. All average balances are daily average balances. Non-accruing loans are included in average loan balances.

The Corporation's net interest margin was 3.87% for the three months ended December 31, 2020, compared with 3.62% for the same period in 2019. FTE net interest income for the three months ended December 31, 2020 increased by \$2,090, or 42.9%, to \$6,964 from \$4,874 for the same prior year period.

Tax-equivalent interest income for the three months ended December 31, 2020 increased by \$1,667, or 28.5%, from the same prior year period. Interest income was positively impacted by a \$182,889, or 34.1%, increase in average interest-earning assets from the same prior year period due to the assets acquired from the Peoples acquisition as well as organic loan growth. Additionally, interest income was positively impacted by the accretion of origination fees from the PPP loans and from a change in the earning asset mix, with higher yielding loans increasing faster than lower yielding securities. However, a reduction in the accretion of origination fees from PPP loans as these loans are forgiven and the significant decline in interest rates will continue to impact the yield on interest-earning assets and could ultimately result in a decline in interest income. The Corporation's yield on average interest-earning assets was 4.18% for the three months ended December 31, 2020 compared with 4.35% for the same period last year.

Interest expense for the three months ended December 31, 2020 decreased by \$423, or 43.1%, from the same prior year period primarily due to a reduction in deposit and borrowing costs as a result of lower market interest rates. The Corporation's cost of funds was 0.46% for the three months ended December 31, 2020 compared with 1.02% for the same prior year period.

The Corporation's net interest margin was 3.86% for the six months ended December 31, 2020, compared with 3.62% for the same period in 2019. FTE net interest income for the six months ended December 31, 2020 increased by \$4,124, or 42.8%, to \$13,756 from \$9,632 for the same prior year period.

Tax-equivalent interest income for the six months ended December 31, 2020 increased by \$3,318, or 28.5%, from the same prior year period. Interest income was positively impacted by a \$182,309, or 34.4%, increase in average interest-earning assets from the same prior year period due to the assets acquired from the Peoples acquisition as well as organic loan growth. Additionally, interest income was positively impacted by the accretion of origination fees from the PPP loans and from a change in the earning asset mix, with higher yielding loans increasing faster than lower yielding securities. However, a reduction in the accretion of origination fees from PPP loans as these loans are forgiven and the significant decline in interest rates will continue to impact the yield on interest-earning assets and could ultimately result in a decline in interest income. The Corporation's yield on average interest-earning assets was 4.20% for the six months ended December 31, 2020 compared with 4.38% for the same period last year.

Interest expense for the six months ended December 31, 2020 decreased by \$806 from the same prior year period. The Corporation's cost of funds was 0.50% for the six months ended December 31, 2020 compared with 1.05% for the same prior year period. The decline in short term market interest rates had an impact on the rates paid on all interest-bearing deposit products and Federal Home Loan Bank (FHLB) advances.

# Management's Discussion and Analysis of Financial Condition and Results of Operations (continued)

(Dollars in thousands, except per share amounts)

# <u>Average Balance Sheets and Analysis of Net Interest Income for the Three Months Ended December 31,</u> (In thousands, except percentages)

				2020					2019	
		Average			Yield/		Average			Yield/
		Balance		Interest	Rate		Balance	]	Interest	Rate
Interest-earning assets:										
Taxable securities	\$	74,051	\$	344	1.90%	\$	79,160	\$	480	2.43%
Nontaxable securities (1)		67,892		529	3.24		61,281		475	3.17
Loans receivable (1)		552,897		6,587	4.73		390,708		4,865	4.94
Federal bank and other restricted stocks		2,472		21	3.37		1,723		20	4.61
Interest bearing deposits and federal funds sold		21,742		42	0.77		3,293		16	1.93
Total interest-earning assets		719,054		7,523	4.18%		536,165		5,856	4.35%
Noninterest-earning assets		30,865					31,384			
m . I .	Φ	740.010				Ф	5(7,540			
Total Assets	<u>\$</u>	749,919				\$	567,549			
Interest-bearing liabilities:										
NOW	\$	107,191	\$	38	0.14%	\$	84,140	\$	131	0.62%
Savings		235,414		84	0.14		170,287		210	0.49
Time deposits		112,543		365	1.29		111,806		569	2.02
Short-term borrowings		8,596		2	0.09		3,915		13	1.32
FHLB advances		20,006		70	1.39		12,627		59	1.85
Total interest-bearing liabilities		483,750		559	0.46%		382,775		982	1.02%
Noninterest-bearing liabilities:										
Noninterest-bearing checking accounts		193,587					126,270			
Other liabilities		5,907					4,900			
Total liabilities		683,244					513,945			
Shareholders' equity		66,675					53,604			
Total liabilities and shareholders' equity	\$	749,919				\$	567,549			
Net interest income, interest rate spread (1)			\$	6,964	3.72%			\$	4,874	3.33%
Net interest margin (net interest as a percent of average interest-earning assets) (1)					3.87%					3.62%
Federal tax exemption on non- taxable securities and loans included in interest income			\$	106				\$	78	
Average interest-earning assets to interest-bearing liabilities		148.64%	, )				140.07%	, D		

(1) calculated on a fully taxable equivalent basis utilizing a statutory federal income tax rate of 21.0%

26

# Management's Discussion and Analysis of Financial Condition and Results of Operations (continued)

(Dollars in thousands, except per share amounts)

## Average Balance Sheets and Analysis of Net Interest Income for the Six Months Ended December 31,

(In thousands, except percentages)

				2020					2019	
		Average			Yield/		Average			Yield/
	]	Balance		Interest	Rate	]	Balance		Interest	Rate
Interest-earning assets:										
Taxable securities	\$	76,240	\$	716	1.92%	\$	81,399	\$	990	2.43%
Nontaxable securities (1)		67,183		1,043	3.24		61,029		949	3.18
Loans receivable (1)		548,530		13,080	4.73		382,035		9,628	5.00
Federal bank and other restricted stocks		2,472		39	3.13		1,723		40	4.61
Interest bearing deposits and federal funds sold		18,409		89	0.96		4,339		42	1.92
Total interest-earning assets		712,834		14,967	4.20%		530,525		11,649	4.38%
Noninterest-earning assets		30,600					31,107			
Total Assets	\$	743,434				\$	561,632			
Interest-bearing liabilities:										
NOW	\$	104,646	\$	83	0.16%	\$	83,316	\$	276	0.66%
Savings		231,954		189	0.16		168,234	•	432	0.51
Time deposits		113,680		792	1.38		112,224		1,147	2.03
Short-term borrowings		8,238		6	0.14		3,620		24	1.32
FHLB advances		22,059		141	1.27		14,003		138	1.95
Total interest-bearing liabilities		480,577		1,211	0.50%		381,397		2,017	1.05%
Noninterest-bearing liabilities:										
Noninterest-bearing checking accounts		191,360					122,263			
Other liabilities		5,851					5,071			
Total liabilities		677,788					508,731			
Shareholders' equity		65,646					52,901			
Total liabilities and shareholders' equity	\$	743,434				\$	561,632			
Net interest income, interest rate spread (1)			\$	13,756	3.70%			\$	9,632	3.33%
Net interest margin (net interest as a percent of average interest-earning assets) (1)					3.86%					3.62%
Federal tax exemption on non- taxable securities and loans included in interest income			\$	204				\$	155	
Average interest-earning assets to interest-bearing liabilities		148.33%	, )				139.10%	ó		

(1) calculated on a fully taxable equivalent basis utilizing a statutory federal income tax rate of 21.0%

27

## Management's Discussion and Analysis of Financial Condition and Results of Operations (continued)

#### **Provision for Loan Losses**

The provision for loan losses represents the charge to income necessary to adjust the allowance for loan losses to an amount that represents management's assessment of the estimated probable incurred credit losses in the Bank's loan portfolio that have been incurred at each balance sheet date. For the six-month period ended December 31, 2020, the provision for loan losses was \$260 compared with \$315 for the same prior year period. Net charge-offs of \$26 were recorded during the six-month period ended December 31, 2020 compared with \$8 during the six-month period ended December 31, 2019.

Non-performing loans were \$1,940 as of December 31, 2020 compared with \$1,226 as of June 30, 2020 and \$427 as of December 31, 2019. Non-performing loans to total loans were 0.35% at December 31, 2020 and 0.23% at June 30, 2020. Non-accrual loans have primarily increased within the 1-4 family owner occupied loan category in part due to the COVID-19 pandemic impact on certain borrowers.

The allowance for loan losses as a percentage of loans was 1.06% at December 31, 2020 and 1.05% at June 30, 2020. As of December 31, 2020, the ALLL as a percentage of total loans excluding PPP loans was 1.17%. During the second quarter of fiscal year 2021, there was a significant decline in loans classified as substandard due to the full payoff of a large loan relationship that was in this category. Uncertainty remains regarding future levels of criticized and classified loans, non-performing loans and charge-offs, but some deterioration is expected as a result of the COVID-19 pandemic. Management will continue to closely monitor changes in the loan portfolio and adjust the provision accordingly.

### **Noninterest Income**

Noninterest income increased by \$128 for the second quarter of fiscal year 2021 from the same period last year primarily due to a \$104, or 73.2%, increase in gains from the sale of mortgage loans and a \$61, or 15.9%, increase in debit card interchange income. Debit card interchange income increased as a result of increased debit card usage and an increase in the number of cards issued. These increases were partially offset by a decline of \$46, or 12.8%, in service charges on deposit accounts primarily due to a decline in overdraft charges as the COVID-19 pandemic dramatically impacted consumer spending habits and increased NSF and overdraft fee waivers as we assist customers who have been impacted by COVID-19.

Noninterest income decreased by \$200 for the six-month period ended December 31, 2020 from the same period last year primarily since the prior year period included \$324 of income recognized as a result of proceeds received from a bank owned life insurance policy claim and a \$110 gain on the sale of securities. These reductions were partially offset by a \$205, or 74.0%, increase on gains from the sale of mortgage loans and a \$126, or 16.3%, increase in debit card interchange income, that were partially offset by a decline of \$112, or 15.3%, in service charges on deposit accounts. Gains from the sale of mortgage loans have been positively impacted by record low mortgage rates in 2020 helping increase the volume of loans sold.

### **Noninterest Expenses**

Total noninterest expenses increased by \$896, or 22.85%, for the second quarter of fiscal year 2021 compared with the same period last year. Noninterest expenses for the three-month period ended December 31, 2020 include expenses associated with the three new office locations and additional staff gained as a result of the merger with Peoples that closed on January 1, 2020. In addition, incentive accruals and mortgage commissions also increased during the second quarter of fiscal year 2021. Professional fees of \$59 associated with the announced planned purchase of two branch locations in Columbiana County, Ohio were recognized during the three-month period ended December 31, 2020.

Total noninterest expenses increased by \$1,390, or 16.9%, for the six-month period ended December 31, 2020 compared with the same period last year. Salaries and employee benefit expenses increased by \$1,071, or 24.5%, due to the inclusion of six months of expenses in fiscal year 2021 associated with the three new office locations and additional staff gained as a result of the merger with Peoples, increased incentive accruals, and higher mortgage commissions due to the increase in volume. Data processing expenses declined by \$164, or 31.2%, because fiscal year 2020 included expenses for system deconversion files related to the merger with Peoples. Also, FDIC assessments increased by \$150 for the current year-to-date period since the Small Bank Assessment Credits were applied to the FDIC insurance invoices for the six-month period ended December 31, 2019.

## Management's Discussion and Analysis of Financial Condition and Results of Operations (continued)

(Dollars in thousands, except per share amounts)

#### **Income Taxes**

Income tax expense was \$543 and \$1,048 for the three- and six-month periods ended December 31, 2020, respectively, compared to \$261 and \$473 for the three- and six-month periods ended December 31, 2019, respectively. The effective tax rate was 17.8% and 17.6% for the three- and six-month periods ended December 31, 2020, respectively, compared with 15.3% and 13.8% for the three- and six-month periods ended December 31, 2019, respectively. The effective tax rates differed from the federal statutory rate as a result of tax-exempt income from obligations of states and political subdivisions, loans and bank owned life insurance earnings and death benefit.

### **Financial Condition**

Total assets as of December 31, 2020 were \$754,060 compared to \$740,820 at June 30, 2020, an increase of \$13,240, or an annualized 3.5%.

Total loans increased by \$14,396, or an annualized 5.3%, from \$542,861 as of June 30, 2020 to \$557,257 as of December 31, 2020. As of December 31, 2020, total loans included \$52,539 of PPP loans, a decline of \$14,067, or 21.1%, from June 30, 2020.

As of December 31, 2020, total deposits increased by \$15,007, or an annualized 4.7%, from June 30, 2020. The Corporation has been able to maintain a favorable deposit mix, with 29.9% in noninterest-bearing deposits, 16.2% in interest bearing demand deposits, 37.3% in savings and money market deposits, and 16.6% in time deposits.

### **Non-Performing Assets**

The following table presents the aggregate amounts of non-performing assets and select ratios as of the dates indicated.

	mber 31, 2020	June 30, 2020	D	ecember 31, 2019
Non-accrual loans	\$ 1,940	\$ 1,185	\$	427
Loans past due over 90 days and still accruing	 	 41		<u> </u>
Total non-performing loans	1,940	1,226		427
Other repossessed assets	 	 7		<u> </u>
Total non-performing assets	\$ 1,940	\$ 1,233	\$	427
Non-performing loans to total loans	0.35%	0.23%		0.11%
Allowance for loan losses to total non-performing loans	304.74%	463.13%		959.02%

As of December 31, 2020, impaired loans totaled \$2,343, of which \$1,938 are included in non-accrual loans. Commercial and commercial real estate loans are classified as impaired if management determines that full collection of principal and interest, in accordance with the terms of the loan documents, is not probable. Impaired loans and non-performing loans have been considered in management's analysis of the appropriateness of the allowance for loan losses. Management and the Board of Directors are closely monitoring these loans and believe that the prospects for recovery of principal and interest, less identified specific reserves, are favorable.

### Contractual Obligations, Commitments, Contingent Liabilities and Off-Balance Sheet Arrangements

#### Liquidity

The objective of liquidity management is to ensure adequate cash flows to accommodate the demands of our customers and provide adequate flexibility for the Corporation to take advantage of market opportunities under both normal operating conditions and under unpredictable circumstances of industry or market stress. Cash is used to fund loans, purchase investments, fund the maturity of liabilities, and, at times, to fund deposit outflows and operating activities. The Corporation's principal sources of funds are deposits; amortization and prepayments of loans; maturities, sales and principal receipts from securities; borrowings; and operations. Management considers the asset position of the Corporation to be sufficiently liquid to meet normal operating needs and conditions. The Corporation's earning assets are mainly comprised of loans and investment securities. Management continually strives to obtain the best mix of loans and investments to both maximize yield and ensure the soundness of the portfolio, as well as to provide funding for loan demand as needed.

## Management's Discussion and Analysis of Financial Condition and Results of Operations (continued)

(Dollars in thousands, except per share amounts)

For the six months ended December 31, 2020, net cash inflow from operating activities was \$5,706, net cash outflows from investing activities was \$13,705 and net cash inflows from financing activities was \$7,385. A major source of cash was a \$15,007 increase in deposits and \$20,304 from maturities, calls or principal pay downs on available-for-sale securities. A major use of cash was \$24,975 purchases of available-for-sale securities and a \$14,431 increase in loans. Total cash and cash equivalents were \$9,045 as of December 31, 2020, compared to \$9,659 at June 30, 2020 and \$14,028 at December 31, 2019.

The Bank offers several types of deposit products to its customers. We believe the rates offered by the Bank and the fees charged for them are competitive with the rates and fees charged by other banks for similar deposit products currently available in the market area. Deposits totaled \$648,362 at December 31, 2020 compared with \$633,355 at June 30, 2020.

To provide an additional source of liquidity, the Corporation has entered into an agreement with the FHLB of Cincinnati. At December 31, 2020, advances from the FHLB of Cincinnati totaled \$18,083 compared with \$31,161 at June 30, 2020. As of December 31, 2020, the Bank had the ability to borrow an additional \$41,561 from the FHLB of Cincinnati based on a blanket pledge of qualifying first mortgage and multi-family loans. The Corporation considers the FHLB of Cincinnati to be a reliable source of liquidity funding, secondary to its deposit base.

Short-term borrowings consisted of repurchase agreements, which are financing arrangements that mature daily, and federal funds purchased from correspondent banks. The Bank pledges securities as collateral for the repurchase agreements. Short-term borrowings totaled \$13,275 at December 31, 2020 and \$6,943 at June 30, 2020.

Jumbo time deposits (those with balances of \$250 and over) totaled \$33,418 as of December 31, 2020 and \$36,747 as of June 30, 2020. These deposits are monitored closely by the Corporation and are mainly priced on an individual basis. When these deposits are from a municipality, certain bank-owned securities are pledged to guarantee the safety of these public fund deposits as required by Ohio law. The Corporation has the option to use a fee-paid broker to obtain deposits from outside its normal service area as an additional source of funding. The Corporation, however, does not rely upon these deposits as a primary source of funding. Although management monitors interest rates on an ongoing basis, a quarterly rate sensitivity report is used to determine the effect of interest rate changes on the financial statements. In the opinion of management, enough assets or liabilities could be repriced over the near term (up to three years) to compensate for such changes. The spread on interest rates, or the difference between the average earning assets and the average interest-bearing liabilities, is monitored quarterly.

## **Off-Balance Sheet Arrangements**

In the normal course of business, to meet the financial needs of our customers, we are a party to financial instruments with off-balance sheet risk. These financial instruments generally include commitments to originate mortgage, commercial and consumer loans, and involve to varying degrees, elements of credit and interest rate risk in excess of amounts recognized in the Consolidated Balance Sheets. The maximum exposure to credit loss in the event of nonperformance by the borrower is represented by the contractual amount of those instruments. Since commitments to extend credit have a fixed expiration date or other termination clause, some commitments will expire without being drawn upon and the total commitment amounts do not necessarily represent future cash requirements. The same credit policies are used in making commitments as are used for on-balance sheet instruments and collateral is required in instances where deemed necessary. Undisbursed balances of loans closed include funds not disbursed but committed for construction projects. Unused lines of credit include funds not disbursed, but committed for home equity, commercial and consumer lines of credit. Financial standby letters of credit are conditional commitments issued to guarantee the performance of a customer to a third party. The credit risk involved in issuing letters of credit is essentially the same as that involved in extending loan facilities to customers. Total unused commitments were \$108,614 as of December 31, 2020 and \$101,026 as of June 30, 2020.

### **Capital Resources**

Total shareholders' equity increased to \$67,708 as of December 31, 2020 from \$63,240 as of June 30, 2020. The primary reason for the increase in shareholders' equity was from net income of \$4,908 for the first six months of fiscal year 2021 which was partially offset by cash dividends paid of \$876.

The Bank is subject to various regulatory capital requirements administered by federal regulatory agencies. Capital adequacy guidelines and prompt corrective-action regulations involve quantitative measures of assets, liabilities, and certain off-balance-sheet

	30		

## Management's Discussion and Analysis of Financial Condition and Results of Operations (continued)

(Dollars in thousands, except per share amounts)

As of December 31, 2020, the Bank's common equity tier 1 capital and tier 1 capital ratios were 11.65% and the leverage and total risk-based capital ratios were 8.29% and 12.77%, respectively. This compares with common equity tier 1 capital and tier 1 capital ratios of 11.55% and leverage and total risk-based capital ratios of 8.04% and 12.69%, respectively, as of June 30, 2020. The Bank exceeded minimum regulatory capital requirements to be considered well-capitalized for both periods. Management is not aware of any matters occurring subsequent to December 31, 2020 that would cause the Bank's capital category to change.

### **Critical Accounting Policies**

The financial condition and results of operations for the Corporation presented in the Consolidated Financial Statements, accompanying notes to the Consolidated Financial Statements and Management's Discussion and Analysis of Financial Condition and Results of Operations are, to a large degree, dependent upon the Corporation's accounting policies. The selection and application of these accounting policies involve judgments, estimates and uncertainties that are susceptible to change.

The Corporation has identified the appropriateness of the allowance for loan losses and the evaluation of goodwill for impairment as critical accounting policies and an understanding of these policies is necessary to understand the financial statements. Critical accounting policies are those policies that require management's most difficult, subjective or complex judgments often as a result of the need to make estimates about the effect of matters that are inherently uncertain. Note one (Summary of Significant Accounting Policies - Allowance for Loan Losses and Goodwill and Other Intangible Assets), Note four (Loans), Note six (Goodwill and Intangible Assets) and Management's Discussion and Analysis of Financial Condition and Results of Operation (Critical Accounting Policies and Use of Significant Estimates) of the 2020 Form 10-K provide detail with regard to the Corporation's accounting for the allowance for loan losses. There have been no significant changes in the application of accounting policies since June 30, 2020.

### **Forward-Looking Statements**

Certain statements contained in this Quarterly Report on Form 10-Q, which are not statements of historical fact, constitute forward-looking statements within the meaning of the Private Securities Litigation Reform Act of 1995. The words "may," "continue," "estimate," "intend," "plan," "seek," "will," "believe," "project," "expect," "anticipate" and similar expressions are intended to identify forward-looking statements. These forward-looking statements may involve risks and uncertainties that are difficult to predict, may be beyond our control, and could cause actual results to differ materially from those described in such statements. Any such forward-looking statements are made only as of the date of this report or the respective dates of the relevant incorporated documents, as the case may be, and, except as required by law, we undertake no obligation to update these forward-looking statements to reflect subsequent events or circumstances. The COVID-19 pandemic is affecting us, our customers, employees, and third-party service providers, and the ultimate extent of the impact on our business, financial position, results of operations, liquidity, and prospects is uncertain. Other risks and uncertainties that could cause actual results for future periods to differ materially from those anticipated or projected include, but are not limited to:

- changes in local, regional and national economic conditions becoming less favorable than we expect, resulting in, among
- other things, high unemployment rates, a deterioration in credit quality of our assets or debtors being unable to meet their obligations;
- changes in the level of non-performing assets and charge-offs;
- declining asset values impacting the underlying value of collateral;
- sustained low market interest rates could result in a decline in the net interest margin and net interest income;
- unanticipated changes in our liquidity position, including, but not limited to, changes in the cost of liquidity and our ability to find alternative funding sources;
- the effect of changes in laws and regulations (including laws and regulations concerning taxes, banking, securities and insurance) with which we must comply;
- changes in consumer spending, borrowing and savings habits;
- changes in accounting policies, rules and interpretations that may come as a result of COVID-19 or otherwise;
- the effects of, and changes in, trade, monetary and fiscal policies and laws, including interest rate policies of the Federal Reserve Board;
- competitive pressures on product pricing and services;
- breaches of security or failures of our technology systems due to technological or other factors and cybersecurity threats;
- changes in the reliability of our vendors, internal control systems or information systems; and
- our ability to attract and retain qualified employees.

### Item 4 – Controls and Procedures

## **Evaluation of Disclosure Controls and Procedures**

As of the end of the period covered by the report, an evaluation was performed under the supervision and with the participation of the Corporation's management, including the Chief Executive Officer and Chief Financial Officer, of the effectiveness of the design and operation of the Corporation's disclosure controls and procedures pursuant to Exchange Act Rule 13a-15e. Based on the evaluation, the Chief Executive Officer and Chief Financial Officer concluded that the Corporation's disclosure controls and procedures were effective as of December 31, 2020.

## **Changes in Internal Controls Over Financial Reporting**

There have not been any changes in the Corporation's internal control over financial reporting that occurred during the Corporation's last quarter that have materially affected, or are reasonably likely to materially affect, the Corporation's internal control over financial reporting.

### **PART II – OTHER INFORMATION**

<u>Item 1 – Legal Proceedings</u>

None

Item 2 – Unregistered Sales of Equity Securities and Use of Proceeds

None

Item 3 – Defaults Upon Senior Securities

None

Item 4 – Mine Safety Disclosures

Not Applicable

Item 5 – Other Information

None

Item 6 – Exhibits

Exhibit

Number Description

Exhibit 10.11 Form of Salary Continuation Agreement. Reference is made to Form 8-K of the Corporation filed December

29, 2020, which is incorporated herein by reference.

Exhibit 10.12 Branch Purchase and Assumption Agreement entered into with CFBank National Association on December

29, 2020.

Exhibit 31.1 Rule 13a-14(a)/15d-14(a) Certification of Chief Executive Officer.

Exhibit 31.2 Rule 13a-14(a)/15d-14(a) Certification of Chief Financial Officer.

Exhibit 32.1 <u>Certification of Chief Executive Officer and Chief Financial Officer Pursuant to 18 U.S.C. Section 1350, as</u>

Adopted Pursuant to Section 906 of the Sarbanes Oxley Act of 2002.

Exhibit 101 The following materials from Consumers Bancorp, Inc.'s Form 10-Q Report for the quarterly period ended

December 31, 2020, formatted in XBRL (Extensible Business Reporting Language) include: (1) Unaudited Consolidated Balance Sheets, (2) Unaudited Consolidated Statements of Income, (3) Unaudited Consolidated Statements of Comprehensive Income, (4) Unaudited Condensed Consolidated Statement of Changes in Shareholders' Equity, (5) Unaudited Condensed Consolidated Statements of Cash Flows, and (6) the Notes

to Unaudited Condensed Consolidated Financial Statements.

## **SIGNATURES**

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

CONSUMERS BANCORP, INC.

(Registrant)

Date: February 12, 2021 /s/ Ralph J. Lober

Ralph J. Lober, II

President & Chief Executive Officer

(principal executive officer)

Date: February 12, 2021 /s/ Renee K. Wood

Renee K. Wood

Chief Financial Officer & Treasurer

(principal financial officer)

34

## BRANCH PURCHASE AND ASSUMPTION AGREEMENT

by and between

## **CONSUMERS NATIONAL BANK**

and

## CFBANK, NATIONAL ASSOCIATION

December 29, 2020

## BRANCH PURCHASE AND ASSUMPTION AGREEMENT

## **TABLE OF CONTENTS**

1.	DEFI	NITIONS	1
2.	PURG	CHASE AND ASSUMPTION	7
	2.01	Purchase and Sale of Assets	7
	2.02	Transfer of Assets	7
	2.03	Acceptance and Assumption	8
	2.04	Payment of Funds	9
	2.05	Loan Participations	11
3.	CON	DUCT OF THE PARTIES PRIOR TO CLOSING	12
	3.01	Covenants of Seller	12
	3.02	Covenants of Purchaser	15
	3.03	Covenants of Both Parties	16
	3.04	Review and Update of Branch Loans	16
	3.05	Review and Selection of Residential Loans	17
4.	REPF	RESENTATIONS AND WARRANTIES	17
	4.01	Representations and Warranties of Seller	17
	4.02	Representations and Warranties of Purchaser	22
5.	ACTI	ONS RESPECTING EMPLOYEES AND EMPLOYEE BENEFIT PLANS	23
	5.01	General	23
	5.02	Action to be Taken by Seller	25
	5.03	Actions to be Taken by Seller and Purchaser	25
	5.04	Continuing Rights to Amend or Terminate	25
6.	CON	DITIONS PRECEDENT TO CLOSING	25
	6.01	Conditions to Seller's Obligations	25
	6.02	Conditions to Purchaser's Obligations	26
	6.03	Non-Satisfaction of Conditions Precedent	27
7.	CLOS	SING	27
	7.01	Closing and Closing Date	27
	7.02	Seller's Actions at Closing	27
	7.03	Purchaser's Actions at Closing	28
	7.04	Methods of Payment	29
	7.05	Effectiveness of Closing	29
		-i-	

CERT	TAIN TRANSITIONAL MATTERS	29
8.01	Customer Communications	29
8.02	General	30
8.03	Post-Closing Adjustments	33
8.04	Post-Closing Reconciliation	34
GENI	ERAL COVENANTS AND INDEMNIFICATION	35
9.01	Information After Closing	35
9.02	Confidentiality Obligations of Purchaser	35
9.03	Confidentiality Obligations of Seller	35
9.04	Indemnification by Seller	36
9.05	Indemnification by Purchaser	36
9.06	Claims for Indemnity	36
9.07	Limitations on Indemnification	37
9.08	Tax Benefits; Insurance Proceeds	37
9.09	Solicitation of Customers by Purchaser Prior to Closing	37
9.10	Solicitation of Customers by Seller After Closing	38
9.11	Further Assurances	38
TERN	MINATION.	38
10.01	Termination by Mutual Agreement	38
10.02	Termination by Seller	38
10.03	Termination by Purchaser	39
10.04	Effect of Termination	39
REPU	JRCHASE OF DEFECTIVE LOANS	40
11.01	Notice of Defective Loans to Seller	40
11.02	Right to Cure	40
11.03	Obligation to Repurchase	40
11.04	Reassignment to Seller	40
11.05	General	40
11.06	Limitation on Repurchase Obligation	40
MISC	CELLANEOUS PROVISIONS	41
12.01	Tax Matters	41
12.02	Survival of Representations, Warranties and Covenants	41
12.03	Limitation and Disclaimer of Representations and Warranties	42
12.04	Waivers	42
12.05	Notices	42
12.06	Parties in Interest; Assignment; Amendment	43
	-ii-	
	8.01 8.02 8.03 8.04 GENI 9.01 9.02 9.03 9.04 9.05 9.06 9.07 9.08 9.09 9.10 10.03 10.04 REPU 11.01 11.02 11.03 11.04 11.05 11.06 MISC 12.01 12.02 12.03 12.04 12.05	8.02 General 8.03 Post-Closing Adjustments 8.04 Post-Closing Reconciliation GENERAL COVENANTS AND INDEMNIFICATION 9.01 Information After Closing 9.02 Confidentiality Obligations of Purchaser 9.03 Confidentiality Obligations of Seller 9.04 Indemnification by Seller 9.05 Indemnification by Purchaser 9.06 Claims for Indemnify 9.07 Limitations on Indemnification 9.08 Tax Benefits; Insurance Proceeds 9.09 Solicitation of Customers by Purchaser Prior to Closing 9.10 Solicitation of Customers by Purchaser Prior to Closing 9.11 Further Assurances TERMINATION. 10.01 Termination by Mutual Agreement 10.02 Termination by Purchaser 10.04 Effect of Termination REPURCHASE OF DEFECTIVE LOANS 11.01 Notice of Defective Loans to Seller 11.02 Right to Cure 11.03 Obligation to Repurchase 11.04 Reassignment to Seller 11.05 General 11.06 Limitation on Repurchase Obligation MISCELLANEOUS PROVISIONS 12.01 Tax Matters 12.02 Survival of Representations, Warranties and Covenants 12.03 Limitation and Disclaimer of Representations and Warranties 12.04 Waivers 12.05 Notices

12.07 Headings	43
12.08 Terminology	43
12.09 Flexible Structure	43
12.10 Press Releases	43
12.11 Entire Agreement	44
12.12 Governing Law and Jurisdiction	44
12.13 Counterparts	44

## **EXHIBITS**

EXHIBIT 1.23	DEPOSIT ACCOUNTS
EXHIBIT 1.63	REAL ESTATE
EXHIBIT	LIST OF FURNITURE, FIXTURES, AND EQUIPMENT
2.02(a)	
EXHIBIT	CONTRACTS AND AGREEMENTS
2.02(c)	
EXHIBIT	ACQUIRED LOANS
2.02(d)	
EXHIBIT	SUBORDINATED DEBT SECURITIES
2.02(e)	
EXHIBIT	FORM OF PRELIMINARY SETTLEMENT STATEMENT
2.04(e)	
	FORM OF PARTICIPATION AGREEMENT
EXHIBIT	CLOSING CERTIFICATE – PURCHASER
6.01(e)	
EXHIBIT	CLOSING CERTIFICATE – CFBANK, NATIONAL ASSOCIATION
6.02(e)	
EXHIBIT	BILL OF SALE
7.02(b)	
EXHIBIT	ASSIGNMENT, TRANSFER AND APPOINTMENT OF SUCCESSOR CUSTODIAN FOR INDIVIDUAL
7.02(e)	RETIREMENT ACCOUNT
EXHIBIT	ASSIGNMENT OF MORTGAGES
7.02(f)	ACCIONMENT OF BIGHT TITLE AND DITERECT
EXHIBIT	ASSIGNMENT OF RIGHT, TITLE AND INTEREST
7.02(g) EXHIBIT	LIMITED WADDANTY DEED
7.02(k)	LIMITED WARRANTY DEED
EXHIBIT	INSTRUMENT OF ASSUMPTION
7.03(a)	INSTRUMENT OF ASSUMPTION
( )	FORM OF FINAL SETTLEMENT STATEMENT
EAHDH 6.03	TOKINI OF THAL SETTLEMENT STATEMENT

### BRANCH PURCHASE AND ASSUMPTION AGREEMENT

THIS BRANCH PURCHASE AND ASSUMPTION AGREEMENT (this "<u>Agreement</u>"), is made as of the 29th day of December, 2020, by and between Consumers National Bank, a national banking association chartered under the laws of the United States of America with its principal office located at 614 E. Lincoln Way, Minerva, Ohio 44657 ("<u>Purchaser</u>"), and CFBank, National Association, a national banking association chartered under the laws of the United States of America with its principal office located at 7000 N. High St., Worthington, Ohio 43085 ("<u>Seller</u>").

WHEREAS, Seller desires to sell certain assets related to two of its branch locations to Purchaser, wherein Purchaser would assume certain liabilities associated therewith;

WHEREAS, Purchaser has determined that it would be in its best interest to acquire such assets and assume certain liabilities associated therewith;

WHEREAS, Purchaser desires to purchase and Seller desires to sell certain loans and interests in loans and certain subordinated debt obligations of other persons; and

WHEREAS, Seller and Purchaser wish to provide for certain undertakings, conditions, warranties, representations and covenants in connection with the transactions contemplated hereby.

NOW, THEREFORE, in consideration of the mutual promises hereinafter contained and other good and valuable consideration, Seller and Purchaser hereby agree as follows:

### 1. **DEFINITIONS**.

Capitalized terms used in this Agreement will have the meanings given to them in this Article 1 unless defined elsewhere in this Agreement.

- 1.01 "Accounts" means Acquired Loans and Deposit Accounts.
- 1.02 "ACH" means automated clearing house.
- 1.03 "Acquired Loans" has the meaning assigned to such term in Section 2.02(d) hereof.
- 1.04 "Acquisition" has the meaning assigned to such term in Section 2.01 hereof.
- 1.05 "Acquisition Consideration" has the meaning assigned to such term in Section 2.04(a) hereof.
- 1.06 "Affiliate" means, with respect to any Person, any other Person directly or indirectly controlling, controlled by or under common control with such Person. For purposes of this definition, "control" means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of a Person, whether through a management position, the ownership of voting securities, by contract or otherwise.
  - 1.07 "Agreement" has the meaning assigned to such term in the introductory paragraph to this Agreement.
- 1.08 "Applicable Law" means any federal, state or local statute, law, ordinance, rule, administrative interpretation, regulation, order, code, writ, injunction, directive, judgment, decree, or other requirement of, or agreement with, any Governmental Authority applicable to the Person or any of its Affiliates or any of their respective operations, properties, assets, officers, directors, partners, employees, consultants or agents (in connection with such officers', directors', partners', employees', consultants' or agents' activities on behalf of such Person or any of its Affiliates).

- 1.09 "Assets" has the meaning assigned to such term in Section 2.02 hereof.
- 1.10 "Assumed Liabilities" has the meaning assigned to such term in Section 2.03(b) hereof.
- 1.11 "Bona Fide Prospective Purchaser" has the same meaning as set forth in 42 U.S.C. §9601(40), as amended.
- 1.12 "Branches" means the facilities located at 565 Main Street, Wellsville, Ohio 43968 and 49028 Foulks Drive, Calcutta, Ohio 43920.
- 1.13 "Branch Employees" means all employees of Seller assigned to the Branches as of the Closing Date (including full-time and part-time employees and any employees then on temporary leave of absence for military service, medical or disability or other approved purposes).
  - 1.14 "Branch Loans" has the meaning assigned to such term in Section 2.02(d) hereof.
- 1.15 "<u>Burdensome Regulatory Condition</u>" means any condition or restriction set forth in a regulatory approval or authorization that could reasonably be expected to have a material adverse effect on Purchaser or the Branches.
- 1.16 "Business Day" means any day other than a Saturday, Sunday or a day on which Seller or Purchaser is closed in accordance with the laws of the State of Ohio or the United States of America. Any action, notice or right which is to be taken or given or which is to be exercised or lapse on or by a given date which is not a Business Day may be taken, given or exercised, and will not lapse, until the next Business Day following.
- 1.17 "<u>CERCLA</u>" means the Comprehensive Environmental Response, Compensation and Liability Act, 42 U.S.C. §9601 et. seq., as amended.
  - 1.18 "Closing" and "Closing Date" have the meanings assigned to such terms in Section 7.01 hereof.
- 1.19 "COBRA" means the requirements of Part 6 of Subtitle B of Title I of ERISA and Section 4980B of the Code and of any similar state law.
  - 1.20 "Commercial Loans" has the meaning assigned to such term in Section 2.05 hereof.
  - 1.21 "Cure Period" has the meaning assigned to such term in Section 11.02 hereof.
  - 1.22 "<u>Defective Loan</u>" has the meaning assigned to such term in Section 11.01 hereof.
- 1.23 "Deposit Accounts" means all deposit accounts attributable to the Branches of any type or character, including, without limitation, passbook accounts, statement savings accounts, money market accounts, checking accounts, NOW accounts, all IRAs that Purchaser may legally assume, and certificates of deposit (excluding QwickRate certificates of deposit) and will include any overdrawn Deposit Account not otherwise covered by an overdraft protection plan (unless such overdraft is unpaid for a period exceeding thirty (30) days prior to the close of business on the Closing Date). A list of the Deposit Accounts as of November 30, 2020 is set forth in Exhibit 1.23 hereto. Notwithstanding anything to the contrary contained herein, the "Deposit Accounts" shall not include (a) any deposit accounts designated as closed prior to the Closing, (b) any deposit accounts constituting security for any loans of Seller other than the Acquired Loans, (c) any deposits that are "Brokered Deposits" as that term is defined in 12 CFR 337.6, (d) any Deposit Account owned or maintained by any Affiliate of Seller or (e) any deposits whose primary owner is located outside of the Indentified Market.

- 1.24 "Deposit Liabilities" means Seller's obligations, duties and liabilities relating to all Deposit Accounts, which, as reflected on the books of Seller as of the close of business on the Closing Date, are attributable to the Branches. The "obligations, duties and liabilities" referred to in the immediately preceding sentence include, without limitation, the obligation to pay and otherwise process all Deposit Liabilities in accordance with Applicable Law and their respective contractual terms, any interest owed but not yet credited to any Deposit Accounts, and the duty to supply all applicable reporting forms for post-Closing periods including, without limitation, Internal Revenue Service reporting forms, relating to the Deposit Liabilities, but will not in any event include any claim or other liability arising out of or from any breach of contract, breach of warranty, tort, infringement or violation of law, rule or regulation relating to the origination or administration of any Deposit Account or Deposit Liability prior to the close of business on the Closing Date. With regard to each IRA included within the Deposit Liabilities, Purchaser will also assume the plan pertaining thereto and the custodial arrangement in connection therewith.
  - 1.25 "Disputed Items" has the meaning assigned to such term in Section 8.03(b)(i) hereof.
- 1.26 "Employee Benefit Plan" means "employee benefit plan" as defined in Section 3(3) of ERISA and any other bonus, incentive compensation, deferred compensation, profit sharing, stock option, stock appreciation right, stock bonus, stock purchase, employee stock ownership, savings, severance, change in control, supplemental unemployment, layoff, salary continuation, retirement, severance, pension, health, life insurance, disability, accident, group insurance, vacation, holiday, sick leave, fringe benefit or welfare plan, and any other employment, consulting, employee compensation or benefit plan, agreement, policy, practice, commitment, contract or understanding (whether qualified or nonqualified, currently effective or terminated, written or unwritten) related thereto (i) that is or was maintained or contributed to by Seller or any entity or trade or business (whether or not incorporated) that together with Seller is treated as a single employer under any of Sections 414(b), (c), (m) or (o) of the Code or Section 4001(a)(14) of ERISA (each, an "ERISA Affiliate"), (ii) with respect to which Seller or any of its ERISA Affiliates has or may have any liability or liabilities, and/or (iii) provides benefits, or describes policies or procedures applicable, to any current or former employee, officer, director, consultant, service provider or contractor of Seller or an ERISA Affiliate, regardless of how (or whether) liabilities for the provision of benefits are accrued or assets are acquired or dedicated with respect to the funding thereof.
- 1.27 "Environmental Laws" means all applicable laws, statutes, ordinances, rules, regulations, orders, or determinations of any federal, state, or local government or agency thereof pertaining to the environment currently in effect in any jurisdiction in which a Branch is conducting business, or where any of the Assets are located, or where any hazardous substances generated by or disposed of by the Branch are located. "Environmental Laws" will include, but not be limited to, the Clean Air Act, 42 U.S.C. §7401 et seq., as amended, CERCLA, the Federal Water Pollution Control Act, 33 U.S.C. §1251 et seq., as amended, RCRA, the Safe Drinking Water Act, 42 U.S.C. §300f et seq., as amended, the Toxic Substances Control Act, 15 U.S.C. §2601 et seq., as amended, and all other laws, statutes, ordinances, rules, regulations, orders and determinations of any federal, state, or local government or agency thereof relating to (a) the control of any hazardous substance or protection of the air, water or land, (b) solid, gaseous or liquid waste generation, or disposal and (c) exposure to hazardous, toxic or other substances regulated by law. The terms "hazardous substance," "release" and "threatened release" have the meanings specified in CERCLA, and the terms "solid waste" and "disposal" (or "disposed") have the meanings specified in RCRA.

- 1.28 "Environmental Problem" has the meaning assigned to such term in Section 3.01(f) hereof.
- 1.29 "ERISA" means the Employee Retirement Income Security Act of 1974, as amended.
- 1.30 "Excluded Items" has the meaning assigned to such term in Section 2.02 hereof.
- 1.31 "Excluded Liabilities" has the meaning assigned to such term in Section 2.03(b) hereof.
- 1.32 "Excluded Loan" has the meaning assigned to such term in Section 3.04(a) hereof.
- 1.33 "<u>Final Payment Amount</u>" means the amount of such payment determined in accordance with Section 8.03 and <u>Exhibit</u> 8.03 hereto.
  - 1.34 "GAAP" means generally accepted accounting principles in the United States of America, consistently applied.
- 1.35 "Governmental Authority" means any government or governmental regulatory body thereof, or political subdivision thereof, whether foreign, federal, state or local, or any agency, instrumentality or authority thereof, or any court or arbitrator (public or private).
- 1.36 "<u>Hazardous Materials</u>" means (i) any petroleum or petroleum products, asbestos in any form, and polychlorinated biphenyls and (ii) any chemicals, materials or substances, whether solid, liquid or gas defined as or included in the definition of "contaminant," "dangerous goods," "pollutant," "hazardous substances," "hazardous wastes," "hazardous materials," "extremely hazardous wastes," "restricted hazardous wastes," "toxic substances," or "toxic pollutants" under any applicable Environmental Law.
- 1.37 "Identified Market" means the following counties: Ashtabula, Belmont, Carroll, Columbiana, Cuyahoga, Harrison, Jefferson, Mahoning, Medina, Monroe, Portage, Stark, Summit, Tuscarawus, and Trumbull, Ohio; Beaver, Green, Lawrence, Mercer, and Washington, Pennsylvania; and Brooke, Hancock, and Ohio, West Virginia
  - 1.38 "Inclearing Period" has the meaning assigned to such term in Section 8.04(a) hereof.
  - 1.39 "IRAs" means Individual Retirement Accounts.
- 1.40 "Knowledge" means, with respect to Seller, the actual knowledge of Seller's President and Chief Executive Officer and Seller's Senior Vice President and Senior Financial Officer.
- 1.41 "<u>Legal Proceeding</u>" means any judicial, administrative or arbitral actions, suits, mediations, proceedings (public or private) or claims (including counterclaims) by or before a Governmental Authority, including any civil, criminal or investigative actions.
- 1.42 "<u>Legal Requirement</u>" means any federal, state, or local law, constitution, ordinance, code, regulation, statute or treaty, as in effect as of or prior to the date of this Agreement that is applicable to Seller's operation of its business, or Seller's use or ownership of the Assets.

- 1.43 "<u>Lien</u>" means any lien, easement, restriction, pledge, charge, encumbrance, security interest, mortgage, deed of trust, lease, option or other adverse claim of any nature whatsoever and of any kind or description.
  - 1.44 "Loan Documents" has the meaning assigned to such term in Section 4.01(n)(i) hereof.
  - 1.45 "Loan Files" has the meaning assigned to such term in Section 4.01(n)(i) hereof.
- 1.46 "Loss" or "Losses" means the amount of losses, liabilities, damages (including forgiveness or cancellation of obligations) and reasonable expenses (including reasonable expenses of investigation and reasonable attorneys' fees and expenses in connection with any action, suit or proceeding) actually incurred, suffered or sustained by the indemnified party or its Affiliates in connection with the matters described in Sections 9.04 and 9.05, less any amount actually recovered by the indemnified party or its Affiliates under insurance policies (net of all third party costs and expenses incurred in pursuing any such insurance recovery, including, but not limited to, those relating to deductibles and actual premium adjustments directly resulting from such damage, loss, liability or expense) or third party indemnification obligations or other actual recovery directly related to such losses or Tax benefit available to an Indemnified Party (or any Affiliate) to the extent such Tax benefit survives the payment of indemnification.
- 1.47 "Net Book Value" means, with respect to the Real Property and the furniture, fixtures, equipment and other fixed assets listed on Schedule 2.02(a), the values at which such Assets are carried on Seller's general ledger as of the month end immediately preceding the Closing Date.
  - 1.48 "New Branch Loans" has the meaning assigned to such term in Section 3.04(b) hereof.
  - 1.49 "Nonperforming Loans" has the meaning assigned to such term in Section 2.02(d) hereof.
  - 1.50 "Notice of Disagreement" has the meaning assigned to such term in Section 8.03(b)(i) hereof.
  - 1.51 "Paper Items" has the meaning assigned to such term in Section 8.04(a) hereof.
  - 1.52 "Participation Period" has the meaning assigned to such term in Section 2.05 hereof.
  - 1.53 "Participations" has the meaning assigned to such term in Section 2.05 hereof.
  - 1.54 "Permitted Encumbrances" has the meaning assigned to such term in Section 3.02(c).
- 1.55 "Person" will mean any individual, corporation, partnership, limited liability company, association, joint venture, trust, estate or unincorporated organization or Governmental Authority.
- 1.56 "<u>Post-Closing Tax Period</u>" means each taxable period beginning after the Closing Date and the portion beginning on the day after the Closing Date for any taxable period that includes (but does not end on) the Closing Date.
  - 1.57 "Pre-Closing Statement" has the meaning assigned to such term in Section 2.04(e) hereof.
- 1.58 "<u>Pre-Closing Tax Period</u>" means a taxable period ending on or before the Closing Date and the portion through the end of the Closing Date for any taxable period that includes (but does not end on) the Closing Date.

- 1.59 "Purchaser" has the meaning assigned to such term in the introductory paragraph to this Agreement.
- 1.60 "Purchaser's Notice" has the meaning assigned to such term in Section 11.01 hereof.
- 1.61 "Release" with respect to any Hazardous Materials includes, but is not limited to, any release, deposit, discharge, emission, leaking, leaching, spilling, seeping, migrating, injecting, pumping, pouring, emptying, escaping, dumping, disposing or growth, proliferation, reproduction, or dispersal or other movement or production of Hazardous Materials into or through the environment.
  - 1.62 "RCRA" means the Resource Conservation and Recovery Act of 1976, as amended.
- 1.63 "Real Estate" means the real property described in Exhibit 1.63 hereto, together with all improvements thereon and thereto.
  - 1.64 "Repurchase Price" has the meaning assigned to such term in Section 11.03 hereof.
  - 1.65 "Residential Loan" has the meaning assigned to such term in Section 3.05 hereof.
  - 1.66 "Seller" has the meaning assigned to such term in the introductory paragraph to this Agreement.
  - 1.67 "Straddle Period" has the meaning assigned to such term in Section 12.01(c) hereof.
  - 1.68 "Subordinated Debt Securities" has the meaning assigned to such term in Section 2.02(e) hereof.
- 1.69 "<u>Subsidiary</u>" means a corporation, partnership, limited liability company, joint venture or other business organization more than 50% of the voting securities of which are beneficially owned or controlled by the indicated parent of such entity.
- 1.70 "Tax" or "Taxes" means all taxes, assessments, charges, duties, fees, levies, imposts or other similar charges imposed by a Governmental Authority, including without limitation all income, franchise, profits, capital gains, estimated, capital stock, transfer, gross receipts, sales, use, service, occupation, ad valorem, property, excise, severance, windfall profits, premium, stamp, license, payroll, employment, social security, unemployment, disability, environmental (including taxes under Section 59A of the Code), alternative minimum, add-on, value-added, withholding and other taxes, assessments, charges, duties, fees, levies, imposts or other similar charges of any kind whatsoever (whether payable directly or by withholding and whether or not requiring the filing of a Tax Return), including any interest, penalties or additions to tax or additional amounts in respect of the foregoing, imposed, assessed or collected by or under the authority of any Governmental Authority.
- 1.71 "<u>Tax Return</u>" means any return (including any information return), report, statement, schedule, notice, form, or other document or information filed with or submitted to, or required to be filed with or submitted to, any Governmental Authority in connection with the determination, assessment, collection, or payment of any Tax or in connection with the administration, implementation, or enforcement of or compliance with any Applicable Law relating to any Tax.
  - 1.72 "<u>Transfer Taxes</u>" has the meaning assigned to such term in Section 12.01(a) hereof.

1.73 "Unresolved Changes" has the meaning assigned to such term in Section 8.03(b)(iii) hereof.

## 2. PURCHASE AND ASSUMPTION.

- 2.01 <u>Purchase and Sale of Assets</u>. At the Closing, Purchaser will purchase from Seller, and Seller will sell to Purchaser, the "Assets" as such term is defined in Section 2.02 below (and subject to the exceptions stated therein), pursuant to the terms and conditions set forth herein and subject to the exceptions set forth herein. The acquisition by Purchaser from Seller of the Assets on the terms and conditions set forth herein is sometimes referred to herein as the "Acquisition".
- 2.02 <u>Transfer of Assets</u>. Subject to the terms and conditions of this Agreement, Seller will assign, transfer, convey and deliver to Purchaser, as of the Closing, the following assets, free and clear of any and all Liens (collectively, the "<u>Assets</u>"):
- (a) Owned Real Estate, Furniture, Fixtures and Equipment. All right, title and interest in and to the Real Estate, together with all of Seller's rights in and to all improvements thereon, and all right, title and interest in and to all furniture, fixtures and equipment located and used at the Branches and owned or (to the extent of Seller's interest as lessee) leased by Seller (including any manufacturer's warranties thereon which are in effect on the Closing Date and which are assignable to the Purchaser), as listed on (and subject to any exceptions noted in) Exhibit 2.02(a) attached hereto;
- (b) <u>Cash on Hand</u>. All of Seller's vault and teller cash, petty cash and other cash on hand at the Branches, ATM cash relating to the Branches, and cash equivalents relating to the Branches as of the close of business on the Closing Date (collectively, the "<u>Cash on Hand</u>");
- (c) <u>Contracts and Agreements</u>. All right, title and interest of Seller in and to the fixed asset leases and maintenance, service and other contracts pertaining exclusively to the Branches, including, without limitation, those contracts listed on <u>Exhibit 2.02(c)</u> attached hereto;
- (d) <u>Acquired Loans</u>. All right, title and interest of Seller in and to (i) all of the loans attributable to the Branches identified on <u>Exhibit 2.02(d)</u> attached hereto (collectively, the "<u>Branch Loans</u>"), as such <u>Exhibit 2.02(d)</u> may be modified to reflect the exclusion and/or addition of Branch Loans pursuant to Section 3.04 hereof prior to Closing and (ii) those Residential Loans of Seller selected by Purchaser pursuant to Section 3.05, in each case, that remain outstanding as of the Closing (collectively, the "<u>Acquired Loans</u>"), together with all security thereon and collateral and rights in relation thereto (including servicing rights), at their respective then outstanding principal amounts, which will reflect any write-downs arising from adverse classifications, together with accrued interest receivable thereon and associated accrued late fees, but excluding loan loss reserves and general reserves; <u>provided</u>, <u>however</u>, that unless Purchaser otherwise consents in writing, the Acquired Loans will not include any loans that, as of the Closing Date, (x) are on a non-accrual basis, (y) are ninety (90) days or more past due, and/or (x) are in foreclosure (each, a "<u>Nonperforming Loan</u>");
- (e) <u>Subordinated Debt Securities</u>. All right, title and interest of Seller in and to the subordinated debt securities listed on <u>Schedule 2.02(e)</u> attached hereto (collectively, the "<u>Subordinated Debt Securities</u>"), including but not limited to all rights of Seller under note purchase agreements or similar agreements or documents related to the Subordinated Debt Securities;
  - (f) <u>Safe Deposit Boxes</u>. All safe deposit boxes located at the Branches and the leases and contracts related thereto;

- (g) <u>Branch Records, Etc.</u> All of (i) the books and records (or copies thereof) relating to one or both of the Branches which are (A) maintained in the ordinary course of business at the Branch, (B) reasonably required to comply with Applicable Law and/or (C) necessary for Purchaser's ownership of the Assets and/or the Deposit Liabilities, (ii) the Loan Files for the Acquired Loans, and (ii) manuals, keys, passcodes, security codes and the like related to the Assets or the Deposit Liabilities as of the Closing; and
- (h) <u>Claims</u>. All causes of actions and claims held by Seller and related primarily to any of the Assets (but excluding any refunds of Taxes for Pre-Closing Tax Periods to which Seller is entitled pursuant to Section 12.01(b)).

All other assets except those listed above will be excluded from the Assets, including, without limitation, the following items (the "Excluded Items"):

- (i) any non-assignable licenses, permits, leases or contracts;
- (ii) any software of Seller or any Affiliate of Seller;
- (iii) all rights of Seller under any fixed asset leases, maintenance or service contracts, or similar contracts in effect as of the Closing Date with non-affiliated third-party service providers that are non-assignable or that relate to the operations of other offices of Seller as well as one or both of the Branches;
  - (iv) all loans of Seller other than the Acquired Loans;
  - (v) all mortgage servicing rights for mortgage loans of Seller other than the Acquired Loans;
- (vi) Seller's rights in and to the name "CFBank" and any of Seller's corporate logos, trademarks, trade names, signs, paper stock forms and other supplies containing any such logos, trademarks or trade names, and all other intellectual property of Seller;
  - (vii) all routing numbers of Seller;
  - (viii) trust, brokerage, and other similar relationships; and
  - (ix) Seller signage at the Branches.

Except as otherwise agreed upon by Seller and Purchaser, Seller will, at its own cost, remove or cause to be removed the Excluded Items from the Branches on or prior to the Closing Date, except that Purchaser will remove all signs and other external Excluded Items at Purchaser's expense.

- 2.03 <u>Acceptance and Assumption</u>. Subject to the terms and conditions of this Agreement, on and as of the close of business on the Closing Date, Purchaser will:
- (a) <u>Assets</u>. Receive and accept all of the Assets assigned, transferred, conveyed and delivered to Purchaser by Seller pursuant to this Agreement and, subject to the express representations of Seller in Section 4.01, receive the Assets in "AS IS, WHERE IS" condition, with no implied warranties or guaranties by Seller as to future performance, fitness for particular purpose, merchantability or otherwise; and

- (b) <u>Liabilities</u>. Assume and thereafter pay, perform and discharge, as applicable, in full and as due, the liabilities, duties, responsibilities and obligations of Seller that are expressly set forth below in this Section 2.03(b), to the extent that they are to be paid, performed or discharged, as applicable, from and after the close of business on the Closing Date (collectively, the "<u>Assumed Liabilities</u>"):
  - (i) the Deposit Liabilities, together with all interest accrued and unpaid thereon as of the close of business on the Closing Date, in accordance with the records of Seller relating to the Branches as of the Closing Date, absent manifest error with respect to such records;
  - (ii) all liabilities and obligations of Seller under the Acquired Loans as of the Closing, other than liabilities of Seller for Seller's violations of Applicable Law pertaining to the Acquired Loans or Seller's breach of the terms and conditions of the Acquired Loans; and
  - (ii) all other liabilities and obligations of Seller, if any, under the contracts, leases, licenses and other agreements included in the Assets to the extent (A) attributable or related to the Assets and (B) incurred by Seller in the ordinary course of business.

As used in this Agreement, the term "Excluded Liabilities" means any and all liabilities, duties, responsibilities and/or obligations of Seller of whatever kind or nature other than the Assumed Liabilities. Notwithstanding any provision of this Agreement to the contrary, including without limitation the foregoing provisions of this Section 2.03(b), (A) Seller will retain and pay, perform and discharge, as applicable, all of the Excluded Liabilities, and (B) the term "Liabilities" will not include, and Purchaser will not assume or otherwise be bound by, (1) any liability, duty, responsibility or obligation that results from, arises out of, relates to, is in the nature of, or was caused by any breach of contract, breach of warranty, tort, infringement or violation of law, rule or regulation that occurred at or prior to the close of business on the Closing Date, (2) any Tax liability or obligation of Seller or any of its Affiliates, except as may be otherwise expressly set forth in this Agreement, (3) any liability or obligation of Seller or any of its Affiliates for any Employee Benefit Plan, including any payments under any Employee Benefit Plan, and/or (4) any liability or obligation of Seller under this Agreement.

- 2.04 Payment of Funds. Subject to the terms and conditions hereof, at the Closing:
- (a) <u>Acquisition Consideration</u>. As consideration for the purchase of the Assets, Purchaser will pay Seller a purchase price equal to the sum of the following (the "<u>Acquisition Consideration</u>"):
  - (i) the Net Book Value of the Real Estate and the furniture, fixtures, equipment and other fixed assets shown on <u>Schedule 2.02(a)</u> as of the Closing Date plus Thirty-Five Thousand Dollars (\$35,000);
  - (ii) a premium equal to 1.75% of the amount of the Deposit Accounts calculated based upon the average daily balance of the Deposit Accounts for the thirty (30) calendar days ending five (5) Business Days preceding the Closing Date;
  - (iii) an amount for each Acquired Loan equal to the unpaid principal amount of such Acquired Loan (excluding loan loss and general reserves) as shown on the books of Seller as of the Closing Date, plus accrued and unpaid interest and earned and unpaid fees thereon;

- (iv) an amount for each Subordinated Debt Security equal to its par value (i.e.,the outstanding principal amount plus all accrued but unpaid interest thereon) as of the Closing Date; and
- (v) 100% of the amount of Seller's Cash on Hand at the Branches as of the close of business on the Closing Date.
  - (b) Payment of Deposit Liability Amounts. Seller will pay to Purchaser the amount of the Deposit Liabilities.
- (c) Reimbursement and Proration of Certain Income and Expenses. The Federal Deposit Insurance Corporation premium, and all real estate taxes, utility payments, insurance policy premiums and similar income and expenses relating to the Branches, and other expenses relating to the Assets purchased and Deposit Liabilities assumed by Purchaser hereunder, will be prorated between Seller and Purchaser as of the Closing Date, and each party will reimburse the other party, as applicable, for its pro rata portion of any such item. Any reimbursement payment due from a party to the other party pursuant to the terms of this Section 2.04(c) will be made in the manner specified in Section 7.04 herein.
- (d) Other Fees and Expenses. Except as otherwise provided in this Agreement, each party hereto acknowledges its sole and exclusive responsibility for payment of all costs and other expenses incurred by it in connection with the Acquisition.
- Pre-Closing Statement. Seller will prepare a statement (the "Pre-Closing Statement") as of a date not earlier than ten (10) Business Days prior to the Closing Date reflecting the Assets to be sold and assigned hereunder and the Assumed Liabilities to be transferred and assumed hereunder. Seller will deliver the Pre-Closing Statement to Purchaser not less than three (3) Business Days before the Closing Date. Seller agrees to pay to Purchaser at the Closing, in immediately available funds, the excess, if any, of (i) the amount of Deposit Liabilities assumed by Purchaser pursuant to Section 2.03(b)(i) above and the amount of any prorated items owed by Seller to Purchaser pursuant to Section 2.04(c) above, as reflected by the Pre-Closing Statement, over (ii) the aggregate amount of the Acquisition Consideration computed in accordance with Section 2.04(a) above and the amount of any prorated items owed by Purchaser to Seller pursuant to Section 2.04(c) above, as reflected by the Pre-Closing Statement. Purchaser agrees to pay to Seller at the Closing, in immediately available funds, the excess, if any, of (y) the aggregate amount of the Acquisition Consideration computed in accordance with Section 2.04(a) above and the amount of any prorated items owed by Purchaser to Seller pursuant to Section 2.04(c) above, as reflected by the Pre-Closing Statement, over (z) the amount of Deposit Liabilities assumed by Purchaser pursuant to Section 2.03(b)(i) above and the amount of any prorated items owed by Seller to Purchaser pursuant to Section 2.04(c) above, as reflected by the Pre-Closing Statement. The calculation of the payment of funds at Closing required under this Agreement will be made on a preliminary settlement statement in the form attached as Exhibit 2.04(e) hereto. The payment by Purchaser to Seller or by Seller to Purchaser will be subject to subsequent adjustment based on the final settlement statement in the form attached hereto as Exhibit 8.03, which final settlement statement will be prepared in the same manner as the preliminary settlement statement.

Tax Matters. Seller and Purchaser agree to allocate the Acquisition Consideration and the Assumed Liabilities among the Assets in accordance with their respective fair market values and the principles of Section 1060 of the Internal Revenue Code of 1986, as amended (the "Code"). At the Closing, Seller and Purchaser shall reduce the allocation to writing, subject to post-Closing adjustment to reflect adjustments to the Acquisition Consideration as provided in Section 8.03 of this Agreement. Within thirty (30) days after the determination of the final Acquisition Consideration in accordance with Section 8.03 of this Agreement, Purchaser will provide to Seller Purchaser's proposed allocation of the Acquisition Consideration and the Assumed Liabilities among the Assets as finally determined hereunder. Within thirty (30) days after the receipt of such allocation, Seller will propose to Purchaser any changes to such allocation or otherwise will be deemed to have agreed with such allocation. Seller and Purchaser will use good faith efforts to resolve any such change to such allocation. If Seller and Purchaser are unable to resolve such changes within thirty (30) days of Seller's proposal, then such dispute will be resolved using the same principles as described in Section 8.03(b)(iii). Seller and Purchaser will reduce such allocation to writing, including jointly executing a final, completed Internal Revenue Service Form 8594, and any other forms and statements required to be filed therewith. Seller and Purchaser will file timely any such forms and statements with the Internal Revenue Service. To the extent consistent with Applicable Law, Seller and Purchaser will not file any Tax Return or other documents or otherwise take any position with respect to Taxes which is inconsistent with such allocation of the final Acquisition Consideration and Liabilities; provided, however, nothing herein shall prevent Seller or Purchaser from settling any proposed deficiency or adjustment by any Governmental Authority based on such final allocation, and neither Seller nor Purchaser will be required to litigate any proposed adjustment by any Governmental Authority challenging such final allocation. For the avoidance of doubt, neither Purchaser nor Seller (nor any of their Affiliates) shall take a position inconsistent with the final allocation as a defense or offset to any position taken by a Governmental Authority. In the event that the final allocation is disputed by any Governmental Authority, the party receiving the notice of the contest shall provide the other party with prompt written notice thereof (which, in any event, shall be within thirty (30) days of receiving notice of such contest from the Governmental Authority).

Loan Participations. During the period commencing on the date hereof and ending on the 12-month anniversary of the date hereof (the "Participation Period"), Seller shall provide to Purchaser the opportunity to purchase at least \$15 million in aggregate principal amount of participation interests ("Participations") in commercial and commercial real estate loans ("Commercial Loans") originated by Seller and held in Seller's portfolio. Purchaser and Seller shall cooperate in good faith prior to the Closing to identify and consummate at least \$7.5 million in aggregate principal amount of such Participations to be purchased by Purchaser from Commercial Loans in Seller's then-existing portfolio, and the remaining Participations shall be identified and consummated following the Closing and prior to the end of the Participation Period. All Participations hereunder shall (a) be purchased by Purchaser for an amount equal to (i) the then-outstanding principal amount of the applicable Commercial Loan plus all accrued and unpaid interest thereon, multiplied by (ii) the applicable participation interest therein purchased by Purchaser; (b) continue to be serviced by Seller, who shall retain exclusive rights to contact with the borrower under the applicable Commercial Loan (provided that Purchaser shall have the right to contact any such borrower with which it has an existing relationship with regard to transactions and accounts unrelated to the applicable Commercial Loan); (c) shall not be subject to ongoing service fees payable by Purchaser to Seller; and (d) shall be evidenced by a participation agreement between Seller and Purchaser, substantially the form attached hereto as Exhibit 2.05, containing provisions consistent with this Section 2.05 and other terms and conditions that are customary for loan participation transactions in the industry and market area (and generally consistent with prior participation transactions between Purchaser and Seller). If some of the opportunities Seller offers to Purchaser prior to the Closing under this Section 2.05 are unacceptable to Purchaser, Seller shall continute to offer different opportunities to Purchaser as provided in this Section 2.05 prior to the Closing until the aggregate of Participations consummated is at least \$7.5 million as of the Closing Date, and if some of the opportunities Seller offers to Purchaser after the Closing Date under this Section 2.05 are unacceptable to Purchaser, then Seller shall continue to offer different Participations to Purchaser as provided in this Section 2.05 until the aggregate of Participations consummated is \$15 million. For avoidance of doubt, the transactions contemplated by this Section 2.05 are not included in the defined term "Acquisition" and the Participations are not included in the defined terms "Assets" or "Acquired Loans."

## 3. <u>CONDUCT OF THE PARTIES PRIOR TO CLOSING.</u>

- 3.01 <u>Covenants of Seller</u>. Seller hereby covenants to Purchaser that, from the date hereof until the Closing, Seller will do or cause the following to occur, except as may be (x) otherwise required by a Governmental Authority or by Applicable Law, (y) otherwise expressly provided in this Agreement, or (z) authorized or approved in writing by Purchaser:
- (a) Operation of Branches. Seller will continue to operate the Branches in a manner that is compliant with Applicable Law in all material respects and substantially equivalent to that manner and system of operation employed immediately prior to the date of this Agreement. Seller will use all reasonable efforts to preserve intact its business relationships with depositors, customers and others having business relationships with the Branches, it being understood and agreed that Seller will be permitted to pursue all collection procedures in connection with loans and that Seller may decrease or suspend its marketing efforts in soliciting new loans or deposits for the Branches after the date hereof. Seller may also make changes to its operations of the Branches to the extent such changes are made at all other branches operated by Seller. In addition to the foregoing, Seller will not:
  - (i) set interest rates at the Branches in a manner inconsistent with prudent, safe and sound banking practices, or, without limiting the generality of the foregoing, (A) accept any deposits, without the prior written consent of Purchaser, on which interest rates exceed the applicable rates offered as of the date hereof as set forth on Seller's rate schedule delivered to Purchaser on or prior to the date hereof; (B) change interest rates on existing deposits of the Branches in a manner that is not consistent with the interest rate policies used with respect to Seller's other branches; (C) fail to maintain the Assets listed in Section 2.02(a) in their present state of repair, order and condition, reasonable wear and tear and casualty excepted; (D) invest in any improvements related to the Branches except for commitments previously disclosed to Purchaser in writing, made on or before the date of this Agreement, for replacements of furniture, furnishings and equipment, for normal maintenance and refurbishing, in the ordinary course of business, and for emergency and casualty repairs and replacements; or (E) employ any special promotions to increase deposits at the Branches;
  - (ii) except in the ordinary course of business at the unsolicited request of depositors, (A) cause the transfer of any Deposit Account attributable to either Branch to one of Seller's other branches or (B) cause the transfer of any deposit account attributable to a branch of Seller other than one of the Branches to either Branch;
    - (iii) accept at the Branches any "brokered deposit" as that term is defined in 12 CFR 337.6;
  - (iv) Increase or agree to increase the salary or wage rate or incentive opportunity of any Branch Employee, other than normal salary or wage increases in the ordinary course of business consistent with past practice (however, such increases shall, in no event, increase the aggregate cash compensation for Branch Employees by more than 3% on an annualized basis or for any individual Branch Employee by more than 5%) (except in connection with promotions or changes in job responsibilities in the ordinary course of business); or
  - (v) take any action directly or through any existing or future Subsidiary or Affiliate that is intended or is reasonably likely to result in (A) any of the covenants or conditions to the transactions contemplated hereby not being satisfied in all material respects, (B) a violation of any provision of this Agreement in any material respect or (C) the denial, termination, withdrawal, or unreasonable delay of, any of the regulatory approvals referred to in this Agreement, excluding any such actions as shall be required to comply with applicable Legal Requirements.

- (b) Information Concerning Branches. Seller will use its reasonable best efforts to ensure that Purchaser, its officers, attorneys, accountants and other authorized representatives are furnished or otherwise provided reasonable access to, and permitted to make or cause to be made such reasonable investigation of, information and materials relating to the financial, operational, and legal condition of the Branches as Purchaser reasonably deems necessary or advisable within fifteen (15) days following the date hereof to familiarize itself with the Branches and their business and operations; provided, however, that such access and investigation will be reasonably related to the transactions contemplated hereby and will not interfere unnecessarily with the normal operations of the Branches or Seller; and provided further, that nothing in this Section 3.01(b) will be deemed to require Seller to breach any obligation of confidentiality or to reveal any proprietary information, trade secrets or marketing or strategic plans. In addition to the above information, no later than thirty (30) days after the date of this Agreement, Seller will provide Purchaser with such other information as is necessary in the reasonable opinion of Purchaser to fulfill Purchaser's obligations under Article 8 of this Agreement, including, without limitation, all account information, as of the most recent practicable date, including complete name and address, account master file, ATM account number information, ATM/debit card information with respect to the Deposit Liabilities.
- (c) Insurance; Destruction or Condemnation of Property. Seller will maintain adequate insurance on all of the tangible Assets owned by Seller constituting the Branches to be sold pursuant to this Agreement. Seller will give Purchaser prompt written notice of any material fire or casualty on any of the Assets and any actual or threatened condemnation of all or any part of any of the Real Estate. Upon receipt of such notice, Purchaser may, in its sole and exclusive discretion, within fourteen (14) days of receipt of such notice, elect to: (i) close the Acquisition contemplated by this Agreement, excluding the personal property or real property in question, and deducting from the Acquisition Consideration an amount equal to the Net Book Value thereof, and eliminating the proration related thereto pursuant to Section 2.04(c); (ii) elect to close the Acquisition contemplated by this Agreement, including the personal property or real property in question, in which event Seller will assign, transfer and set over unto Purchaser all right, title and interest of Seller in and to any condemnation award, casualty award, insurance policy, insurance payment, or any manner of payment whatever in any way related to the condemnation or casualty; or (iii) in the event of a condemnation of the Real Estate (and the improvements thereon) or a material fire or casualty affecting either of the Branches, elect to terminate this Agreement, in which event neither party will have any further liability or obligation to the other. For purposes hereof a "material fire or casualty" is one whose cost of repair is reasonably estimated by Purchaser to be more than \$50,000.

## (d) Conversion of Accounts.

(i) Seller will use its reasonable best efforts to cooperate with Purchaser in assuring an orderly transition of ownership of the Assets, including the Acquired Loans, and responsibility for the Assumed Liabilities, including the Deposit Liabilities, assumed by Purchaser pursuant hereto and will, in particular, use its reasonable best efforts to cooperate with Purchaser in the conversion of the accounts and records relating to such Assets and Liabilities to Purchaser's information and data processing systems; provided, however, that in the event that Purchaser requests electronic or machine readable account information that is not already maintained by Seller in the form requested, Seller will pay one half of all third-party, out-of-pocket expenses actually incurred by Seller for obtaining the information in such form up to a maximum of \$10,000, and Purchaser will pay the remainder of such expenses.

- (ii) Not later than fifteen (15) days after the date of this Agreement, Seller will provide to Purchaser, in electronic form, depositor information necessary to permit Purchaser to map its conversion of depositor information to its systems.
- (iii) As early as practicable, and no later than 5:00 p.m., Eastern Time, on the day immediately following the Closing Date, Seller will, at its sole expense, deliver to Purchaser reports containing all pertinent data and descriptive information relating to the Accounts, which reports will constitute Seller's records maintained as of and current through the close of business on the Closing Date with respect to the Accounts. If Purchaser desires tapes containing such information, the tapes will be provided at Purchaser's sole expense.
- (e) <u>Branch Employees</u>. Seller and Purchaser will cooperate in convening a meeting of the Branch Employees to disclose the execution of this Agreement, which meeting will occur no later than three (3) Business Days after the date of this Agreement, and at which Purchaser's representatives will have an opportunity to discuss Purchaser's plans for the Branches and the Branch Employees, subject to Seller's approval of the substance of the discussion, which approval will not be unreasonably withheld.
- Environmental Matters. Purchaser, in its sole discretion and at its own expense, may perform a Phase 1 Environmental Site Assessment (a "Phase 1") for the Real Estate within thirty (30) Business Days after the date of this Agreement. If Purchaser does not perform a Phase 1 within this time frame, Purchaser's right to perform a Phase 1 or later Phase 2 Environmental Site Assessment (a "Phase 2") under this Section 3.01(f) is waived. Within sixty (60) Business Days of this Agreement, Purchaser will provide to Seller written notice of any dissatisfaction with the contents of the Phase 1 and Purchaser's decision to perform a Phase 2, which may include environmental sampling related to the Real Estate. Purchaser will provide Seller a full and complete copy of a Phase 1 or Phase 2 immediately upon Seller's request. If Purchaser requests to perform a Phase 2, Seller will decide whether to consent or not within ten (10) Business Days of receipt of Purchaser's request. If Seller does not consent to a Phase 2, Purchaser may terminate this Agreement without penalty. If Seller consents to a Phase 2, Purchaser will have thirty (30) Business Days to Perform the Phase 2 from the day Seller consents to the Phase 2. If the cost of taking all remedial or corrective actions and measures (i) required by applicable Environmental Law, (ii) recommended in any Phase 1 or Phase 2 report in order to establish the Bona Fide Prospective Purchaser defense under CERCLA or (iii) prudent in light of serious life, health or safety concerns, in the aggregate, exceeds \$50,000 as reasonably estimated by an environmental expert retained for such purpose by Purchaser and reasonably acceptable to Seller, or if the cost of such remedial or corrective actions and measures cannot be so reasonably estimated by such expert to be such amount or less with a reasonable degree of certainty, such circumstances will be deemed an "Environmental Problem" and Purchaser will provide Seller written notice and details of any such Environmental Problem. If an Environmental Problem is identified during the period of time the Phase 2 is prepared, Purchaser, by giving written notice to Seller within five (5) Business Days following Purchaser's receipt of the Phase 2 report, will have the right to (A) terminate this Agreement without penalty or (B) elect to proceed with the Acquisition of the Real Estate. All costs of any Phase 1 or Phase 2 requested or prepared by Purchaser pursuant to this Section 3.01(f) and the cost of restoring the Real Estate to its prior condition if Purchaser has performed a Phase 2, will be at Purchaser's sole cost and expense. In the event of a waiver of the right to perform a Phase 1, Purchaser's termination of this Agreement pursuant to this Section 3.01(f), or a decision by Purchaser to proceed to Closing after receiving a Phase 1, or a Phase 2, Seller will have no liability or obligation to Purchaser for breach of any provision of this Agreement related to environmental matters with respect to the Real Estate.

(g) <u>Cooperation with Title Investigation</u>. Seller shall furnish to Purchaser copies of all deeds and other documents relevant to Seller's title to the Real Estate that are in the possession of Seller.

- (h) Infrastructure Installation. Within twenty (20) Business Days of the date of this Agreement, Seller shall permit Purchaser and/or its representatives reasonable access to review the Branches for the purpose of planning to install automated equipment for use by Branch personnel. Following the receipt of the Regulatory Approvals (except for the expiration of statutory waiting periods), Seller grants to Purchaser a license at each Branch to (i) install voice and data circuits to the main point of entry at the Branch, (ii) install Purchaser's network interface equipment (router/switches), and (iii) extend circuit demarcation points from the main point of entry at the applicable Branch to such network interface equipment (collectively, the "Infrastructure Installation"); it being agreed that, under no circumstance, shall the Infrastructure Installation include the installation or modification of station cabling for equipment, including printers, phones, personal computers and security cameras or otherwise interfere with Seller's existing infrastructure in operation at the Branches through the Closing Date. In the event that this Agreement is terminated pursuant to Article 10 after any Infrastructure Installation has occurred hereunder, Purchaser shall be solely responsible for all costs and expenses of removal of all infrastructure so installed at the Branches and shall fully indemnify Seller for any and all damage to the Branches arising or resulting from the Infrastructure Installation and/or the removal thereof.
- 3.02 <u>Covenants of Purchaser</u>. Purchaser hereby covenants to Seller that, from the date hereof until the Closing, Purchaser will do or cause the following to occur, except as may be otherwise required by a Governmental Authority or by Applicable Law, or except as may be permitted in writing by Seller:
- (a) Certain Applications. As soon as practicable, but in no event later than thirty (30) days after the date hereof, Purchaser will prepare and submit for filing, at no expense to Seller (other than professional fees that Seller may choose to incur in reviewing proposed filings), any and all applications, filings, and registrations with, and notifications to, all federal and state authorities required on the part of Purchaser for the Acquisition to be consummated at the Closing as contemplated in Section 7.01 herein. Thereafter, Purchaser will use its reasonable best efforts to pursue all such applications, filings, registrations and notifications diligently and in good faith, and will promptly file such supplements, amendments and additional information in connection therewith as any such regulatory authorities may require and as Purchaser considers to be reasonably necessary for the Acquisition to be consummated at such Closing. Purchaser will deliver to Seller a copy of each and all of such applications, filings, registrations and notifications (except for any confidential portions thereof), and any supplement, amendment or item of additional information, including all correspondence, in connection therewith (except for any confidential portions thereof), at least three (3) Business Days prior to the date it is filed with such federal and state authorities, and shall consider and incorporate in good faith Seller's comments thereof to the extent practicable. Purchaser will deliver to Seller a copy of each notice, order, opinion, approval, authorization and other item of correspondence received by Purchaser from such federal and state authorities (except for any confidential portions thereof) no later than three (3) Business Days following its receipt by Purchaser and will promptly advise Seller of developments and progress with respect to such matters.
- (b) Other Transactions. Purchaser will not, prior to Closing, directly or through any existing or future Subsidiary or Affiliate take any action that would be in conflict with, or result in the denial, termination, withdrawal, or unreasonable delay of, any of the regulatory approvals referred to in this Agreement.

- (c) <u>Title Investigation</u>. Purchaser will, at Purchaser's sole expense and within thirty (30) days after the date of this Agreement, take all actions necessary to investigate to Purchaser's satisfaction title to the Real Estate. In the event that Purchaser's investigation of the title to the Real Estate does not satisfy Purchaser, in Purchaser's reasonable discretion, that Seller has good and marketable title to the Real Estate, free and clear of all Liens, except with respect to (i) easements and encumbrances of record which do not adversely affect the use of the Real Estate as a bank branch and (ii) any mortgages or other monetary liens upon the Real Estate, which will be discharged by Seller at the time of Closing (collectively, (i) and (ii) are "Permitted Encumbrances"), then Purchaser will provide written notice to Seller within thirty (30) days after the date of this Agreement of such lack of satisfaction and its decision to terminate this Agreement without penalty. If no notice is received by Seller within such thirty (30) day period, Purchaser shall have no further rights under this Section 3.02(c). Seller will have no liability or obligation to Purchaser for breach of any provision of this Agreement arising from failure of title to the Real Estate to satisfy Purchaser to the extent of any defect, lien or encumbrance that exists at the time of such investigation of the title to the Real Estate, whether or not Purchaser terminates the Agreement or proceeds with the Acquisition.
- 3.03 Covenants of Both Parties. Except as otherwise specifically provided in this Agreement, Seller hereby covenants to Purchaser, and Purchaser hereby covenants to Seller, that from the date hereof until the Closing, such party will use commercially reasonable efforts to obtain all consents, approvals, permits, or authorizations which are required to be obtained pursuant to any federal or state law, or any federal or state regulation thereunder, for or in connection with the transactions described and contemplated in this Agreement, including, without limitation, any regulatory consents, approvals, permits or authorizations which are required to be obtained in order to consummate the Acquisition.

## 3.04 Review and Update of Branch Loans.

- (a) Excluded Loans. Commencing upon the execution of this Agreement and continuing until the Closing Date, Purchaser will be provided an opportunity to review and examine all of the Branch Loans and all Loan Files relating thereto, which review and examination by Purchaser shall be coordinated between Seller and Purchaser so as not to interfere unnecessarily with the normal operations of the Branches or Seller. In the event that, at any time prior to the Closing Date, Purchaser determines that it does not wish to purchase and assume any Branch Loan, whether because such Branch Loan is nonperforming, does not meet Purchaser's underwriting criteria, or otherwise, then Purchaser may notify Seller in writing that such loan shall be considered an "Excluded Loan" and shall not be included in the Acquired Loans.
- (b) New Branch Loans. With respect to loans attributable to the Branches that are proposed to be made by Seller after the date hereof and prior to the Closing ("New Branch Loans"), Seller will only make New Branch Loans that are consistent with Seller's historical underwriting practices. In the case of each proposed Branch Loan in the original principal amount greater than \$100,000, Seller will provide Purchaser an opportunity to review Seller's Loan File relating to such proposed New Branch Loan at or prior to the time of the making of the loan. Seller will provide information to Purchaser with respect to all other New Branch Loans in the ordinary course of due diligence with respect to the Acquisition. If any such New Branch Loan is made, Purchaser may, at its option and in its sole discretion, by notice to Seller prior to the Closing Date, accept any such New Branch Loans to be included as a Branch Loan in the Acquired Loans. If Purchaser has accepted any such New Branch Loan as of the Closing, such New Branch Loan which will become an Acquired Loan subject to all terms and conditions of this Agreement.
- (c) <u>Updated List of Branch Loans</u>. At the Closing, Seller shall deliver to Purchaser an updated Exhibit 2.02(d) containing an updated list of the Branch Loans, excluding any Excluded Loans under Section 3.04(a) and including any New Branch Loans accepted by Purchaser pursuant to Section 3.04(b).

3.05 Review and Selection of Residential Loans. Purchaser and Seller agree that Purchaser shall be provided the opportunity to acquire, as Acquired Loans hereunder, up to \$13.5 million in aggregate principal amount of single family residential mortgage loans and home equity lines of credit (collectively, "Residential Loans") to be identified and selected in accordance with this Section 3.05. Commencing upon the execution of this Agreement and continuing until the Closing Date, Purchaser will be provided an opportunity to review and examine the Residential Loans then-outstanding (including any new Residential Loans made by Seller after the date hereof and prior to the Closing Date) in Seller's identified Northeast Ohio portfolio, including the Loan Files relating to such Residential Loans. Based on such review and examination, Purchaser shall be entitled to select up to \$13.5 million in aggregate principal amount of such Residential Loans to include in the Acquired Loans by providing written notice to Seller identifying the Residential Loans so selected. In the event that there are less than \$13.5 million in aggregate principal amount of Residential Loans in Seller's Northeast Ohio portfolio, after excluding Nonperforming Loans and any other loans that do not meet Purchaser's underwriting criteria, Seller will provide to Purchaser the opportunity to review and examine additional Residential Loans in Seller's portfolio (outside Northeast Ohio) and to select any such additional Residential Loans to be included in the Acquired Loans (by providing written notice to Seller identifying the additional Residential Loans so selected); provided, however, that the total of all Residential Loans selected by Purchaser to include in the Acquired Loans shall not exceed an aggregate principal amount of \$13.5 million as of the Closing Date unless otherwise consented to in writing by Seller. Purchaser and Seller agree that the Residential Loans included in the Acquired Loans in accordance with this Section 3.05 will have a weighted-average yield of not less than 3.58% and will not include any loan to a single customer in excess of \$1.0 million in the aggregate, and Seller shall provide to Purchaser the opportunity to select additional Residential Loans from Seller's portfolio (outside Northeast Ohio) to the extent necessary to enable Purchaser to select Residential Loans in an aggregate of not less than \$13.5 million under this Section 3.05 that satisfy such requirements in the aggregate. If the customer for any of the Residential Loans acquired by Purchaser under this Section 3.05 is in the Identified Market, Purchaser will have all rights to each such customer, including but not limited to the rights to Seller's relationship with each such customer; provided, however, that nothing contained herein shall prohibit or otherwise restrict Seller from continuing to maintain any deposit account that existed with Seller as of the Closing Date (other than a Deposit Account) with any such loan customer. If the customer for any of the Residential Loans acquired by Purchaser under this Section 3.05 is outside of the Identified Market, Seller will be entitled to maintain a relationship with the customer, except as related to the Residential Loan.

### 4. REPRESENTATIONS AND WARRANTIES.

- 4.01 <u>Representations and Warranties of Seller</u>. Except as otherwise disclosed in the Seller Disclosure Schedule, if any, delivered to Purchaser as of the date of this Agreement, Seller represents and warrants to Purchaser as follows:
- (a) Existence and Power of Seller. Seller is a national banking association duly organized, validly existing and in good standing under the laws of the United States of America with corporate power to own its properties and to carry on its business as presently conducted. Seller is an insured depository institution, as defined in the Federal Deposit Insurance Act and applicable regulations thereunder.
- (b) <u>Authorization of Agreement</u>. The execution and delivery of this Agreement, and the transactions contemplated hereby, have been duly authorized by all necessary corporate action on the part of Seller, and this Agreement is a valid and binding obligation of Seller, enforceable against Seller in accordance with its terms, except as enforcement may be limited by federal regulators of Seller or Purchaser or by bankruptcy, insolvency, reorganization, moratorium or other laws of general applicability relating to or affecting creditors' rights, or rules of law governing specific performance, equitable relief and other equitable remedies or the waiver of rights or remedies. Without limiting the foregoing, Seller has (or by Closing will have) obtained and procured all necessary corporate approvals and authorizations required on its part to enable it to fully perform all obligations imposed on it hereunder which must be performed by it at or prior to the Closing. To the Knowledge of Seller, there are no facts or circumstances regarding Seller that would be reasonably likely to result in any necessary regulatory approvals of the Acquisition to be denied, unreasonably delayed, or subject to a Burdensome Regulatory Condition.

- (c) <u>Effective Agreement</u>. Subject to the receipt of any and all necessary regulatory approvals, the execution, delivery, and performance of this Agreement by Seller and the consummation of the transactions contemplated hereby, will not conflict with, result in the breach of, constitute a violation or default, result in the acceleration of payment or other obligations, or create a Lien, under any of the provisions of the Articles of Association or Bylaws of Seller, under any judgment, decree or order, under any law, rule or regulation of any government or agency thereof, or under any material contract, material agreement or material instrument to which Seller is subject or by which Seller is otherwise bound, which such breach, conflict, violation, default, acceleration or Lien could reasonably be expected to have a material adverse effect on the Assets, the Acquisition or Seller's ability to perform its obligations hereunder.
- (d) <u>Title to Furniture, Fixtures and Equipment</u>. Seller is the sole owner of and has good and marketable title to the furniture, fixtures and equipment listed on <u>Exhibit 2.02(a)</u> attached hereto and indicated thereon as owned by Seller, free and clear of all Liens. To the Knowledge of Seller, each item of tangible personal property included in the Assets is in good operating condition, ordinary wear and tear excepted, is suitable for use in the ordinary course of the business and operations of the Branches as currently conducted, and, subject only to the foregoing representations, will be received by Purchaser in "AS IS, WHERE IS" condition, with no implied warranties or guarantees by Seller as to future performance, fitness for particular purpose, merchantability or otherwise.

#### (e) Real Estate.

- (i) Seller is the sole owner of a fee simple interest in the Real Estate. Except as may be disclosed in the Seller Disclosure Schedule, Seller has good, marketable and insurable title to the Real Estate, free and clear of any Liens, except for Permitted Encumbrances.
- (ii) To Seller's Knowledge, none of the Real Estate, as currently used, is in violation in any material respect of any applicable building, zoning, subdivision, Environmental Law or other governmental ordinance, resolution, statute, rule, order or regulation with respect to the ownership, operation, use, maintenance or condition thereof, or requiring any material repair or alteration other than those repairs or alterations that will have been made before the Closing Date;
- (iii) To Seller's Knowledge, there is no material structural defect in any improvement on or to any of the Real Estate;
- (iv) Seller has paid or will pay in full on or before the Closing Date all bills and invoices for labor and materials of any kind arising from the operation, repair, maintenance or leasing or any of the Real Estate by Seller that could result in a mechanics lien or a claim for a mechanics lien;
- (v) Seller has received no notice of any condemnation or eminent domain proceeding pending or threatened against any Real Estate, and has no Knowledge of any change or proposed change in the route, grade or width of, or otherwise affecting, any street or road adjacent to or serving any Real Estate;

- (vi) The water, sewer, electric and other utility services currently available to the Real Estate are adequate for the present use of the Real Estate by Seller in conducting the business of the Branches and are being supplied to Seller by utility companies or municipalities, and no condition presently exists that, to Seller's Knowledge, will result in the termination of the Real Estate's access to any such utility services;
- (vii) All authorizations and rights-of-way, including proof-of-dedication, that are necessary to ensure vehicular and pedestrian ingress and egress to and from the Real Estate have been obtained, and there are no restrictions on such access to the Real Estate from presently existing adjacent public roads, and no condition presently exists that, to Seller's Knowledge, will result in the termination of such access;
- (viii) All real estate taxes, if any, due and payable on or before the Closing Date with respect to any of the Real Estate will have been paid in full on or before the Closing Date;
- (ix) To Seller's Knowledge, no activity or condition exists at or upon the Real Estate that violates any Environmental Law, and no condition has existed or event has occurred with respect to the Real Estate that, with notice or the passage of time, or both, would constitute a violation of any Environmental Law or obligate (or potentially obligate) Seller to remedy, stabilize, neutralize or otherwise alter the environmental condition of the Real Estate where the cost of such actions would be material in the aggregate. Seller has not received any written notice from any person or entity, and to Seller's Knowledge no facts or circumstances reasonably exist, that the operation or condition of any of the Real Estate is or was in violation of any Environmental Law or that a Release of Hazardous Materials has occurred at, on or under the Real Estate or that Seller is responsible (or potentially responsible) for the cleanup or other remediation of any Hazardous Materials at, on or beneath any such property; and
- (x) Except as may be otherwise specifically set forth in this Agreement, the Real Property is sold to Purchaser "AS IS, WHERE IS," with no implied warranties or representations whatsoever.
- (f) No Broker. Except for the retention by Seller of Boenning & Scattergood, Inc., the fees for which will be paid by Seller, no broker or finder, or other party or agent performing similar functions, has been retained by Seller or is entitled to be paid based upon any agreements, arrangements or understandings made by Seller in connection with the transactions contemplated hereby, and no brokerage fee or other commission has been agreed to be paid by Seller on account of the transactions contemplated hereby.
- (g) <u>Deposit Liabilities</u>. The Deposit Liabilities are genuine obligations of Seller and have been originated or extended and administered in material compliance with the documents governing the relevant type of Deposit Liabilities and all applicable Legal Requirements. The Deposit Liabilities are insured by the Federal Deposit Insurance Corporation through the Deposit Insurance Fund to the fullest extent provided for by applicable Legal Requirements. All interest has been accrued on the Deposit Liabilities, and Seller's records accurately reflect in all material respects such accrual of interest in accordance with GAAP.
- (h) <u>Safe Deposit Boxes</u>. Seller is in compliance in all material respects with the terms and conditions of any and all agreements relating to safe deposit boxes at the Branches.

- (i) <u>Branch Business</u>. There are no Legal Proceedings or any order, injunction, judgment, doctrine, decree, ruling, writ, assessment or arbitration award of a Governmental Authority entered or pending or, to the Knowledge of Seller, threatened against or affecting the Assets, the Assumed Liabilities, or the business of Seller as conducted at the Branches. To the Knowledge of Seller, there are no investigations or inquiries by or before a Governmental Authority pending or threatened against or affecting the Assets, the Assumed Liabilities, or the business of Seller as conducted at the Branches.
- (j) <u>Employee Matters</u>. Seller is not a party to or otherwise bound by any collective bargaining agreement, and there are no labor unions representing any employees at the Branches. To Seller's Knowledge, there are no activities by labor unions to organize any employees at the Branches.
- (k) <u>Tax Matters</u>. Seller has withheld and paid all material Taxes required to have been withheld and paid in connection with amounts paid or owing to any employee employed at the Branches, and all Forms W-2 and 1099 required with respect thereto have been properly completed in all material respects and timely filed.

#### (l) ERISA.

- (i) Neither Seller nor any ERISA Affiliate has any liability or liabilities under any Employee Benefit Plan that will become a liability or liabilities of Purchaser or any Affiliate of Purchaser or result in any Lien on the Assets or on any other assets of Purchaser or Purchaser's Affiliates.
- (ii) None of the Assets (A) are "plan assets" of any Employee Benefit Plan, (B) are subject to any Lien relating to any Employee Benefit Plan under ERISA, the Code or otherwise, or (C) otherwise have been identified or earmarked as available for or relating to benefits under any Employee Benefit Plan.
- (m) <u>Insurance</u>. The Assets described at Section 2.02(a) of this Agreement are adequately insured by financially sound and reputable insurers in such amounts and against fire and other risks as is customary with banks of similar size. The Seller Disclosure Schedule sets forth, for each material policy of insurance maintained by Seller with respect to either of the Branches, the amount and type of insurance and the name of the insurer. All amounts due and payable under such insurance policies are fully paid, and all such insurance policies are in full force and effect.
  - (n) Acquired Loans. With respect to the Acquired Loans to be sold by Seller to Purchaser hereunder:
  - (i) The loan files for each Acquired Loan (the "Loan Files") contain originally executed notes, leases and/or other evidences of any indebtedness, including without limitation all originally executed loan agreements, loan participation agreements and certificates, control agreements, security agreements, mortgages, guarantees, UCC financing statements and similar documents evidencing collateral or other financial accommodations relating to the Loans and any document modifying any of the foregoing (the "Loan Documents"). The Loan Files include, without limitation, all loan applications, collectors notes, appraisals, credit reports, disclosures, titles to collateral (titles to cars, boats, etc.), all verifications (including employment verification, deposit verification, etc.), financial statements of borrowers and guarantors, independently prepared financial statements, internally prepared financial statements, commitment letters, guarantees, pledge agreements, intercreditor agreements, participation agreements, sureties and insurance policies (including title insurance policies) and all written modifications, waivers and consents relating to any of the foregoing.

- (ii) Except for any Nonperforming Loans to which Purchaser consents in writing to acquire pursuant to Section 2.02(d) hereof, no Acquired Loan is a Nonperforming Loan. To the Knowledge of Seller, no Acquired Loan is subject to any material default, breach or violation by any obligor under the Acquired Loan, no maker, signatory or guarantor on any Acquired Loan is in bankruptcy, receivership, or any other form of insolvency proceeding, and no Acquired Loan is subject to any offset or claim of offset, or claim of other liability on the part of Seller.
- (iii) Seller is the sole servicer for each of the Acquired Loans. The servicing practices of Seller used with respect to the Acquired Loans have been in compliance in all material respects with all Legal Requirements.
- (iv) No borrower, customer or other party in connection with the Acquired Loans has notified Seller, or has asserted against Seller, in each case in writing, any "lender liability" or similar claim or any defense to the payment in full of any of the Acquired Loans.
- (v) Seller has made no commitment to make or modify the terms or conditions of any Acquired Loan other than as set forth in the Loan Files. No borrower has been released, in whole or in part, from liability or obligations in respect of any Acquired Loan and no property or other collateral has been released from any applicable Lien.
- (vi) Seller is the sole owner and holder of the Acquired Loans, has all power and authority to hold the Acquired Loans and has good and marketable title to the Acquired Loans free and clear of any and all Liens. There has been no previous sale, assignment, pledge or hypothecation of any Acquired Loan or related servicing rights by Seller.
- (vii) With respect to each Acquired Loan, all costs, fees and expenses incurred in making, closing and recording any applicable mortgage have been paid and the full principal amount of the note has been disbursed or advanced to the borrower or disbursed or advanced according to the written direction of the borrower.
- (viii) Each Acquired Loan complies in all material respects with all Applicable Laws, including, but not limited to, the Truth-In-Lending Act, Real Estate Settlement and Procedures Act, Home Ownership and Equity Protection Act of 1994, Consumer Credit Protection Act, Equal Credit Opportunity Act, Fair Housing and Disclosure Act, Regulation Z, and the Home Mortgage Disclosure Act.
- (ix) Each Acquired Loan is secured by lien of first priority (other than home equity lines of credit) in and on the relevant collateral, if any, and, if applicable, any mortgage has been duly executed by the borrower and properly acknowledged and filed or recorded in the appropriate office for public recordation, and all security interests have been perfected in accordance with Applicable Law and all applicable fees relating thereto have been paid. Seller has insurance certificates evidencing that any pledged or mortgaged property is insured by an insurer licensed to do business in the state where such pledged or mortgaged property is located and generally acceptable in the industry against loss by fire, theft, vandalism and hazards as are customarily insured against in the area in which the pledged mortgaged property is located, in an amount which is at least equal to the outstanding principal balance of the related Acquired Loan, and Seller has received no cancellation notices with respect to such insurance coverage. If any mortgaged property is in an area as having special flood hazards and, as of the date of origination or acquisition of the Acquired Loans, Seller was required to maintain flood insurance on the Acquired Loans, a flood insurance policy meeting applicable requirements is in full force and effect with a generally acceptable insurance carrier, in an amount not less than the outstanding principal balance of the Loan. All insurance policies contain a standard mortgagee clause naming Seller, its successors and assigns as an additional insured, and a provision for notice to mortgagee in the event of cancellation of the policy.

- (x) Seller has received no notice of, and otherwise has no knowledge of, any pending or threatened proceeding for the total or partial condemnation of any mortgaged or pledged property.
- (xi) Each Acquired Loan that is a mortgage loan is insured by an ALTA Mortgage Title Insurance Policy, valid and binding in the jurisdiction where the related mortgaged property is located, insuring Seller, its successors and assigns, as to the first priority lien of the mortgage in the original principal amount of the Loan, as increased by deferred and capitalized interest, if applicable, and subject only to permitted exceptions. Notwithstanding anything to the contrary herein, Purchaser shall not be obligated to purchase any Loan if, in its reasonable judgment, it is unsatisfied with any exceptions to title.

Except as otherwise expressly provided herein, the transfer of the Acquired Loans will be made without recourse, without any representation, warranty, or guarantee of any kind, express or implied, and without any reserve for loan losses.

- (o) <u>Subordinated Debt Securities</u>. Seller is the sole owner of and has good and marketable title to each of the Subordinated Debt Securities, free and clear of all Liens. Except as set forth in the Seller Disclosure Schedule, each Subordinated Debt Security is a negotiable instrument, a certificated security, or an uncertificated security as those terms are defined in the Uniform Commercial Code as in effect in the State of Ohio.
- (p) <u>Pre-Closing Statement</u>. The Pre-Closing Statement provided in accordance with this Agreement will accurately reflect in all material respects the Net Book Value of the Assets referred to therein as of their respective dates, recorded at their historical cost and depreciated or otherwise adjusted in accordance with Seller's historical accounting policies, all in accordance with GAAP consistently applied.
- 4.02 <u>Representations and Warranties of Purchaser</u>. Except as otherwise disclosed in Purchaser's Disclosure Schedule, if any, delivered to Seller as of the date of this Agreement, Purchaser represents and warrants to Seller as follows:
- (a) Good Standing and Power of Purchaser. Purchaser is a national banking association duly organized, validly existing, and in good standing under the laws of the United States of America with corporate power to own its properties and to carry on its business as presently conducted. Purchaser is an insured depository institution, as defined in the Federal Deposit Insurance Act and applicable regulations thereunder.
- (b) <u>Authorization of Agreement</u>. The execution and delivery of this Agreement, and the transactions contemplated hereby, have been duly authorized by all necessary corporate action on the part of Purchaser, and this Agreement is a valid and binding obligation of Purchaser, enforceable against Purchaser in accordance with its terms, except as enforcement may be limited by federal regulators of Seller or Purchaser or by bankruptcy, insolvency, reorganization, moratorium or other laws of general applicability relating to or affecting creditors' rights, or rules of law governing specific performance, equitable relief and other equitable remedies or the waiver of rights or remedies. Without limiting the foregoing, Purchaser has obtained and procured all necessary corporate approvals and authorizations required on its part to enable it to fully perform all obligations imposed on it hereunder which must be performed by it at or prior to the Closing.

- (c) Effective Agreement. Subject to the receipt of any and all necessary regulatory approvals, the execution, delivery and performance of this Agreement by Purchaser, and the consummation of the transactions contemplated hereby, will not conflict with, result in the breach of, constitute a violation or default, result in the acceleration of payment or other obligations, or create a Lien, under any of the provisions of the Articles of Association and the Bylaws of Purchaser, under any judgment, decree or order, under any law, rule or regulation of any government or agency thereof, or under any material agreement, material contract or material instrument to which Purchaser is subject or by which Purchaser is otherwise bound, which such breach, conflict, violation, default, acceleration, or Lien could reasonably be expected to have a material adverse effect on the Acquisition or Purchaser's ability to perform its obligations hereunder.
- (d) <u>Broker/Finder</u>. Except for the retention by Purchaser of Piper Sandler & Co., the fees for which will be paid by Purchaser, no broker or finder, or other party or agent performing similar functions, has been retained by Purchaser or is entitled to be paid based upon any agreements, arrangements or understandings made by Purchaser in connection with the transactions contemplated hereby, and no brokerage fee or other commission has been agreed to be paid by Purchaser on account of the transactions contemplated hereby.
- (e) Acquisition Consideration. Purchaser has sufficient funds to enable it to pay the Acquisition Consideration and any amounts required to be paid by Purchaser at the Closing pursuant to the terms hereof, and to consummate the Acquisition in accordance with the terms and conditions of this Agreement, provided that Seller's payment obligations are netted against Purchaser's payment obligations as provided in Section 2.04(e). Purchaser's ability to consummate the transactions contemplated hereby is not contingent on raising any equity capital, obtaining specific financing therefor, consent of any lender or any other matter, other than the receipt of any required regulatory approval(s).
- (f) Regulatory Approvals. No consents or approvals of, or filings or registration with, any Governmental Authority or any third party are required to be made or obtained by Purchaser or any of its subsidiaries in connection with the execution, delivery or performance by Purchaser of this Agreement or to consummate the transactions contemplated hereby, except for the filing of applications and notices, as applicable, with the applicable regulatory authorities and receipt of the approvals set forth in Section 3.02(a). As of the date hereof, Purchaser is not aware of any reason why the requisite regulatory authority approvals will not be received without the imposition of a condition, restriction or requirement of the type that is not customary and usual for approvals of this kind. On the date hereof, Purchaser knows of no reason why any required regulatory application should not receive approval.

#### 5. ACTIONS RESPECTING EMPLOYEES AND EMPLOYEE BENEFIT PLANS.

# 5.01 General

(a) Purchaser may, but is under no obligation to, offer to hire, as of the close of business on the Closing Date, the Branch Employees. Purchaser will make a written offer of employment to each of the Branch Employees selected by Purchaser to be an employee of Purchaser following the Closing Date. Each offer of employment will be effective as of the close of business on the Closing Date. No later than fourteen (14) days prior to the Closing, (i) Purchaser will communicate the offers of employment consistent with the terms of this Section 5.01 to those Branch Employees to whom it determines to extend an offer and (ii) Purchaser will provide Seller with a written list of such Branch Employees, and Seller will take such action as is necessary to terminate the employment of any Branch Employees not included on such list or transfer their employment within Seller, in both cases effective as of the close of business on the Closing Date. Each employee who receives and accepts Purchaser's offer of employment will be a "Hired Employee" for purposes of this Agreement, effective as of the close of business on the Closing Date and this date will be referred to as the Hired Employees's "Transfer Date." Subject to Applicable Law, effective as of the close of business on the Closing Date, the Hired Employees will become employees of Purchaser, and Purchaser will thereafter have the right to dismiss any or all Hired Employees at any time, with or without cause, and to change the terms and conditions of their employment (including compensation and employee benefits provided to them). Each employee who is not offered employment by Purchaser, or who fails to accept Purchaser's offer of employment, will be an "Excluded Employee" for purposes of this Agreement. Nothing in this Agreement will give any Branch Employee any rights to claim status as a third-party beneficiary of this Agreement.

- (b) Seller will pay, discharge, and be responsible for, and will indemnify Purchaser and its Affiliates for, (i) all salary, wages (including, without limitation, payment for any and all paid time off, vacation, sick time or personal days accrued by the Hired Employees as of the Transfer Date), bonuses, commissions and any other form of compensation (including, without limitation, any deferred compensation) arising out of the employment of the Hired Employees prior to or on the Transfer Date, and (ii) any employee benefits under the Seller's Employee Benefit Plans arising out of Seller's employment of the Hired Employees, prior to or on the Transfer Date, including, without limitation, welfare benefits with respect to claims incurred prior to or on the Transfer Date but reported after the Transfer Date.
- (c) Seller and the Seller's Employee Benefit Plans will retain responsibility for all claims incurred by Branch Employees of Seller prior to or on the Transfer Date (in accordance with the terms of such plans with respect to claims incurred), including, without limitation, those made by Hired Employees.
- (d) Seller will be solely and fully responsible for the Excluded Employees in the transfer of Excluded Employees to other positions with Seller or in the termination of their employment with Seller. To the extent Seller is terminating any Excluded Employees, Seller will be fully and completely responsible for said termination(s), paying those Excluded Employees all accrued wages, commissions, vacation pay and any other compensation due and owing, and providing notice of such Excluded Employees rights and obligations pursuant to COBRA and administering COBRA benefits, as applicable. Seller will be fully liable for any claims, demands, damages, orders, awards and/or judgments arising out of or relating in any way to Seller's treatment of and/or termination of Excluded Employees.
- (e) It is understood and agreed that (i) Purchaser's employment of any Hired Employee as set forth in this Section 5.01 will not constitute an employee benefit plan or a commitment, contract or understanding (express or implied) of an obligation on the part of Purchaser to a post-Closing employment relationship of any fixed term or duration or upon any terms or conditions other than those that Purchaser may establish pursuant to individual offers of employment, and (ii) employment offered by Purchaser to any Hired Employee is "at will" and may be terminated by Purchaser or by a Hired Employee at any time for any reason (subject to any written commitments to the contrary made by Purchaser to a Hired Employee and subject to any Applicable Law). Nothing in this Agreement will be deemed to prevent or restrict in any way the right of Purchaser to terminate, reassign, promote or demote any of the Hired Employees after the Closing Date or to change adversely or favorably the title, powers, duties, responsibilities, functions, locations, salaries, other compensation, or other terms or conditions of employment of such Hired Employees. Nothing in this Agreement will create any employee benefit plan or be construed as requiring any compensation or employee benefit plan, program or arrangement to be maintained by Purchaser for any Hired Employee at, or for any specified period after, the Closing Date. Notwithstanding anything else in this Section 5.01, Purchaser will use its best efforts to give Hired Employees credit for service with Seller toward eligibility for participation in and vesting under Employee Benefit Plans of Purchaser.

- (f) For the avoidance of doubt, Purchaser will have no responsibility to pay any severance or to fund benefits under any of the employee benefit programs maintained by Seller prior to the close of business on the Closing Date.
- 5.02 <u>Action to be Taken by Seller</u>. Seller covenants to Purchaser that, for twelve (12) months following the Closing Date, Seller will not compete directly for or solicit as officers or employees of Seller any Branch Employees; <u>provided</u>, <u>however</u>, that this provision will not preclude Seller from (i) making any general advertisement or solicitation for employees not targeting former Branch Employees or (ii) hiring any such Branch Employees who elect not to become or remain employed by Purchaser, who are not hired by Purchaser or whose employment has been terminated by Purchaser.
- 5.03 Actions to be Taken by Seller and Purchaser. After the Closing, Seller and Purchaser each will cooperate with the other in providing reasonable access to all information required for the operation of, or the preparation and submission of reports or notices required in connection with the operation of Seller's tax-qualified retirement plans, the Purchaser's retirement plan(s), or any other employee benefit program maintained by Seller or Purchaser which covers any of the Branch Employees, including, without limitation, the preparation and submission of reports or notices to the Pension Benefit Guaranty Corporation, the Department of Labor, the Internal Revenue Service, or any other agency of the U.S. Government, or the State of Ohio.
- 5.04 <u>Continuing Rights to Amend or Terminate</u>. The provisions of Purchaser's retirement plan(s) and any of Seller's retirement plans and the provisions of any other employee benefit program of Purchaser or Seller relating to amendment or termination by any employer sponsor or other party to such plan or program will in no event be abridged hereby.

#### 6. CONDITIONS PRECEDENT TO CLOSING.

- 6.01 <u>Conditions to Seller's Obligations</u>. The obligations of Seller to consummate the Acquisition are subject to the satisfaction, or the waiver by Seller to the extent permitted by Applicable Law, of the following conditions at or prior to the Closing:
- (a) <u>Prior Regulatory Approval</u>. All filings and registrations with, and notifications to, all federal and state authorities required for consummation of the Acquisition have been made; all approvals and authorizations of all federal and state authorities required for consummation of the Acquisition have been received and are in full force and effect and evidence of same will have been provided to Seller; and all applicable waiting periods have expired.
- (b) <u>Corporate Action</u>. Purchaser will have taken all corporate action necessary by it to effectuate this Agreement and the Acquisition; and Purchaser will have furnished Seller with a certified copy of each such resolution adopted by the Board of Directors (and, if applicable, the shareholders) of Purchaser evidencing the same.
- (c) <u>Representations and Warranties</u>. The representations and warranties of Purchaser set forth in this Agreement will be true and correct in all material respects on the Closing Date with the same effect as though all such representations and warranties had been made on and as of such date (except for such representations and warranties that speak as of a specified time or period, which will be true and correct in all material respects as of the time or period specified), and Purchaser has delivered to Seller a certificate (executed by its chief executive officer) to that effect, dated as of the Closing Date.

- (d) <u>Covenants</u>. Each and all of the covenants and agreements of Purchaser to be performed or complied with prior to Closing pursuant to this Agreement have been duly performed or complied with in all material respects by Purchaser, and Purchaser has delivered to Seller a certificate (executed by its chief executive officer) to that effect, dated as of the Closing Date.
- (e) No Proceeding or Prohibition. At the time of the Closing, there is no litigation, investigation, inquiry or proceeding pending or threatened in or by any court or agency of any government or by any third party which, in the good faith judgment of Seller, presents a bona fide claim to restrain, enjoin or prohibit consummation of the transactions contemplated by this Agreement or which might result in divestiture or rescission in connection with such transactions; and Seller has been furnished with a certificate in the form attached as Exhibit 6.01(e) hereto, dated as of the Closing Date and signed by the chief executive officer of Purchaser, to the effect that no such litigation, investigation, inquiry or proceeding is pending, or, to the best of his knowledge, threatened.
- 6.02 <u>Conditions to Purchaser's Obligations</u>. The obligations of Purchaser to consummate the Acquisition in whole or in part are subject to the satisfaction, or the waiver by Purchaser to the extent permitted by Applicable Law, of the following conditions at or prior to the Closing:
- (a) <u>Prior Regulatory Approval</u>. All filings and registrations with, and notifications to, all federal and state authorities required for consummation of the Acquisition have been made; all approvals and authorizations of all federal and state authorities required for consummation of the Acquisition will have been received, have been received and are in full force and effect, and do not contain any Burdensome Regulatory Conditions; and all applicable waiting periods have expired.
- (b) <u>Corporate Action</u>. Seller has taken all corporate action necessary by it to effectuate this Agreement and the Acquisition; and Seller has furnished Purchaser with a certified copy of each such resolution adopted by the Board of Directors (and, if applicable, the shareholders) of Seller evidencing the same.
- (c) <u>Representations and Warranties</u>. The representations and warranties of Seller set forth in this Agreement are true and correct in all material respects on the Closing Date with the same effect as though all such representations and warranties had been made on and as of such date (except for such representations and warranties that speak as of a specified time or period, which will be true and correct in all material respects as of the time or period specified), and Seller has delivered to Purchaser a certificate (executed by its chief executive officer) to that effect, dated as of the Closing Date.
- (d) <u>Covenants</u>. Each and all of the covenants and agreements of Seller to be performed or complied with prior to Closing pursuant to this Agreement have been duly performed or complied with in all material respects by Seller, and Seller has delivered to Purchaser a certificate (executed by its chief executive officer) to that effect, dated as of the Closing Date.
- (e) No Proceedings or Prohibition. At the time of the Closing, there is no litigation, investigation, inquiry or proceeding pending or threatened in or by any court or agency of any government or by any third party which, in the good faith judgment of Purchaser, presents a bona fide claim to restrain, enjoin or prohibit consummation of the transactions contemplated by this Agreement or which might result in divestiture or rescission in connection with such transactions; and Purchaser has been furnished with a certificate in the form attached as Exhibit 6.02(e) hereto, dated as of the Closing Date and signed by the chief executive officer of Seller, to the effect that no such litigation, investigation, inquiry or proceeding is pending, or, to the best of Seller's Knowledge, threatened.

- (f) <u>Participations</u>. Seller shall have performed its obligations under Section 2.05 to the extent such obligations are to be performed before the Closing.
- 6.03 <u>Non-Satisfaction of Conditions Precedent</u>. The non-occurrence or delay of the Closing of the Acquisition by reason of the failure of timely satisfaction of all conditions precedent to the obligations of any party hereto to consummate the Acquisition will in no way relieve such party of any liability to the other party hereto, nor be deemed a release or waiver of any claims the other party hereto may have against such party, if and to the extent the failure of timely satisfaction of such conditions precedent is attributable to the actions or inaction of such party.

#### 7. CLOSING.

7.01 Closing and Closing Date. The Acquisition will be consummated and closed (the "Closing") at such location as may be mutually agreed upon by Seller and Purchaser, at a date and time to be mutually established by Seller and Purchaser and confirmed by written notice of Seller to Purchaser (the "Closing Date"). Unless otherwise agreed by the parties, the Closing Date will take place as soon as practicable after the receipt of all necessary regulatory approvals to consummate the Acquisition and the expiration of all applicable waiting periods, and the satisfaction or waiver of all other conditions set forth in Section 6 of this Agreement; provided, however, that (i) the Closing Date will be established for a Friday (unless a legal holiday and in that event the prior Business Day), and (ii) the Closing will commence at a time sufficient to conclude the Closing prior to the time established for the normal close of business of Seller at the Branches and to result in actual receipt on the Closing Date by payees of all funds to be delivered on the Closing Date.

#### 7.02 Seller's Actions at Closing. At the Closing, Seller will:

- (a) deliver to Purchaser such of the Assets purchased hereunder as are capable of physical delivery (including delivery by their presence at the Real Estate), including a complete set of Seller's keys to the safe deposit boxes and to the Branches, including without limitation, keys to any night deposit boxes and the like; and
- (b) execute and deliver a bill of sale in the form attached as <u>Exhibit 7.02(b)</u> hereto relating to furniture, fixtures and equipment and other instruments of conveyance, assignment and transfer as are reasonably necessary or advisable to consummate the sale, assignment and transfer of the Assets, other than the Real Estate, sold or assigned to Purchaser hereunder;
- (c) assign, transfer and deliver to Purchaser such of the following records (in whatever form or medium then maintained by Seller) pertaining to the Deposit Liabilities and accrued interest thereon of the Branches assumed by Purchaser hereunder as exist and are available:
  - (i) signature cards, orders and contracts between Seller and Branch depositors and safe deposit box customers, and records of similar character; and
  - (ii) cancelled checks or negotiable orders of withdrawal representing charges to depositors (except to the extent the same have been returned to such depositors); and
    - (iii) a trial balance listing of records of account;

- (d) assign, transfer and deliver to Purchaser such safekeeping files and records (in whatever form or medium then maintained by Seller) and all other property and other records, if any, held by the Branches for their customers as of the Closing Date (subject to the terms and agreements or receipts relating to the same);
  - (e) execute and deliver an assignment of IRAs in the form attached as Exhibit 7.02(e) hereto;
- (f) execute, acknowledge and deliver to Purchaser a blanket assignment of mortgages in the form attached as Exhibit 7.02(f) hereto, for mortgage loans transferred to Purchaser pursuant to this Agreement;
- (g) execute and deliver to Purchaser an assignment of right, title and interest in the form attached as Exhibit 7.02(g) hereto and transfer and deliver all Loan Files and records as exist and are available (in whatever form or medium then maintained by Seller), including all original copies as exist and are available, pertaining to the Acquired Loans transferred to Purchaser hereunder;
- (h) make, if applicable, any payment shown as owing to Purchaser by Seller on the Pre-Closing Statement, as required by Section 2.04 of this Agreement;
- (i) execute, acknowledge and deliver to Purchaser all certificates and other documents required to be delivered to Purchaser by Seller at or as a condition to Closing pursuant to the terms hereof or as Purchaser may reasonably require to effect the transactions provided for in this Agreement;
- (j) execute and deliver to Purchaser a certificate, dated as of the Closing Date, that meets the requirements of Treasury Regulation Section 1.1445-2(b) and is signed by an authorized officer of Seller to the effect that Seller is not a "foreign person" as defined in Section 1445 of the Code;
- (k) execute, acknowledge and deliver to Purchaser a limited warranty deed for each piece of Real Estate in the form attached as Exhibit 7.02(k) hereto; and
- (l) execute and deliver to Purchaser an assignment, in such form required by the applicable issuer, and any other documents, and perform any actions, required by the applicable issuer or book-entry system to assign and transfer to Purchaser all of Seller's right, title and interest in each of the Subordinated Debt Securities.

#### 7.03 <u>Purchaser's Actions at Closing.</u> At the Closing, Purchaser will:

- (a) execute, acknowledge, and deliver to Seller, to evidence the assumption of the Deposit Liabilities and certain other obligations of Seller by Purchaser hereunder, an instrument of assumption in the form attached as Exhibit 7.03(a) to this Agreement and an assignment of IRAs in the form attached as Exhibit 7.02(e) to this Agreement. Copies of such instruments may be recorded in the public records at the option of either party hereto. The execution and acknowledgement of such instruments will not be deemed to be a waiver of any rights or obligations of any party to this Agreement;
- (b) receive, accept and acknowledge delivery (including delivery by their presence at the Real Estate) of all Assets, and all records and documentation relating thereto, sold, assigned, transferred, conveyed or delivered to Purchaser by Seller hereunder;

- (c) execute and deliver to Seller such written receipts for the Assets, properties, records and other materials assigned, transferred, conveyed or delivered to Purchaser hereunder as Seller may reasonably request at or before the Closing;
- (d) make, as applicable, any payment shown as owing to Seller by Purchaser on the Pre-Closing Statement, as required by Section 2.04 of this Agreement; and
- (e) execute, acknowledge and deliver to Seller all certificates and other documents required to be delivered to Seller by Purchaser at or as a condition to Closing pursuant to the terms hereof or as Seller may reasonably require to effect the transactions provided for in this Agreement.
- 7.04 Methods of Payment. Any payment due to Purchaser or to Seller, as the case may be, pursuant to the terms of Section 2.04 hereof, will be calculated and adjusted as provided in this Section 7.04 and such payment will be made in immediately available United States dollars to the account of the respective party at a bank designated by such party. At least two (2) Business Days prior to the Closing, each party hereto will provide written notice to the other party hereto indicating the account and bank to which such funds will be wire transferred. Prorations and expenses that cannot be reasonably calculated at the Closing will be settled and paid in the same manner as set forth in Section 8.03. In order to facilitate the Closing, the parties agree that, within thirty (30) Business Days after the Closing, the parties will make appropriate post-Closing adjustments through the close of business on the Closing Date, consistent with the provisions of Section 2.04 hereof, based upon transactions which took place prior to the close of business on the Closing Date but which were not reflected on the preliminary settlement statement. In addition, prorations of prepaid and deferred expenses, if any, that cannot be reasonably calculated at the Closing will be settled and paid based on actual figures as soon as possible after the Closing.
- 7.05 <u>Effectiveness of Closing</u>. All actions taken and documents to be delivered at the Closing will be deemed to have been taken and delivered simultaneously, and no action will be deemed taken nor any document delivered until all have been taken and delivered. Upon the satisfactory completion of such Closing, the Acquisition will be deemed to be effective, and the Closing will be deemed to have occurred, as of the close of business on the Closing Date.

#### 8. <u>CERTAIN TRANSITIONAL MATTERS</u>.

## 8.01 <u>Customer Communications</u>.

- (a) Purchaser and Seller will, as soon as practicable after the date of this Agreement and, in any event, not more than thirty (30) days after the date of this Agreement, prepare and mail to each of the borrowers under the Acquired Loans and each of the depositors and other customers of the Branches a joint letter, in form and substance reasonably satisfactory to each of the parties, informing such customers of the nature of the transactions contemplated by this Agreement and the continuing availability of services to be provided by Purchaser from and after the close of business on the Closing Date. All costs and expenses of any such notice or communication will be borne by Purchaser.
- (b) Purchaser will separately furnish appropriate customers by mail with brochures, forms and other written materials related or necessary to the acquisition of the Acquired Loans and the assumption of the Deposit Liabilities by Purchaser, and the conversion of said accounts to Purchaser accounts, including (i) provisions for checks to appropriate depositors and line of credit borrowers using the forms of Purchaser with instructions to such customers to utilize such Purchaser checks after the Closing Date and thereafter to destroy any unused checks on Seller's forms, and (ii) the reissuance of ATM and debit cards associated with applicable Deposit Liabilities. The mailing containing such brochures, forms or other written materials of Purchaser will be sent no later than the earlier of ten (10) days or such other period as may otherwise be required by Applicable Law prior to the Closing Date and no later than the time prior to the Closing Date reasonably deemed necessary by Purchaser to provide depositors with all necessary forms to accomplish the assumption of the Deposit Liabilities, provided that check forms of Purchaser will not be mailed to depositors any earlier than a date reasonably believed to result in receipt by depositors no earlier than the Closing Date. The expenses of the printing, processing and mailing of such information of Purchaser will be borne by Purchaser. The expenses of providing new Purchaser checks and other forms and written materials to appropriate customers will be that of Purchaser. Before Closing, except as required or otherwise expressly permitted by this Agreement, Purchaser will not contact Seller's customers except as may occur in connection with advertising or solicitations directed to the public generally or except as necessary to accomplish an orderly transition of the Deposit Accounts to Purchaser, provided that such contacts do not occur until after the mailing of the materials described in Section 8.01(a) hereof.

- (c) Purchaser shall, as soon as practicable after the Closing Date, prepare and transmit at Purchaser's expense to the obligors on all Acquired Loans transferred to Purchaser pursuant to this Agreement a notice to the effect that the loan has been transferred and directing that payment be made to Purchaser at any address of Purchaser specified by Purchaser, with Purchaser's name as payee on any checks or other instruments used to make payments, and, with respect to all Acquired Loans on which payment notices or coupon books have been issued, to issue new notices or coupon books reflecting the name and an address of Purchaser as the person to whom and the place at which payments are to be made.
- (d) For each Deposit Account that would normally have a periodic statement, Seller will generate and mail, as soon as practicable after the Closing Date, a customer statement prepared as of the Closing, which will also confirm that the Acquisition has closed.
- (e) In addition to the foregoing, each of Seller and Purchaser will provide, or join in providing where appropriate, all notices to customers of the Branches and other Persons that Seller or Purchaser is required to give under Applicable Law or the terms of any agreement between Seller and any such customer in connection with the transactions contemplated hereby. A party required to send or publish any notice or communication pursuant to this Section 8.01(e) will furnish to the other party a copy of the proposed form of such notice or communication at least five (5) Business Days in advance of the date of the first mailing, posting, or other dissemination thereof to customers, and will not unreasonably refuse to incorporate any changes that the other such party reasonably proposes. All costs and expenses of any such notice or communication sent or published by Purchaser or Seller will be the responsibility of the party sending such notice or communication and all costs and expenses of any such joint notice or communication will be borne by Purchaser.
  - 8.02 General. Prior to and after the Closing, unless otherwise indicated:
    - (a) After the Closing, Purchaser will:
  - (i) pay in accordance with Applicable Law and its customary banking practices all properly drawn and presented checks, drafts, debits and withdrawal orders presented to Purchaser by mail, over the counter, through electronic media, or through the check clearing procedures of the banking industry, by depositors of the Deposit Liabilities assumed by Purchaser hereunder, whether drawn on checks, drafts or withdrawal order forms provided by Seller or by Purchaser; and

(ii) in all other respects discharge, in the usual course of its banking business, the duties and obligations of Seller with respect to the balances due and owing to the depositors whose Deposit Accounts are assumed by Purchaser hereunder;

<u>provided</u>, <u>however</u>, that Purchaser's obligations pursuant to this Section 8.02(a) to honor checks, drafts and withdrawal orders on forms provided by Seller and carrying its imprint (including its name and transit routing number) will not apply to any such check, draft or withdrawal order presented to Purchaser more than ninety (90) days following the Closing Date.

- (b) After the Closing, if any depositor of a Deposit Liability assumed by Purchaser hereunder, instead of accepting the obligation of Purchaser to pay the Deposit Liabilities (including accrued interest thereon) assumed hereunder, demands payment from Seller for all or any part of any such assumed Deposit Liabilities (including accrued interest thereon), Purchaser acknowledges that Seller will not be liable or responsible for making such payment. Purchaser and Seller will make appropriate arrangements with each other at or prior to the Closing to provide for settlement by Purchaser of checks, returns and other items which are presented to Seller after the Closing Date and which are drawn on or chargeable to Deposit Accounts associated with Deposit Liabilities assumed by Purchaser hereunder.
- (c) Seller will (i) resign as of the close of business on the Closing Date as the custodian of each IRA Deposit Liability of which it is the custodian, (ii) to the extent permitted by the documentation governing each such IRA and Applicable Law, appoint Purchaser as successor custodian as of the close of business on the Closing Date of each such IRA, and Purchaser agrees to accept each such custodianship and, to the extent set forth in Section 2.03 hereof, assume all fiduciary obligations with respect thereto as of the close of business on the Closing Date, and (iii) deliver to the IRA grantor of each such IRA notice of the foregoing as is required by the documentation governing any such IRA or Applicable Law. If, pursuant to the terms or documentation governing any such IRA or Applicable Law, (y) Seller is not permitted to name Purchaser as successor custodian or the IRA grantor objects in writing to such assignment, or is entitled to, and does, in fact, name a successor custodian other than Purchaser, or (z) such IRA includes assets that are not Deposit Liabilities of Seller and are not being transferred to Purchaser, and the assumption of the Deposit Liabilities of Seller included in such IRA would result in a loss of qualification of such IRA under the Code or applicable regulations, all Deposit Liabilities of Seller held under such IRA will be excluded from the Deposit Liabilities and either retained by Seller or transferred to a different successor custodian in accordance with any directives properly provided by the IRA grantor.
- (d) Purchaser will promptly pay to Seller an amount equivalent to the amount of any checks, negotiable orders of withdrawal, drafts or withdrawal orders credited as of the close of business on the Closing Date to a Deposit Account associated with a Deposity Liability assumed by Purchaser hereunder which are returned uncollected to Seller on or after the Closing Date (including an amount equivalent to holds placed upon such Deposit Account for items cashed by Seller as of the close of business on the Closing Date which items are subsequently dishonored).
- (e) Subject to the obligations of Seller in Section 3.01(d), all tasks and obligations concerning the provision of data processing services to or for the Assets and Deposit Liabilities after the Closing are the sole and exclusive responsibility of, and will be performed solely and exclusively by, Purchaser.
- (f) If the balance due on any Acquired Loan transferred to Purchaser pursuant to this Agreement has been reduced by Seller as a result of a payment by check or draft received prior to the close of business on the Closing Date, which item is returned unpaid to Seller on or after the Closing Date, the asset value represented by the loan transferred shall be correspondingly increased and an amount in cash equal to such increase shall be promptly paid by Purchaser to Seller, and if any payment is received by Seller on or after the Closing Date on an Acquired Loan which has been transferred to Purchaser, Seller shall promptly remit any such payments to Purchaser.

#### (g) <u>Interest Reporting and Withholding.</u>

- (i) Unless otherwise agreed to by the parties, Seller will report to applicable taxing authorities and holders of Deposit Liabilities, with respect to the period from January 1 of the year in which the Closing occurs through the Closing Date, all interest (including dividends and other distributions with respect to money market accounts) credited to, withheld from and any early withdrawal penalties imposed upon the Deposit Liabilities. Purchaser will report to applicable taxing authorities and holders of Deposit Liabilities, with respect to all periods from the day after the Closing Date, all such interest credited to, withheld from and any early withdrawal penalties imposed upon the Deposit Liabilities. Any amounts required by any Governmental Authority to be withheld from any of the Deposit Liabilities through the Closing Date will be withheld by Seller in accordance with Applicable Law or appropriate notice from any Governmental Authority and will be remitted by Seller to the appropriate agency on or prior to the applicable due date. Any such withholding required to be made subsequent to the Closing Date will be withheld by Purchaser in accordance with Applicable Law or appropriate notice from any Governmental Authority and will be remitted by Purchaser to the appropriate agency on or prior to the applicable due date.
- (ii) Unless otherwise agreed by the parties, Seller will be responsible for delivering to payees all Internal Revenue Service notices with respect to information reporting and Tax identification numbers required to be delivered through the Closing Date with respect to the Deposit Liabilities, and Purchaser will be responsible for delivering to payees all such notices required to be delivered following the Closing Date with respect to the Deposit Liabilities.
- (iii) Unless otherwise agreed by the parties, Purchaser will timely make all required reports to applicable taxing authorities and to obligors on Loans, with respect to the period from January 1 of the year in which the Closing occurs through the Closing Date concerning all interest and points received by Seller, and Purchaser will timely make all required reports to applicable taxing authorities and to obligors on Loans, with respect to all periods from the day after the Closing Date concerning all such interest and points received. Seller shall provide to Purchaser information regarding interest and points received on Acquired Loans during the calendar year in which the Closing occurs through (and including) the Closing Date.
- (h) On the date that is thirty (30) days after the Closing Date, Seller will pay to Purchaser an amount equal to the lesser of (i) the amount of any overdraft which existed as of the Closing Date on each Deposit Account associated with Deposit Liabilities assumed by Purchaser hereunder; or (ii) the amount of any remaining overdraft which continues to exist on each Deposit Account associated with Deposit Liabilities which was originally overdrawn as of the close of business on the Business Day immediately preceding the Closing Date. If the overdraft has been reduced to zero at any time during such thirty (30) day period, no amount will be payable by Seller to Purchaser with respect to such overdraft.
- (i) If any Loan Documents are not in the possession or control of Seller on the Closing Date, Seller shall use commercially reasonable efforts to obtain those Loan Documents and deliver them to Purchaser as soon as practicable after the Closing Date.

(j) Seller and Purchaser will use their reasonable best efforts to cooperate with each other in assuring an orderly transition of ownership of the Assets and responsibility for the Assumed Liabilities assumed by Purchaser hereunder, including, without limitation, with respect to any items the transition of which is not specifically addressed in this Article 8.

# 8.03 <u>Post-Closing Adjustments.</u>

- (a) As promptly as reasonably practicable after the Closing Date, but no later than forty-five (45) days thereafter, Seller will provide Purchaser with a final settlement statement, in the form attached as Exhibit 8.03, which will reflect any adjustments that are necessary to accurately reflect the balances as shown on the financial records of Seller as of the close of business on the Closing Date. Concurrently therewith, Seller will amend Exhibits 2.02(a), 2.02(d), 2.02(e) and 2.04(e), which were provided to Purchaser at the Closing, to reflect any required adjustments; provided, however, that all such amendments will be subject to the dispute resolution process set forth in Section 8.03(b) below.
- (b) Purchaser will have the right to review any and all documents necessary or desirable to confirm the accuracy of final Exhibits 2.02(a), 2.02(d), 2.02(e) and 2.04(e). Notwithstanding any other provision of this Agreement, the final settlement statement will be prepared by Seller based in part upon a cash count to be mutually conducted by Seller and Purchaser at the close of business on the Closing Date. If, within fifteen (15) Business Days following the date of receipt by Purchaser of the final settlement statement, Purchaser does not dispute any items contained in the final settlement statement will be final and binding upon the parties. In the event Purchaser disputes any items contained in the final settlement statement, such disputes will be resolved in the following manner:
  - (i) Purchaser will notify Seller in writing (the "<u>Notice of Disagreement</u>") of such dispute within fifteen (15) Business Days after Purchaser's receipt of the final settlement statement, which notice will specify in reasonable detail the nature of the dispute, indicating those specific items that are in dispute (the "<u>Disputed Items</u>"). Any items in the final settlement statement that are not Disputed Items will be final, binding and conclusive for all purposes hereunder.
  - (ii) During the thirty (30) day period following Seller's receipt of a Notice of Disagreement from Purchaser, Seller and Purchaser will use their reasonable best efforts to resolve the Disputed Items. If, at the end of such thirty (30) day period, the parties have reached written agreement with respect to all or a portion of the matters covered by a Notice of Disagreement, the final settlement statement will be adjusted to reflect such mutual written agreement and will become final, binding and conclusive upon the parties hereto with respect to all matters that are not Unresolved Changes.
  - (iii) If, at or before the end of the thirty (30) day period specified in Section 8.03(b)(ii) above, Purchaser and Seller have failed to reach a written agreement with respect to all or a portion of such Disputed Items (those Disputed Items that remain in dispute at the end of such period are the "<u>Unresolved Changes</u>"), then Purchaser and Seller will promptly refer the Unresolved Changes to Plante Moran, or, in the event such accounting firm refuses or is unable to make a determination, a mutually agreeable nationally or regionally recognized independent certified public accounting firm with an established practice in the audit of banks (the "Firm") to make a determination as to the subject matter of the Unresolved Changes. If Purchaser and Seller fail to agree on a Firm within fifteen (15) days after the end of the thirty (30) day period specified in this Section 8.03(b)(iii), above, the Firm will be selected by Plante Moran. The Firm will be directed to issue its written decision regarding the Unresolved Changes as promptly as practicable and in any event within thirty (30) days following the submission of the Unresolved Changes to the Firm for resolution, and such decision, and the final settlement statement as adjusted to reflect the Firm's determination, will be final, binding and conclusive on the parties. Seller and Purchaser agree to fully cooperate with and provide any information requested by such Firm. In the event Unresolved Changes are submitted to the Firm for resolution as provided herein, the fees, charges and expenses of the Firm will be borne and paid equally by Purchaser and Seller.

(c) If the Final Payment Amount shown on the final settlement statement as finalized pursuant to Section 8.03(b) above is different from the Cash Payment made at Closing, then a payment will be made in the following manner: (i) if the Cash Payment was greater than the Final Payment Amount, the party receiving the Cash Payment will refund the difference between such amounts; and (ii) if the Cash Payment was less than the Final Payment Amount, the party making the Cash Payment will pay the difference between such amounts. Such payment will be made by wire transfer of immediately available funds, together with interest thereon for the number of days from and including the Closing Date to such settlement date, but excluding such settlement date, at the rate per annum equal to the average during such period of the average of the daily high and low rates for federal funds on each Business Day during such period, as such rates are published in the Eastern Edition of the Wall Street Journal, computed on the basis of a 365-day year.

#### 8.04 Post-Closing Reconciliation.

(a) Inclearing Items. As of the opening of business on the Closing Date, Seller will advise the Federal Reserve Bank of Cleveland (the "Federal Reserve") that the account number ranges of the routing transit number associated with the Branches should be reassigned to Purchaser and will further provide the information in order to expedite the clearing and sorting of all checks, drafts instruments and other commercial paper relative to the Deposit Liabilities and/or the Acquired Loans (hereinafter collectively referred to as the "Paper Items"). Purchaser will bear all charges and costs imposed by the Federal Reserve in connection with reassignment of account number ranges for sorting the Paper Items. In the event the Federal Reserve fails or refuses to direct sort such Paper Items for delivery to Purchaser with the result that such Paper Items are presented to Seller, for a period of sixty (60) days following the Closing Date (the "Inclearing Period"), by 5:00 p.m. Eastern Time of every banking day during the Inclearing Period, Seller will transmit to Purchaser a list of inclearing items, setting forth the amount and the Seller account number of each item received by Seller for payment that day. Seller will deliver such information to Purchaser in the form of an X9 file or, if such format is not reasonably practicable, in paper form via courier, the costs and expenses of which will be paid by Seller but promptly reimbursed to Seller by Purchaser. Upon expiration of the Inclearing Period, Seller will cease forwarding to Purchaser items presented against the Deposit Liabilities and such items will be returned marked "Refer to Maker." Magnetic tape or other written media will utilize the most recent account number designated by Seller for each of the Deposit Accounts. Seller and Purchaser will settle amounts due under this Section 8.04(a) by wire transfer.

(b) <u>ACH Transactions</u>. Seller will provide to Purchaser no later than sixty (60) days prior to the Closing Date, the customer name, customer account number and the originator, originator identification number and receiver for each ACH entry for the Deposit Accounts and will further provide, within seven days following the Closing Date, an updated list as of the close of business on the Closing Date. For a period of ninety (90) days following the Closing Date, Seller agrees to continue to accept and immediately forward to Purchaser all ACH entries and corresponding funds. Seller also agrees to include the originator identification number, and Purchaser agrees to immediately notify and instruct the originator of the ACH to reroute the entries directly to Purchaser.

- (c) <u>Direct Debits</u>. For a period of ninety (90) days following the Closing, Seller will transfer to Purchaser all received direct debits on accounts constituting Deposit Liabilities each Business Day in accordance with Seller's customary procedures. Purchaser will send Notice of Change compliant with the Nacha Operating Rules of the National Automated Clearinghouse Association on each direct debit received. Thereafter, Seller may discontinue forwarding such entries and return them to the originators marked "Account Sold."
- (d) Over-the-Counter Returned Items. For a period of sixty (60) days following the Closing Date, Seller will by facsimile or secure e-mail provide Purchaser with a list of over-the-counter returned items on the day they are received by Seller. Over-the-counter returned items are those items that are included within the Deposit Liabilities transferred to Purchaser but that are returned unpaid to Seller after the Closing Date. By 11:00 a.m. Eastern Time on the next banking day after receipt of an over-the-counter returned item, Purchaser will notify Seller of any unacceptable returned items due to non-sufficient funds. On the same banking day Purchaser transmits the list, Seller will send only items deemed acceptable by Purchaser, by overnight mail to Purchaser for "next banking day" delivery. On the same day, Purchaser will transmit to Seller in immediately available funds by wire transfer, the sum of the accepted over-the-counter returned items.

#### 9. GENERAL COVENANTS AND INDEMNIFICATION.

- 9.01 <u>Information After Closing</u>. For a period of seven (7) years following the Closing, upon written request of the other party, a party will provide the other party with reasonable access to, or copies of, information and records relating to the Branches which is in the possession or control of such party reasonably necessary to permit the other party or any of its subsidiaries, successors or Affiliates to comply with or contest any applicable legal, tax, banking, accounting, or regulatory policies or requirements, or any legal or regulatory proceeding thereunder. Except as otherwise provided in Section 9.04 or Section 9.05 of this Agreement, the requesting party will reimburse the other party for all third-party, out-of-pocket expenses actually incurred by the other party in connection therewith.
- 9.02 <u>Confidentiality Obligations of Purchaser</u>. From and after the date hereof, except to the extent otherwise required by Applicable Law, Purchaser will, and will cause its subsidiaries and Affiliates (for all purposes of this Section 9.02, including their respective directors, officers, employees and advisers) to, treat all information received from Seller concerning the business, assets, operations, financial condition and customers of Seller (including, without limitation, the Branches) as confidential, unless and to the extent that Purchaser can demonstrate that such information was already known to Purchaser or such subsidiaries or Affiliates prior to being provided to Purchaser by Seller or in the public domain, and Purchaser will not, and will cause its subsidiaries and Affiliates to not, use any such information required by this Agreement to be treated as confidential for any purpose except in furtherance of the transactions contemplated hereby. The covenants of Purchaser contained in this Section 9.02 are of the essence and will survive any termination of this Agreement but will terminate at the Closing with respect to any information that is limited to the Branches, the Assets, or the Assumed Liabilities.
- 9.03 Confidentiality Obligations of Seller. From and after the date hereof, except to the extent otherwise required by Applicable Law, Seller will, and will cause its subsidiaries and Affiliates (for all purposes of this Section 9.03, including Seller's and their respective directors, officers, employees and advisers) to, treat all information received from Purchaser concerning the business, assets, operations and financial condition of Purchaser as confidential, unless and to the extent Seller can demonstrate that such information was already known to Seller or such subsidiaries or Affiliates or in the public domain, and Seller will not, and will cause its subsidiaries and Affiliates to not, use any such information for any purpose except in furtherance of the transactions contemplated hereby. From and after the Closing Date, Seller will, and will cause its subsidiaries and Affiliates (for all purposes of this Section 9.03, including Seller's and their respective directors, officers, employees and advisers) to, treat all Seller information related to the Assets and Assumed Liabilities as confidential information of Purchaser, unless and to the extent Seller can demonstrate that such information was already in the public domain, and Seller will not, and will cause its subsidiaries and Affiliates to not, use any such information for any purpose except in a manner consistent with the transactions contemplated hereby.

- 9.04 <u>Indemnification by Seller</u>. Seller will indemnify, hold harmless and defend Purchaser and its subsidiaries and Affiliates (for all purposes of this Section 9.04, including their respective directors, officers, employees and advisers) from and against any and all Losses that they may suffer, incur or sustain arising out of or in connection with:
- (a) any breach of representation or warranty of Seller in this Agreement or in any certificate or instrument delivered in connection herewith:
- (b) any breach of covenant or agreement on the part of Seller under this Agreement or in any certificate or instrument delivered in connection herewith;
- (c) any Legal Proceeding (or any investigation or inquiry by or before a Governmental Authority) to the extent arising from Seller's operation of the Branches or Seller's ownership or use of the Assets or Assumed Liabilities prior to the close of business on the Closing Date; or
  - (d) the Excluded Liabilities or Excluded Items.
- 9.05 <u>Indemnification by Purchaser</u>. Purchaser will indemnify, hold harmless and defend Seller and its subsidiaries and Affiliates (for all purposes of this Section 9.05, including their respective directors, officers, employees and advisers) from and against any and all Losses that they may suffer, incur or sustain arising out of or in connection with:
- (a) any breach of representation or warranty of Purchaser in this Agreement or in any certificate or instrument delivered in connection herewith;
- (b) any breach of covenant or agreement on the part of Purchaser under this Agreement or in any certificate or instrument delivered in connection herewith;
- (c) any Legal Proceeding (or any investigation or inquiry by or before a Governmental Authority) to the extent arising from Purchaser's operation of the Branches or Purchaser's ownership or use of the Assets or Assumed Liabilities after the close of business on the Closing Date; or
  - (d) the Assumed Liabilities.

#### 9.06 <u>Claims for Indemnity</u>.

(a) A claim for indemnity under Section 9.04 or 9.05 of this Agreement may be made by the claiming party at any time during the applicable survival period set forth in Section 12.02, by the giving of written notice thereof to the other party. Such written notice will set forth in reasonable detail the basis upon which such claim for indemnity is made. In the event that any such claim is made within the prescribed period, the indemnity relating to such claim will survive until such claim is resolved. Claims not made within such period will cease and no indemnity will be made therefor.

- In the event that any Person not a party to this Agreement makes any demand or claim or files or threatens to file any lawsuit, which demand, claim or lawsuit may result in any Losses to one party hereto of the kind for which such party is entitled to indemnification pursuant to Section 9.04 or 9.05 hereof, then, after written notice is provided by the indemnified party to the indemnifying party of such demand, claim or lawsuit, the indemnifying party will have the option, at its cost and expense, to retain counsel for the indemnified party to defend any such demand, claim or lawsuit; provided, however, that the indemnified party will be entitled to participate in any such defense with separate counsel, at the indemnifying party's cost and expense, if (i) so requested by the indemnifying party to participate, (ii) the indemnified party determines in good faith that a conflict or potential conflict exists between the interests of the indemnified party and the indemnifying party that would make such separate representation advisable, or (iii) such demand, claim or lawsuit is based upon or otherwise involves an investigation, inquiry, or other proceeding by a Governmental Authority. In the event that the indemnifying party fails to respond within ten (10) calendar days after receipt of such notice of any such demand, claim or lawsuit, then the indemnified party may retain counsel and conduct the defense of such demand, claim or lawsuit as it may in its discretion deem proper, at the cost and expense of the indemnifying party. In effecting the settlement of any such demand, claim or lawsuit, an indemnified party will act in good faith, will consult with the indemnifying party and will enter into only such settlement as the indemnifying party will approve (the indemnifying party's approval will be implied if it does not respond within ten (10) calendar days of its receipt of the notice of such settlement offer), such approval not to be unreasonably withheld. In addition, in the event that the indemnifying party exercises the foregoing option to defend any such demand, claim or lawsuit, in effecting the settlement of any such demand, claim or lawsuit, an indemnifying party will act in good faith, will consult with the indemnified party and will enter into only such settlement as the indemnified party will approve (the indemnified party's approval will be implied if it does not respond within ten (10) calendar days of its receipt of the notice of such settlement offer), such approval not to be unreasonably withheld, provided, however, that the indemnified party shall not be required to approve any settlement, compromise, discharge or consent to an entry of judgment in each case if it (i) obligates the indemnifying party to pay the full amount of the liability in connection with such claim and unconditionally releases the indemnified party and its Affiliates from all liability or obligation in connection with such claim and (ii) does not impose injunctive or other non-monetary relief against the indemnified party or its Affiliates, or their respective businesses.
- 9.07 <u>Limitations on Indemnification</u>. Notwithstanding anything to the contrary contained in this Section 9, no indemnification shall be required to be made by either party until the aggregate amount of all such claims by a party exceeds \$25,000. Once such aggregate amount exceeds \$25,000, such party shall thereupon be entitled to indemnification for all amounts in excess of such \$25,000. Notwithstanding the foregoing, such limitation shall not apply to reimbursement of third-party, out-of-pocket expenses paid or incurred by a party for which this Agreement provides for reimbursement or for Closing or post-Closing settlement. IN ADDITION, THE PARTIES SHALL HAVE NO OBLIGATIONS UNDER THIS SECTION 9 FOR ANY CONSEQUENTIAL LIABILITY, DAMAGE OR LOSS THE INDEMNIFIED PARTY MAY SUFFER AS THE RESULT OF ANY DEMAND, CLAIM OR LAWSUIT.
- 9.08 <u>Calculation of Payments</u>. Indemnification payments under Section 9.04 or 9.05 shall be calculated in a manner consistent with the definition herein of Losses.
- 9.09 Solicitation of Customers by Purchaser Prior to Closing. At any time prior to the Closing Date, Purchaser will not, and will not permit any of its Affiliates to, conduct any marketing, media or customer solicitation campaign or customer calling program which is specifically targeted to induce customers whose Deposit Liabilities are to be assumed by Purchaser pursuant to this Agreement to open accounts with Purchaser prior to the Closing Date and/or to discontinue their account relationships with Seller, except: (i) as may occur in connection with advertising or solicitations directed to the public generally; (ii) as may occur as a result of the solicitations by it or its Affiliates of deposits or loans from or to a major or statewide depositor such as, for example, a company with more than one location or a local, state or federal government; (iii) as may occur as a result of the solicitation by it or its Affiliates of deposits or loans outside of the county within which a Branch is situated; or (iv) as may occur as a result of any lending, deposit, trust, credit card or other banking relationship existing between Purchaser and such customer prior to the effective date of this Agreement; provided, however, that nothing herein will prevent Purchaser or its Affiliates from providing any notice required by any Governmental Authority or by any law, rule or regulation in connection with the Acquisition or any notice required or otherwise expressly permitted under this Agreement.

- 9.10 Solicitation of Customers by Seller After Closing. On and after the Closing Date and ending on that date which is twenty-four (24) months after the Closing Date, Seller will not, and will not permit any of its Affiliates to, (a) conduct any marketing, media or customer solicitation campaign which is specifically targeted to induce customers whose Deposit Liabilities or Acquired Loan are to be assumed or purchased by Purchaser pursuant to this Agreement to discontinue their account relationships with Purchaser or (b) otherwise solicit deposits, loans or other banking business from the customers whose Accounts are transferred to Purchaser pursuant to this Agreement, except: (i) as may occur in connection with advertising or solicitations directed to the public generally; (ii) as may occur as a result of the solicitations by it or its Affiliates of deposits or loans from or to a major or statewide depositor such as, for example, a company with more than one location or a local, state or federal government; (iii) as may occur as a result of the solicitation by it or its Affiliates of deposits or loans outside of the county within which the Branch is situated; or (iv) as may occur as a result of any lending, deposit, trust, credit card or other banking relationship existing prior to the Closing Date and attributable to or domiciled at any branch offices or other facilities of it or its Affiliates other than the Branches; provided, however, that nothing herein will prevent Seller or its Affiliates from providing any notice required by any Governmental Authority or by any law, rule or regulation in connection with the Acquisition. Seller agrees not to operate a branch banking office or loan production office within a twenty-five (25) mile radius of either of the Branches until the expiration of the twenty-four (24) month period, but nothing herein will prohibit Seller from acquiring a branch or loan production office through the merger of Seller and/or an Affiliate with, or the acquisition by Seller and/or an Affiliate of, a third party which has a branch or loan production office located within 25 miles of either of the Branches and also has branches outside of such 25 mile radius.
- 9.11 <u>Further Assurances</u>. From and after the date hereof, each party hereto agrees to execute and deliver such instruments and to take such other actions as the other party hereto may reasonably request in order to carry out and implement this Agreement. Without limiting the foregoing, Seller agrees to execute and deliver such deeds, bills of sale, acknowledgements and other instruments of conveyance and transfer as will be necessary and appropriate to vest in Purchaser the legal and equitable title to the Assets of Seller being conveyed to Purchaser hereunder. The covenants of each of the parties hereto pursuant to this Section 9.11 will survive the Closing.

#### 10. TERMINATION.

- 10.01 <u>Termination by Mutual Agreement</u>. This Agreement may be terminated and the transactions contemplated hereby may be abandoned by the mutual written agreement of each of Seller and Purchaser.
- 10.02 <u>Termination by Seller</u>. This Agreement may be terminated by Seller and the transactions contemplated hereby abandoned by Seller:
- (a) in the event of a material breach by Purchaser of this Agreement and, if capable of being cured, such breach has not been cured by Purchaser by the earlier of thirty (30) calendar days after the giving of notice to Purchaser of such breach or the Closing Date;

- (b) in the event any of the conditions precedent specified in Section 6.01 of this Agreement has not been met as of the date required by this Agreement and, if not so met, has not been waived in writing by Seller, provided that Seller's failure to perform an obligation in breach of its obligations under this Agreement was not the cause of, or did not result in, Purchaser's failure to fulfill such condition; or
- (c) in the event any regulatory approval required for consummation of the Acquisition is denied by the applicable regulatory authority or has not been received within six (6) months from the date of this Agreement.
- 10.03 <u>Termination by Purchaser</u>. This Agreement may be terminated by Purchaser and the transactions contemplated hereby abandoned by Purchaser:
  - (a) in accordance and consistent with Section 3.02(c);
- (b) in the event of a material breach by Seller of this Agreement and such breach, if capable of being cured, has not been cured by Seller by the earlier of thirty (30) calendar days after the giving of notice to Seller of such breach or the Closing Date;
- (c) in the event any of the conditions precedent specified in Section 6.02 of this Agreement has not been met as of the date required by this Agreement and, if not so met, has not been waived in writing by Purchaser, provided that Purchaser's failure to perform an obligation in breach of its obligations under this Agreement was not the cause of, or did not result in, Seller's failure to fulfill such condition;
- (d) in the event any regulatory approval required for consummation of the Acquisition is denied by the applicable regulatory authority or has not been received within six (6) months from the date of this Agreement;
  - (e) in accordance and consistent with Section 3.01(c)(iii); or
  - (f) in accordance and consistent with Section 3.01(f).
- 10.04 Effect of Termination. In the event of termination pursuant to this Article 10, and except as otherwise stated herein, written notice thereof will be given to the other party, and this Agreement will terminate immediately upon receipt of such notice unless an extension is consented to by the party having the right to terminate, provided that Sections 9.02, 9.03, 9.04, 9.05, 9.06, 9.07, 9.08, 12.03 and 12.05 will survive any such termination and the termination of this Agreement pursuant to this Article 10 will not release any party hereto from any liability or obligation to the other party hereto arising from a breach of any provision of this Agreement occurring prior to the termination hereof. In addition to any other remedies to which Seller may be entitled, in the event that Purchaser terminates this Agreement other than as permitted pursuant to this Article 10, or in the event Seller terminates this Agreement pursuant to Section 10.02(a), Purchaser will promptly reimburse Seller for any third-party, out-of-pocket expenses incurred or paid by Seller, including those in connection with the conversion of accounts, to the extent such amounts were reasonably required to be paid or incurred prior to the Closing but which are, pursuant to this Agreement, the responsibility of Purchaser.

## 11. REPURCHASE OF DEFECTIVE LOANS

- 11.01 <u>Notice of Defective Loans to Seller</u>. Upon discovery by Purchaser or Seller of an Acquired Loan that breaches any representation or warranty made by Seller in this Agreement (a "<u>Defective Loan</u>"), the party that discovered such Defective Loan will provide written notice within fifteen (15) Business Day of discovering such defect to the other party (if given by Purchaser, the "<u>Purchaser's Notice</u>").
- 11.02 <u>Right to Cure</u>. Seller will have the right and option to cure a Defective Loan, which is reasonably curable; provided, that such a cure is completed to the reasonable satisfaction of Purchaser within thirty (30) calendar days of Seller's discovery of the Defective Loan or Seller's receipt of Purchaser's Notice referenced in Section 11.01, or such shorter period as may be required by Applicable Law (the "<u>Cure Period</u>").
- 11.03 Obligation to Repurchase. Upon Purchaser's written request, Seller will be obligated to repurchase a Defective Loan that is not cured prior to the expiration of the Cure Period at the purchase price of such Acquired Loan plus (a) accrued interest and fees to the date of repurchase, less (b) interest and fees paid as of the date of repurchase (the "Repurchase Price"). Seller will repurchase the Defective Loan within fifteen (15) Business Days of the expiration of the Cure Period at the Repurchase Price. Notwithstanding anything to the contrary contained herein, Seller shall have no obligation to repurchase any Defective Loan hereunder if any of the following has occurred with respect to such Acquired Loan following the Closing: (i) any modification, extension, forgiveness or other material change in the terms or conditions of such Acquired Loan, including, without limitation, any change in the term, interest rate or method of computation of interest (other than interest adjustments required under the terms of the Acquired Loan); (ii) any release of collateral, in whole or in part (other than a release of collateral made in connection with the substitution of new collateral of equivalent value or a release of collateral in connection with a reduction by cash payment in the outstanding principal balance of the Acquired Loan, provided that after such payment the balance of the Acquired Loan is fully secured by a valid first lien), foreclosure, or other change in the collateral position which Seller held at the Closing Date with respect to such Acquired Loan; or (iii) any prior sale, transfer, pledge, hypothecation or assignment, in whole or in part, by Purchaser of any right, title or interest in or to such Acquired Loan (including, without limitation, any interest thereon or any collateral related thereto) to any third party without the express prior written consent of Seller.
- 11.04 <u>Reassignment to Seller</u>. Upon any repurchase of a Defective Loan hereunder, Purchaser will endorse over, reassign, deliver and transfer to Seller (a) all portions of the Defective Loan and related note with appropriate completed endorsements, which will be without recourse and without representation or warranty, so as to vest Seller with title to the reassigned loan; and (b) deliver to Seller a recordable assignment of any applicable mortgage, unless the defect in the Defective Loan prevents compliance with (a) and (b). Purchaser will execute such other documents as Seller may reasonably request to accomplish the reassignment and transfer of any loan and any related mortgage or other document or instrument pursuant to this Section 11.04.
- 11.05 <u>General</u>. The rights granted to Purchaser in this Article 11 shall be the exclusive rights and remedies of Purchaser with respect to any Defective Loans and any breaches of any representations and warranties made by Seller with respect to the Acquired Loans, and Purchaser shall not be entitled to any rights to indemnification under Section 9.04 of this Agreement with respect thereto.
- 11.06 <u>Limitation on Repurchase Obligation</u>. Seller's repurchase obligations under this Article 11 will expire on the nine (9) month anniversary of the Closing Date. Notwithstanding the foregoing, any repurchase pending under this Article 11 prior to the nine (9) month anniversary of the Closing Date will be unaffected and may be completed pursuant to the terms hereof after such date. Thereafter, Seller will have no obligation to repurchase any Acquired Loans unless otherwise required by Applicable Law.

#### 12. MISCELLANEOUS PROVISIONS.

#### 12.01 Tax Matters.

- (a) Any sales, use, excise, value added, goods and services, transfer, recording, documentary, registration, conveyancing and other similar Taxes that may be imposed on the sale and transfer of the Assets (including any stamp, duty or other tax chargeable in respect of any instrument transferring property and any recording fees or expenses payable in connection with the sale and transfer of the any property) (together with any and all penalties, interest and additions to tax with respect thereto, "<u>Transfer Taxes</u>.") shall be borne by Purchaser. Purchaser and Seller shall cooperate in timely making all filings, returns, reports and forms as may be required to comply with the provisions of Applicable Law in connection with the payment of any such Transfer Taxes. Purchaser and Seller shall cooperate in providing each other with appropriate resale exemption certification and other similar tax and fee documentation.
- (b) All Tax liabilities attributable to the Assets for any Pre-Closing Tax Periods shall be borne by Seller, and all Tax liabilities attributable to the Assets for any Post-Closing Tax Periods shall be borne by Purchaser. All refunds of Taxes (including any interest thereon) attributable to the Assets for any Pre-Closing Tax Periods shall be for the benefit of Seller, and all refunds of Taxes (including any interest thereon) attributable to the Assets for any Post-Closing Tax Periods shall be for the benefit of Purchaser.
- (c) With respect to Taxes (other than income taxes) attributable to the Assets that are payable for a taxable period that begins on or before the Closing Date and ends after the Closing Date (a "Straddle Period"), the portion of such Taxes included in the Pre-Closing Tax Period shall (i) in the case of any real property taxes, personal property taxes and similar *ad valorem* obligations, be deemed to be the amount of any such Tax for the entire taxable period multiplied by a fraction the numerator of which is the number of days in the entire taxable period ending at the close of business on the Closing Date and the denominator of which is the number of days in the entire taxable period, and (ii) in the case of any Taxes based upon or related to sales, wages, capital expenditures or expenses, be deemed equal to the amount that would be payable if the taxable period ended at the close of business on the Closing Date.
- (d) Any refund of Taxes (including any interest thereon) for a Straddle Period will be allocated between the Pre-Closing Tax Period and the Post-Closing Tax Period in accordance with the principles of Section 12.01(c).
- (e) If either party pays any Taxes to be borne by the other party under this Section 12.01, the other party shall promptly reimburse such party for the Taxes paid. If either party receives any refunds or credits which are the property of the other party under this Section 12.01, such party shall promptly pay the amount of such refunds or credits to the other party.

# 12.02 <u>Survival of Representations, Warranties and Covenants.</u>

- (a) <u>Survival of Representations and Warranties</u>. The respective representations and warranties of Seller and Purchaser contained or referred to in this Agreement, including in Article 4 of this Agreement, or in any certificate, exhibit or other instrument delivered or to be delivered pursuant to this Agreement, will survive for twelve (12) months after the Closing, except that:
  - (i) the express representations and warranties of Seller contained or referred to in Sections 4.01(a) (Existence and Power of Seller), 4.01(b) (Authorization of Agreement), 4.01(c) (Effective Agreement) and 4.01(d) (Title to Assets) of this Agreement will survive for sixty (60) months after the Closing; and

- (ii) the express representations and warranties of Purchaser contained or referred to in Sections 4.02(a) (Good Standing and Power of Purchaser), 4.02(b) (Authorization of Agreement) and 4.02(c) (Effective Agreement) of this Agreement will survive for sixty (60) months after the Closing.
- (b) <u>Survival of Covenants and Agreements</u>. The respective covenants and other agreements of the parties contained or referred to in this Agreement or in any certificate, exhibit or other instrument delivered or to be delivered pursuant to this Agreement shall survive in perpetuity, unless and to the extent that such covenant or other agreement is limited by a specific period of time expressly set forth therein or non-compliance with such covenant or other agreement is waived in writing by the party entitled to such performance.
- 12.03 <u>Limitation and Disclaimer of Representations and Warranties</u>. Except as expressly stated in this Agreement, no party to this Agreement makes any representations or warranties to any other party hereto. Without limiting the foregoing, each party hereto specifically disclaims any and all representations and warranties not expressly stated in this Agreement, including, without limitation: (a) any representation or warranty implied by law and (b) any representation or warranty concerning the amount of deposits that the Branches will have at, or after, the Closing, whether any customers of the Branches will remain customers of Purchaser after the Closing, or the financial performance of the Branches after the Closing.
- 12.04 <u>Waivers</u>. Each party hereto, by written instrument signed by an officer of such party, may extend the time for the performance of or waive any of the obligations or other acts or any provision of this Agreement, as it affects the party signing such instrument.
- 12.05 <u>Notices</u>. Any notice or other communication required or permitted pursuant to this Agreement will be effective only if it is in writing and delivered personally, by facsimile transmission, or by registered or certified return receipt mail, postage prepaid, addressed as follows:

If to Purchaser:	Consumers National Bank		
	P.O. Box 256		
	614 E. Lincoln Way		
	Minerva, Ohio 44657		
	Attention: Ralph J. Lober, II		
	President and CEO		
	Fax: ()		
	With a copy to:		
	Glenn E. Morrical		
	Tucker Ellis LLP		
	950 Main Avenue, Suite 1100		
	Cleveland, Ohio 44113		
	Fax: (216) 592-5009		

-42-

If to Seller: CFBank, National Association

7000 N. High St.

Worthington, Ohio 43085 Attention: Timothy T. O'Dell,

President and CEO Fax: (614) 505-5697

(- )----

With a copy to:

Anthony D. Weis Vorys, Sater, Seymour and Pease LLP 52 East Gay St. Columbus, Ohio 43215

Fax: (614) 719-4776

or such other Person or address as any such party may designate by notice to the other party, and will be deemed to have been given as of the date received.

- 12.06 Parties in Interest; Assignment; Amendment. This Agreement is binding upon and is for the benefit of the parties hereto and their respective successors, legal representatives and assigns, and no Person who is not a party hereto (or a successor or assignee of such party) will have any rights or benefits under this Agreement, either as a third party beneficiary or otherwise except with respect to rights to indenmnification as provided in Section 9.04. This Agreement cannot be assigned, and this Agreement cannot be amended or modified, except by a written agreement executed by the parties hereto or their respective successors or permitted assigns.
- 12.07 <u>Headings</u>. The headings in this Agreement are inserted for convenience of reference only and are not intended to be a part of or to affect the meaning or interpretation of this Agreement.
- 12.08 Terminology. The specific terms of art that are defined in various provisions of this Agreement will apply through this Agreement (including, without limitation, each exhibit hereto), unless expressly indicated otherwise. Unless expressly indicated otherwise in a particular context, the terms "herein," "hereunder," "hereto," "hereof" and similar references refer to this Agreement in its entirety and not to specific Articles, Sections, exhibits or Subsections of this Agreement. Unless expressly indicated otherwise in a particular context, references in this Agreement to enumerated Articles, Sections and Subsections refer to designated portions of this Agreement (but do not refer to portions of any exhibit unless such exhibit is specifically referenced) and do not refer to any other document.
- 12.09 <u>Flexible Structure</u>. References in this Agreement to federal or state laws or regulations, jurisdictions or chartering or regulatory authorities will be interpreted broadly to allow maximum flexibility in consummating the transactions contemplated hereby in light of changing business, economic and regulatory conditions. Without limiting the foregoing, in the event Seller and Purchaser agree in writing to alter the legal structure of the Acquisition contemplated by this Agreement, references in this Agreement to such laws, regulations, jurisdictions and authorities will be deemed to be altered to reflect the laws, regulations, jurisdictions and authorities that are applicable in light of such change.
- 12.10 <u>Press Releases</u>. Seller and Purchaser will consult with each other concerning the form and substance of any press release or other public disclosure of any matters relating to this Agreement. Seller and Purchaser agree that, except as provided in Section 3.01(e), no announcements to or communications with the Branch Employees relating to this Agreement will be made before the Closing without the prior approval of both Seller and Purchaser. Notwithstanding the foregoing provisions of this Section 12.10, nothing in this Section 12.10 will prevent any party or its Affiliates from providing any notice required by any Governmental Authority or by any law, rule or regulation in connection with the Acquisition or any notice required or otherwise expressly permitted under this Agreement.

- 12.11 <u>Entire Agreement</u>. This Agreement supersedes any and all oral or written agreements and understanding heretofore made relating to the subject matter hereof. All exhibits and appendices to this Agreement are incorporated into this Agreement by reference and made a part hereof.
- 12.12 Governing Law and Jurisdiction. This Agreement will be governed by, and construed in accordance with, the laws of the State of Ohio, without regard to its conflicts of law provisions. Any proceeding arising out of or relating to this Agreement may be brought only in the courts of the State of Ohio, County of Franklin, or, if it has or can acquire jurisdiction, in the United States District Court for the Southern District of Ohio, and each of the parties irrevocably submits to the exclusive jurisdiction of each such court in any such proceeding, waives any objection it may now or hereafter have to venue or to convenience of forum, agrees that all claims in respect of the proceeding shall be heard and determined only in any such court and agrees not to bring any proceeding arising out of or relating to this Agreement in any other court.
- 12.13 <u>Counterparts</u>. This Agreement may be executed in several counterparts and by facsimile or electronic transmission, each of which will be deemed an original, but all of which together will constitute one and the same instrument.

[Remainder of Page Intentionally Left Blank]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed by their respective officers thereunto duly authorized, all as of the date first above written.

#### **SELLER:**

CFBANK, NATIONAL ASSOCIATION

By: /s/Timothy T. O'Dell
Name: Timothy T. O'Dell
Title: President and CEO

# **PURCHASER:**

CONSUMERS NATIONAL BANK

By: /s/Ralph J. Lober, II
Name: Ralph J. Lober, II
Title: President and CEO

-45-

Schedules and similar attachments Regulation S-K. See page iii of the Tai The Registrant will supply a copy o	ble of Contents of this ag	reement for a list and l	brief description of the	omitted attachments.

#### EXHIBIT 31.1

# I, Ralph J. Lober, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of Consumers Bancorp, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a 15(e) and 15d 15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - a. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - b. Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - c. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - d. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors:
  - a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

February	12,	2021
Date		

By: /s/ Ralph J. Lober

Ralph J. Lober, II

President & Chief Executive Officer

#### EXHIBIT 31.2

## I, Renee K. Wood, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of Consumers Bancorp, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a 15(e) and 15d 15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - a. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - b. Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - c. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - d. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors:
  - a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

February 12, 2021	By: /s/ Renee K. Wood
Date	Renee K. Wood
	Chief Financial Officer & Treasurer

#### Exhibit 32.1

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report of Consumers Bancorp, Inc. (the "Corporation") on Form 10-Q for the period ended December 31, 2020 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), each undersigned officer of the Corporation does hereby certify that:

- a) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934, as amended; and
- b) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Corporation.

Date: February 12, 2021

/s/ Ralph J. Lober

Ralph J. Lober, II President & Chief Executive Officer

/s/ Renee K. Wood

Renee K. Wood Chief Financial Officer & Treasurer

# **Document And Entity Information - shares**

# 6 Months Ended Dec. 31, 2020

Feb. 10, 2021

**Document Information [Line Items]** 

Entity Registrant Name CONSUMERS BANCORP INC /OH/

Entity Central Index Key 0001006830 Current Fiscal Year End Date --06-30

Entity Filer Category Non-accelerated Filer

Entity Current Reporting StatusYesEntity Emerging Growth CompanyfalseEntity Small BusinesstrueEntity Interactive Data CurrentYes

Entity Common Stock, Shares Outstanding (in shares) 3,028,100

Entity Shell Company false
Document Type 10-Q

Document Period End Date Dec. 31, 2020

Document Fiscal Year Focus2021Document Fiscal Period FocusQ2Amendment Flagfalse

Consolidated Balance Sheets (Current Period Unaudited) - USD (\$) \$ in Thousands	Dec. 31, 2020	Jun. 30, 2020
<u>ASSETS</u>		
Cash on hand and noninterest-bearing deposits in financial institutions	\$ 8,996	\$ 8,429
Federal funds sold and interest-bearing deposits in financial institutions	49	1,230
Total cash and cash equivalents	9,045	9,659
Certificates of deposit in other financial institutions	8,599	11,635
Securities, available-for-sale	145,665	143,918
Securities, held-to-maturity (fair value of \$3,605 at December 31, 2020 and \$3,868 at June 30, 2020)	3,336	3,541
Equity securities, at fair value	400	
Federal bank and other restricted stocks, at cost	2,472	2,472
Loans held for sale	4,770	3,507
<u>Total loans</u>	557,257	542,861
Less allowance for loan losses	(5,912)	(5,678)
Net loans	551,345	537,183
Cash surrender value of life insurance	9,573	9,442
Premises and equipment, net	14,972	14,901
Goodwill	836	836
Core deposit intangible, net	242	256
Accrued interest receivable and other assets	2,805	3,470
<u>Total assets</u>	754,060	740,820
<u>LIABILITIES</u>		
Noninterest-bearing demand	194,180	190,233
Interest bearing demand	104,777	99,173
<u>Savings</u>	241,854	228,567
<u>Time</u>	107,551	115,382
<u>Total deposits</u>	648,362	633,355
Short-term borrowings	13,275	6,943
Federal Home Loan Bank advances	18,083	31,161
Accrued interest and other liabilities	6,632	6,121
<u>Total liabilities</u>	686,352	677,580
Commitments and contingent liabilities		
SHAREHOLDERS' EQUITY		
Preferred stock (no par value, 350,000 shares authorized, none outstanding)	0	0
Common stock (no par value, 8,500,000 shares authorized; 3,124,053 shares issued as of	20.011	10.074
December 31, 2020 and June 30, 2020)	20,011	19,974
Retained earnings	44,492	40,460
Treasury stock, at cost (95,953 and 108,475 common shares as of December 31, 2020 and	(1,324)	(1,454)
June 30, 2020, respectively)	(1,347)	(1,727)
Accumulated other comprehensive income	4,529	4,260
Total shareholders' equity	67,708	63,240

Total	liabilities	and s	hareho	lders <sup>2</sup>	equity
IOui	Habitities	, und	ilul Clio.	IGCID	equity

\$ 754,060 \$ 740,820

# Consolidated Balance Sheets (Current Period Unaudited) (Parentheticals) - USD (\$)

\$ / shares in Thousands, \$ in

# Dec. 31, 2020 Jun. 30, 2020

# **Thousands**

Held-to-maturity securities, fair value	\$ 3,605	\$ 3,868
Preferred stock, par value (in dollars per share)	\$ 0	\$ 0
Preferred stock, shares authorized (in shares)	350,000	350,000
Preferred stock, shares outstanding (in shares)	0	0
Common stock, par value (in dollars per share)	\$ 0	\$ 0
Common stock, shares authorized (in shares)	8,500,000	8,500,000
Common stock, shares issued (in shares)	3,124,053	3,124,053
Treasury stock, shares (in shares)	95,953	108,475

<b>Consolidated Statements of</b>	3 Mon	3 Months Ended 6 Months 1				
Income (Unaudited) - USD (\$) \$ in Thousands	Dec. 31, 2020	Dec. 31, 2019	Dec. 31, 2020	Dec. 31, 2019		
Interest and dividend income						
Loans, including fees	\$ 6,583	\$ 4,862	\$ 13,072	\$ 9,623		
Securities, taxable	344	480	716	990		
Securities, tax-exempt	427	400	847	799		
Federal bank and other restricted stocks	21	20	39	40		
Federal funds sold and other interest-bearing deposits	42	16	89	42		
Total interest and dividend income	7,417	5,778	14,763	11,494		
<u>Interest expense</u>						
<u>Deposits</u>	487	910	1,064	1,855		
Short-term borrowings	2	13	6	24		
Federal Home Loan Bank advances	70	59	141	138		
<u>Total interest expense</u>	559	982	1,211	2,017		
Net interest income	6,858	4,796	13,552	9,477		
Provision for loan losses	130	185	260	315		
Net interest income after provision for loan losses	6,728	4,611	13,292	9,162		
Noninterest income						
Gain on sale of mortgage loans	246	142	482	276		
Bank owned life insurance death benefit				324		
Bank owned life insurance income	65	66	131	134		
Securities gains, net	8	4	8	110		
<u>Other</u>	75	69	151	142		
Total noninterest income	1,153	1,025	2,294	2,494		
Noninterest expenses						
Salaries and employee benefits	2,760	2,207	5,451	4,380		
Occupancy and equipment	636	595	1,276	1,127		
Data processing expenses	176	141	362	526		
Debit card processing expenses	216	194	454	395		
<u>Professional and director fees</u>	249	133	486	390		
FDIC assessments	87	5	148	(2)		
<u>Franchise taxes</u>	108	95	217	190		
Marketing and advertising	116	93	250	274		
<u>Telephone and network communications</u>	79	70	163	144		
Amortization of intangible	7		14			
<u>Other</u>	397	402	809	816		
<u>Total noninterest expenses</u>	4,831	3,935	9,630	8,240		
<u>Income before income taxes</u>	3,050	1,701	5,956	3,416		
Income tax expense	543	261	1,048	473		
Net income	\$ 2,507	\$ 1,440	\$ 4,908	\$ 2,943		
Basic and diluted earnings per share (in dollars per share)	\$ 0.83	\$ 0.53	\$ 1.63	\$ 1.07		

Deposit Account [Member]				
Noninterest income				
Noninterest income (in scope of Topic 606)	\$ 314	\$ 360	\$ 621	\$ 733
Debit Card [Member]				
Noninterest income				
Noninterest income (in scope of Topic 606)	\$ 445	\$ 384	\$ 901	\$ 775

<b>Consolidated Statements of</b>	3 Mont	hs Ended	6 Months Ended			
Comprehensive Income (Unaudited) - USD (\$) \$ in Thousands	Dec. 31, 2020	Dec. 31, 2019	Dec. 31, 2020	Dec. 31, 2019		
Net income	\$ 2,507	\$ 1,440	\$ 4,908	\$ 2,943		
Other comprehensive income, net of tax:						
Unrealized gains (losses) arising during the period	278	(28)	348	790		
Reclassification adjustment for gains included in income	[1],[2](8)	(4)	(8)	(110)		
Net unrealized gains (losses)	270	(32)	340	680		
Income tax effect	(57)	6	(71)	(144)		
Other comprehensive income (loss)	213	(26)	269	536		
<u>Total comprehensive income</u>	\$ 2,720	\$ 1,414	\$ 5,177	\$ 3,479		

<sup>[1]</sup> Income tax expense

<sup>[2]</sup> Securities (gains) losses, net

Condensed Consolidated Statements of Changes in Shareholders' Equity (Unaudited) - USD (\$) \$ in Thousands	Common Stock [Member]	Retained Earnings [Member]	Treasury Stock [Member]	AOCI Attributable to Parent [Member]	Total
Balance at Jun. 30, 2019	\$ 14,656	\$ 36,487	\$ (1,543)	\$ 1,566	\$ 51,166
Net income		2,943			2,943
Other comprehensive income (loss)				536	536
Cash dividends declared		(739)			(739)
Shares associated with stock awards	41	( )	89		130
Balance at Dec. 31, 2019	14,697	38,691	(1,454)	2,102	54,036
Balance at Sep. 30, 2019	14,697	37,622	(1,454)	2,128	52,993
Net income		1,440			1,440
Other comprehensive income				(26)	(26)
(loss)		(271)			
Cash dividends declared	14.607	(371)	(1.454)	2 102	(371)
Balance at Dec. 31, 2019 Balance at Jun. 30, 2020	14,697 19,974	38,691 40,460	(1,454) (1,454)	2,102 4,260	54,036 63,240
Net income	19,974	4,908	(1,434)	4,200	4,908
Other comprehensive income		1,500			•
(loss)				269	269
Cash dividends declared		(876)			(876)
Shares associated with stock	37		130		167
<u>awards</u>					
Balance at Dec. 31, 2020	20,011	44,492	(1,324)	4,529	67,708
Balance at Sep. 30, 2020	20,011	42,424	(1,324)	4,316	65,427
Net income		2,507			2,507
Other comprehensive income (loss)				213	213
Cash dividends declared		(439)			(439)
Balance at Dec. 31, 2020	<b>A. B.</b> O. O. C.		<b>*</b> (4.25.1)	<b>*</b> 4 <b>* * * * * * * * * *</b>	\$
	\$ 20,011	\$ 44,492	\$ (1,324)	\$ 4,529	67,708

<b>Condensed Consolidated</b>	3 Mont	ths Ended	6 Months Ended		
Statements of Changes in Shareholders' Equity (Unaudited) (Parentheticals)	Dec. 31, 2020	Dec. 31, 2019	Dec. 31, 2020	Dec. 31, 2019	
- \$ / shares  Common stock, dividends, per share, declared (in dollars per	¢ 0 145	\$ 0.135	\$ 0.29	¢ 0.27	
share) Shares associated with stock awards (in shares)	\$ 0.143	\$ 0.133	12.522	\$ 0.27 11.813	

Condensed Consolidated Statements of Cash Flows	6 Mon	ths Ended
(Unaudited) - USD (\$) \$ in Thousands	Dec. 31, 20	20 Dec. 31, 2019
Cash flows from operating activities		
Net cash from operating activities	\$ 5,706	\$ 2,214
Cash flow from investing activities		
Purchases of securities, available-for-sale	(24,975)	(6,678)
Maturities, calls and principal pay downs of securities, available-for-sal		11,902
Sale of securities, available-for-sale	2,733	6,110
Principal pay downs of securities, held-to-maturity	205	206
Purchase of equity securities	(400)	
Net decrease in certificate of deposit in other financial institutions	3,036	989
Net increase in loans	(14,431)	(27,226)
Proceeds from BOLI death benefit		753
Premises and equipment purchases	(194)	(219)
Sale of other repossessed assets	17	
Net cash from investing activities	(13,705)	(14,163)
Cash flow from financing activities		
Net increase in deposit accounts	15,007	15,471
Net change in short-term borrowings	6,332	184
Proceeds from Federal Home Loan Bank advances	1,300	9,500
Repayments of Federal Home Loan Bank advances	(14,378)	(7,900)
<u>Dividends paid</u>	(876)	(739)
Net cash from financing activities	7,385	16,516
Increase (decrease) in cash or cash equivalents	(614)	4,567
Cash and cash equivalents, beginning of period	9,659	9,461
Cash and cash equivalents, end of period	9,045	14,028
<b>Supplemental disclosure of cash flow information:</b>		
<u>Interest</u>	1,230	2,022
Federal income taxes	1,055	350
Non-cash items:		
<u>Transfer of loans to other repossessed assets</u>	9	
<u>Issuance of treasury stock for stock awards</u>	167	89
Right of use assets obtained in exchange for lease liabilities		\$ 582

## Note 1 - Summary of Significant Accounting Policies

Notes to Financial Statements

Organization, Consolidation and Presentation of Financial Statements Disclosure and Significant Accounting Policies [Text Block]

# 6 Months Ended **Dec. 31, 2020**

### Note 1 – Summary of Significant Accounting Policies:

**Nature of Operations:** Consumers Bancorp, Inc. (the Corporation) is a bank holding company headquartered in Minerva, Ohio that provides, through its banking subsidiary, Consumers National Bank (the Bank), a broad array of products and services throughout its primary market area of Carroll, Columbiana, Jefferson, Stark, Summit, Wayne and contiguous counties in Ohio. The Bank's business involves attracting deposits from businesses and individual customers and using such deposits to originate commercial, mortgage and consumer loans in its primary market area.

Basis of Presentation: The consolidated financial statements for interim periods are unaudited and reflect all adjustments (consisting of only normal recurring adjustments), which, in the opinion of management, are necessary to present fairly the financial position and results of operations and cash flows for the periods presented. The unaudited financial statements are presented in accordance with the requirements of Form 10-Q and do not include all disclosures normally required by accounting principles generally accepted in the United States of America. The financial statements should be read in conjunction with the consolidated financial statements and notes thereto included in the Corporation's Form 10-K for the year ended June 30, 2020. The results of operations for the interim period disclosed herein are not necessarily indicative of the results that may be expected for a full year.

The consolidated financial statements include the accounts of the Corporation and the Bank. All significant inter-company transactions and accounts have been eliminated in consolidation.

**Segment Information:** The Corporation is a bank holding company engaged in the business of commercial and retail banking, which accounts for substantially all the revenues, operating income, and assets. Accordingly, all of the Corporation's operations are recorded in one segment, banking.

**Reclassifications:** Certain items in prior financial statements have been reclassified to conform to the current presentation. Any reclassifications had no impact on prior year net income or shareholders' equity.

Recently Issued Accounting Pronouncements Not Yet Effective: In June 2016, the Financial Accounting Standards Board (FASB) issued ASU 2016-13, Financial Instruments—Credit Losses (Topic 326): Measurement of Credit Losses on Financial Instruments. This ASU adds a new Topic 326 to the codification and removes the thresholds that companies apply to measure credit losses on financial instruments measured at amortized cost, such as loans, receivables, and held-to-maturity debt securities. Under current U.S. generally accepted accounting principles, companies generally recognize credit losses when it is probable that the loss has been incurred. The revised guidance will remove all current loss recognition thresholds and will require companies to recognize an allowance for credit losses for the difference between the amortized cost basis of a financial instrument and the amount of amortized cost that the corporation expects to collect over the instrument's contractual life. ASU 2016-13 also amends the credit loss measurement guidance for available-for-sale debt securities and beneficial interests in securitized financial assets. The guidance in ASU 2016-13 is effective for "public business entities," as defined in the guidance, that are SEC filers for fiscal years and for interim periods within those fiscal years beginning after December 15, 2019. Early adoption of the guidance is permitted for fiscal years beginning after December 15, 2018, including interim periods within those fiscal years. However, during July 2019, FASB unanimously voted for a proposal to delay this ASU to January 2023 for smaller reporting companies. On October 16, 2019, FASB approved a final ASU delaying the effective date. The new guidance is effective for annual and interim periods beginning after December 15,

2022 for certain entities, including smaller reporting companies. The Corporation is a smaller reporting company.

In March 2020, the FASB issued ASU 2020-04, "Facilitation of the Effects of Reference Rate Reform on Financial Reporting". The ASU is intended to provide relief for companies preparing for discontinuation of interest rates based on LIBOR, or other reference rates that may be discontinued, and provides optional expedients and exceptions for applying GAAP to contract modifications and hedging relationships, subject to meeting certain criteria. The ASU also provides for a one-time sale and/or transfer to AFS or trading to be made for HTM debt securities that both reference an eligible reference rate and were classified as HTM before January 1, 2020. ASU 2020-04 is effective March 12, 2020 through December 31, 2022. The guidance requires companies to apply the guidance prospectively to contract modifications and hedging relationships while the one-time election to sell and/or transfer debt securities classified as HTM may be made any time after March 12, 2020. The Corporation does not expect ASU 2020-04 to have a material impact on its financial statements and disclosures.

### Note 2 - Acquisition

Notes to Financial
Statements
Business Combination
Disclosure [Text Block]

## 6 Months Ended **Dec. 31, 2020**

#### Note 2 - Acquisition

On December 29, 2020, the Bank entered into a Branch Purchase and Assumption Agreement (P&A Agreement) with CFBank National Association (CFBank) to acquire two branches of CFBank in Columbiana County, Ohio. The P&A Agreement provides for the sale and transfer by CFBank to the Bank the land, buildings and other associated assets of CFBank's drive-up branch location in Wellsville, Ohio and CFBank's branch location in Calcutta, Ohio (the Branches); approximately \$100 million in deposits attributable to the Branches; \$15 million in aggregate principal amount of subordinated debt securities issued by unrelated financial institutions; all performing loans attributable to the Branches which are outstanding at closing (totaling approximately \$3.1 million in aggregate principal amount as of November 30, 2020); and up to \$13.5 million in aggregate principal amount of single family residential mortgage loans and home equity lines of credit to be identified by the parties prior to the closing principally from CFBank's Northeast Ohio loan portfolio. In addition, CFBank will provide the opportunity for the Corporation to purchase at par at least \$15 million in aggregate principal amount of participation interests in commercial and commercial real estate loans originated by and led in CFBank's portfolio. In exchange, Consumers will pay to CFBank the net book value of the land, building and associated assets of the Branches, a deposit premium equal to 1.75% of the average daily deposits of the Branches for the 30 days preceding the closing, and the par value of the subordinated debt securities and loans acquired by Consumers.

The closing of the transactions contemplated by the P&A Agreement is subject to regulatory approval and satisfaction of other customary closing conditions. The parties expect the closing of the transactions to occur early in the third calendar quarter of 2021.

On June 14, 2019, the Corporation entered into an Agreement and Plan of Merger with Peoples Bancorp of Mt. Pleasant, Inc. (Peoples) and its wholly owned subsidiary, The Peoples National Bank of Mount Pleasant (Peoples Bank). On January 1, 2020, Consumers completed the acquisition by merger of Peoples in a stock and cash transaction for an aggregate consideration of approximately \$10,405. In connection with the acquisition, the Corporation issued 269,920 shares of common stock and paid \$5,128 in cash to the former shareholders of Peoples. Immediately following the merger, Peoples Bank, was merged into the Corporation's banking subsidiary, Consumers National Bank.

On December 31, 2019, Peoples had approximately \$72,016 in total assets, \$55,273 in loans and \$60,826 in deposits at its three banking centers located in Mt. Pleasant, Adena, and Dillonvale, Ohio. The assets and liabilities of Peoples were recorded on the Corporation's Balance Sheet at their estimated fair values as of January 1, 2020, the acquisition date, and Peoples' results of operations are included in the Corporation's Consolidated Statements of Income beginning on that date.

The following table summarizes the estimated fair values of the assets acquired and liabilities assumed at the date of acquisition of Peoples. The core deposit intangible will be amortized over ten years on a straight-line basis. Goodwill will not be amortized, but instead will be evaluated for impairment.

Consideration Paid	\$	\$ 10,405
Net assets acquired:		
Cash and cash equivalents	\$ 833	
Certificates of deposit in other financial institutions	11,839	
Securities, available-for-sale	4,051	
Federal bank and other restricted stocks, at cost	154	
Loans, net	55,320	
Premises and equipment	818	
Core deposit intangible	270	
Accrued interest receivable and other assets	140	
Noninterest-bearing deposits	(11,979)	
Interest-bearing deposits	(48,872)	
Federal funds purchased	(2,348)	
Federal Home Loan Bank advances	(491)	
Other liabilities	 (166)	
Total net assets acquired		9,569
Goodwill	9	\$ 836

The acquired assets and liabilities were measured at estimated fair values. Management made certain estimates and exercised judgement in accounting for the acquisition. The fair value of loans was estimated using discounted contractual cash flows. The book balance of the loans at the time of the acquisition was \$55,273 before considering Peoples' allowance for loan losses, which was not carried over. The fair value disclosed above reflects a credit-related adjustment of \$(890) and an adjustment for other factors of \$937. Loans evidencing credit deterioration since origination, purchased credit impaired loans, included in loans receivable, were immaterial. Acquisition costs of \$827 pre-tax, or \$680 after-tax, were recorded during the twelve-month period ended June 30, 2020

### **Note 3 - Securities**

### 6 Months Ended Dec. 31, 2020

## Notes to Financial Statements

Investments in Debt and
Marketable Equity Securities
(and Certain Trading Assets)
Disclosure [Text Block]

### Note 3 – Securities

Available –for-Sale	Amortized Cost				ed Gross Unrealized Gains		Uni	Gross realized Josses	Fair Value
December 31, 2020									
U.S. Treasury	\$	499	\$	2	\$	— \$	501		
Obligations of U.S. government-sponsored entities and agencies		7,793		316		_	8,109		
Obligations of state and political subdivisions		63,359		3,675		_	67,034		
U.S. Government-sponsored mortgage-backed securities-residential		56,757		1,496		(5)	58,248		
U.S. Government-sponsored mortgage-backed securities—commercial		7,004		75		(1)	7,078		
U.S. Government-sponsored collateralized mortgage obligations— residential		4,520		175		_	4,695		
Total available-for-sale securities	\$	139,932	\$	5,739	\$	(6) \$	145,665		

Held-to-Maturity	Amortized Cost								Gross Unrecognized Gains		Unre	Gross Unrecognized Losses		Fair Value	
December 31, 2020															
Obligations of state and political subdivisions	\$	3,336	\$	269	\$	_	\$	3,605							
Total held-to-maturity securities	\$	3,336	\$	269	\$		\$	3,605							

Available-for-Sale	Amortized Cost		Unrealized		Gross Unrealized Losses		Fair Value	
June 30, 2020								
U.S. Treasury	\$	1,248	\$	8	\$	\$	1,256	
Obligations of U.S. government-sponsored entities and agencies		10,133		399		_	10,532	
Obligations of state and political subdivisions		60,343		3,149		_	63,492	
U.S. government-sponsored mortgage-backed securities - residential		48,645		1,515		(4)	50,156	
U.S. government-sponsored mortgage-backed securities - commercial		8,444		55		(2)	8,497	
U.S. government-sponsored collateralized mortgage obligations - residential		9,712		285		(12)	9,985	
Total available-for-sale securities	\$	138,525	\$	5,411	\$	(18) \$	143,918	

Held-to-Maturity	Amortized Cost		Inrecognized		Gross Unrecognized Losses		Fair ⁄alue
June 30, 2020							
Obligations of state and political subdivisions	\$	3,541	\$	327	\$	_	\$ 3,868
Total held-to-maturity securities	\$	3,541	\$	327	\$		\$ 3,868

Proceeds from the sale and call of available-for-sale securities were as follows:

	Three Months Ended			Six Mont		
	Decem	ber 3	31,	December 31,		
	 2020 201		2019	2020	2019	
Proceeds from sales and calls	\$ 2,733	\$	1,650	2,733	6,110	
Gross realized gains	31		4	31	110	
Gross realized losses	23		_	23	_	

The income tax provision related to the net realized gain amounted to \$2 for the three- and six-month periods ended December 31, 2020. The income tax provision related to the net realized gains amounted to \$1 and \$22 for the three- and six-month periods ended December 31, 2019, respectively.

The amortized cost and fair values of debt securities at December 31, 2020, by expected maturity, are shown below. Expected maturities will differ from contractual maturities because borrowers may have the right to call or prepay obligations with or without call or prepayment penalties. Securities not due at a single maturity date, primarily mortgage-backed securities, are shown separately.

Amortized Estimated Fair

Available-for-Sale	Cost	Value
Due in one year or less	\$ 4,930	\$ 5,006
Due after one year through five years	14,015	14,545
Due after five years through ten years	13,700	14,327
Due after ten years	 39,006	 41,766
Total	71,651	75,644
U.S. Government-sponsored mortgage-backed and related securities	69,281	70,021
Total available-for-sale securities	\$ 139,932	\$ 145,665
Held-to-Maturity		
Due after one year through five years	\$ 334	\$ 356
Due after five years through ten years	667	735
Due after ten years	2,335	2,514
Total held-to-maturity securities	\$ 3,336	\$ 3,605

The following table summarizes the securities with unrealized losses at December 31, 2020 and June 30, 2020, aggregated by investment category and length of time that individual securities have been in a continuous unrealized loss position:

	Less than	12 Months	12 Month	s or more	Total			
Available-for-sale	Fair Value	Unrealized Loss	Fair Value	Unrealized Loss	Fair Value	Unrealized Loss		
December 31, 2020								
U.S. Government-sponsored mortgage-backed securities – residential	1,198	(5)	_	_	1,198	(5)		
U.S. Government-sponsored mortgage-backed securities- commercial	1,422	(1)	_	_	1,422	(1)		
Total temporarily impaired	\$ 2,620	\$ (6)	\$	\$	\$ 2,620	\$ (6)		

	Less th	Less than 12 Months			12 Month	s or more	Total			
Available-for-sale	Fair Value		Unrealized Loss		Fair Value	Unrealized Loss	Fair Value	Unrealized Loss		
June 30, 2020										
U.S. Government-sponsored mortgage-backed securities – residential		_	_		625	(4)	625	(4)		
U.S. Government-sponsored mortgage-backed securities – commercial	1,8	306	(2)		_	_	1,806	(2)		
U.S. Government-sponsored collateralized mortgage obligations - residential	1,7	00	(12)		_	_	1,700	(12)		
Total temporarily impaired	\$ 3,5	06	\$ (14)	\$	625	\$ (4)	\$ 4,131	\$ (18)		

Management evaluates securities for other-than-temporary impairment (OTTI) on a quarterly basis, and more frequently when economic or market conditions warrant such an evaluation. The securities portfolio is evaluated for OTTI by segregating the portfolio into two general segments and applying the appropriate OTTI model. Investment securities are generally evaluated for OTTI under FASB ASC Topic 320, Accounting for Certain Investments in Debt and Equity Securities.

In determining OTTI under the ASC Topic 320 model, management considers many factors, including: (1) the length of time and the extent to which the fair value has been less than cost, (2) the financial condition and near-term prospects of the issuer, (3) whether the market decline was affected by macroeconomic conditions, and (4) whether the entity has the intent to sell the debt security or more likely than not will be required to sell the debt security before its anticipated recovery. The assessment of whether an other-than-temporary decline exists involves a high degree of subjectivity and judgment and is based on the information available to management at a point in time.

The unrealized losses within the securities portfolio as of December 31, 2020 have not been recognized into income because the decline in fair value is not attributed to credit quality and management does not intend to sell, and it is not likely that management will be required to sell, the securities prior to their anticipated recovery. The decline in fair value within the securities portfolio is largely due to increases in mortgage backed and collateralized mortgage obligations prepayment speeds impacting the yield on bonds that were purchased at a premium. The mortgage-backed securities and collateralized mortgage obligations were primarily issued by Fannie Mae, Freddie Mac and Ginnie Mae, institutions which the government has affirmed its commitment to support. The Corporation does not own any private label mortgage-backed securities.

### 6 Months Ended Dec. 31, 2020

### Note 4 - Loans

Notes to Financial Statements

Loans, Notes, Trade and Other Note 4 - Loans

Receivables Disclosure [Text Block]

Major classifications of loans were as follows:

	Dec	ember 31, 2020	 June 30, 2020
Commercial	\$	158,911	\$ 158,667
Commercial real estate:			
Construction		3,802	16,235
Other		250,529	229,029
1 – 4 Family residential real estate:			
Owner occupied		93,726	90,494
Non-owner occupied		20,210	19,370
Construction		6,026	9,344
Consumer		24,230	21,334
Subtotal		557,434	544,473
Net Deferred loan fees and costs		(177)	(1,612)
Allowance for loan losses		(5,912)	(5,678)
Net Loans	\$	551,345	\$ 537,183

The commercial loan category in the above table includes PPP loans of \$52,539 as of December 31, 2020 and \$66,606 as of June 30, 2020 and a mortgage loan warehous line of credit to another financial institution with an outstanding balance of \$47,268 as of December 31, 2020 and \$32,869 as of June 30, 2020. The outstanding balance of the warehouse line of credit can fluctuate significantly based on the other financial institution's funding needs.

The following table presents the activity in the allowance for loan losses by portfolio segment for the three months ended December 31, 2020:

	Com	mercial	C	ommercial Real Estate	-4 Family Residential Real Estate	С	onsumer	_	Total
Allowance for loan losses:									
Beginning balance	\$	940	\$	3,694	\$ 997	\$	136	\$	5,767
Provision for loan losses		(82)		122	31		59		130
Loans charged-off		_		_	_		(12)		(12)
Recoveries		_		1	_		26		27
Total ending allowance balance	\$	858	\$	3,817	\$ 1,028	\$	209	\$	5,912

The following table presents the activity in the allowance for loan losses by portfolio segment for the six months ended December 31, 2020:

	Com	mercial_	C	ommercial Real Estate	-4 Family Residential Real Estate	_C	onsumer	 Total
Allowance for loan losses:								
Beginning balance	\$	947	\$	3,623	\$ 989	\$	119	\$ 5,678
Provision for loan losses		(67)		192	39		96	260
Loans charged-off		(22)		_	_		(56)	(78)
Recoveries				2	_		50	52
Total ending allowance balance	\$	858	\$	3,817	\$ 1,028	\$	209	\$ 5,912

The following table presents the activity in the allowance for loan losses by portfolio segment for the three months ended December 31, 2019:

	Com	mercial	Co	ommercial Real Estate	-4 Family Residential Real Estate	C	onsumer	_	Total
Allowance for loan losses:									
Beginning balance	\$	649	\$	2,645	\$ 550	\$	65	\$	3,909
Provision for loan losses		117		6	64		(2)		185
Loans charged-off		_		_	_		(7)		(7)
Recoveries		_		1	1		6		8
Total ending allowance balance	\$	766	\$	2,652	\$ 615	\$	62	\$	4,095

The following table presents the activity in the allowance for loan losses by portfolio segment for the six months ended December 31, 2019:

	Con	nmercial	C	ommercial Real Estate	-4 Family Residential Real Estate	_(	Consumer	 Total
Allowance for loan losses:								
Beginning balance	\$	660	\$	2,575	\$ 494	\$	59	\$ 3,788
Provision for loan losses		106		75	120		14	315
Loans charged-off		_		_	_		(23)	(23)

Recoveries	 	2	1	 12	15
Total ending allowance balance	\$ 766	\$ 2,652	\$ 615	\$ 62	\$ 4,095

The following table presents the balance in the allowance for loan losses and the recorded investment in loans by portfolio segment and based on impairment method as of December 31, 2020. Included in the recorded investment in loans is \$1,607 of accrued interest receivable.

	_Cc	ommercial	С	ommercial Real Estate	-4 Family Residential Real Estate	 Consumer	_	Total
Allowance for loan losses:								
Ending allowance balance attributable to loans:								
Individually evaluated for impairment	\$	3	\$	_	\$ 6	\$ _	\$	9
Acquired loans collectively evaluated for impairment		_		101	84	_		185
Originated loans collectively evaluated for impairment		855		3,716	938	 209		5,718
Total ending allowance balance	\$	858	\$	3,817	\$ 1,028	\$ 209	\$	5,912
						 		_
Recorded investment in loans:								
Loans individually evaluated for impairment	\$	148	\$	1,151	\$ 1,044	\$ _	\$	2,343
Acquired loans collectively evaluated for impairment		661		7,824	23,220	8,935		40,640
Originated loans collectively evaluated for impairment		158,331		245,489	96,733	15,328		515,881
Total ending loans balance	\$	159,140	\$	254,464	\$ 120,997	\$ 24,263	\$	558,864

The following table presents the balance in the allowance for loan losses and the recorded investment in loans by portfolio segment and based on impairment method as of June 30, 2020. Included in the recorded investment in loans is \$1,936 of accrued interest receivable.

Allowance for loan losses:	Со	ommercial	C	ommercial Real Estate	-4 Family Residential Real Estate	 Consumer	_	Total
Ending allowance balance attributable to loans:								
Individually evaluated for impairment	\$	28	\$	6	\$ _	\$ _	\$	34
Acquired loans collectively evaluated for impairment		_		103	94	_		197
Originated loans collectively evaluated for impairment		919		3,514	895	119		5,447
Total ending allowance balance	\$	947	\$	3,623	\$ 989	\$ 119	\$	5,678
Recorded investment in loans:								
Loans individually evaluated for impairment	\$	179	\$	1,045	\$ 699	\$ _	\$	1,923
Acquired loans collectively evaluated for impairment		1,095		8,072	27,252	12,550		48,969
Originated loans collectively evaluated for impairment		156,054		236,840	92,168	8,843		493,905
Total ending loans balance	\$	157,328	\$	245,957	\$ 120,119	\$ 21,393	\$	544,797

The following table presents information related to unpaid principal balance, recorded investment and interest income associated with loans individually evaluated for impairment by class of loans as of December 31, 2020 and for the six months ended December 31, 2020:

		As	of De	cember 31, 2	2020			Six Months ended December 31, 2020					
	Ţ	Jnpaid				lowance for		Average		Interest		Cash Basis	
	P	rincipal	I	Recorded	L	oan Losses		Recorded		Income		Interest	
	E	Balance	Iı	vestment		Allocated		Investment	_]	Recognized	_!	Recognized	
With no related allowance recorded:													
Commercial real estate:													
Other	\$	1,286	\$	1,151	\$	_	\$	867	\$	4	\$	4	
1-4 Family residential real estate:													
Owner occupied		947		797		_		629		11		11	
Non-owner occupied		274		217		_		225		_		_	
With an allowance recorded:													
Commercial		147		148		3		160		4		4	
Commercial real estate:													
Other		_		_		_		205		6		6	
1-4 Family residential real estate:													
Owner occupied		30		30		6	_	5	_		_		
Total	\$	2,684	\$	2,343	\$	9	\$	2,091	\$	25	\$	25	

The following table presents information related to average recorded investment and interest income associated with loans individually evaluated for impairment by class of loans for the three months ended December 31, 2020:

	Ave. Reco Inves	U	 rest ome gnized	Cash Basis Interest Recognized	
With no related allowance recorded:					
Commercial real estate:					
Other	\$	907	\$ 3	\$	3
1-4 Family residential real estate:					
Owner occupied		720	5		5
Non-owner occupied		220	_		_
With an allowance recorded:					
Commercial		150	2		2
Commercial real estate:					
Other		204	3		3
1-4 Family residential real estate:					

Owner occupied	 10	<u> </u>	
Total	\$ 2,211	\$ 13	\$ 13

The following table presents information related to unpaid principal balance, recorded investment and interest income associated with loans individually evaluated for impairment by class of loans as of June 30, 2020 and for the six months ended December 31, 2019:

		As of June 30, 2020							Six Months ended December 31, 2019				
		Unpaid Allowance for Principal Recorded Loan Losses			Average Recorded		Interest Income		ash Basis Interest				
	I	Balance	I	nvestment		Allocated	I	nvestment	Recognized		Re	ecognized	
With no related allowance recorded:													
Commercial real estate:													
Other	\$	922	\$	836	\$	_	\$	303	\$	87	\$	87	
1-4 Family residential real estate:													
Owner occupied		604		463		_		25		7		7	
Non-owner occupied		284		236		_		254		_		_	
With an allowance recorded:													
Commercial		176		179		28		167		5		5	
Commercial real estate:													
Other		209		209		6		219		6		6	
Total	\$	2,195	\$	1,923	\$	34	\$	968	\$	105	\$	105	

The following table presents information related to average recorded investment and interest income associated with loans individually evaluated for impairment by class of loans for the three months ended December 31, 2019:

	Average Recorded Investment	Interest Income Recognized	Cash Basis Interest Recognized	
With no related allowance recorded:	 	_		
Commercial real estate:				
Other	\$ 244	\$ 1	\$	1
1-4 Family residential real estate:				
Owner occupied	10	_		_
Non-owner occupied	250	_		_
With an allowance recorded:				
Commercial	165	2		2
Commercial real estate:				
Other	218	3		3
Total	\$ 887	\$ 6	\$	6

The following table presents the recorded investment in non-accrual and loans past due over 90 days still on accrual by class of loans as of December 31, 2020 and June 30, 2020:

	Decembe	r 31, 2020	June 30	0, 2020
	·	Loans Past Due		Loans Past Due
		Over 90 Days		Over 90 Days
		Still		Still
	Non-accrual	Accruing	Non-accrual	Accruing
Commercial	\$ —	\$ —	\$ 21	\$ —
Commercial real estate:				
Other	902	_	785	_
1 – 4 Family residential:				
Owner occupied	821	_	143	29
Non-owner occupied	217	_	236	_
Consumer				12
Total	\$ 1,940	<u> </u>	\$ 1,185	\$ 41

Non-accrual loans and loans past due 90 days still on accrual include both smaller balance homogeneous loans that are collectively evaluated for impairment an individually classified impaired loans.

The following table presents the aging of the recorded investment in past due loans as of December 31, 2020 by class of loans:

		Days Past Due				
	30 - 59	60 - 89	90 Days or	Total	Loans Not	
	Days	Days	Greater	Past Due	Past Due	Total
Commercial	\$ —	\$ —	\$ —	\$ —	\$ 159,140	\$ 159,140
Commercial real estate:						
Construction	_	_	_	_	3,791	3,791
Other	_	_	629	629	250,044	250,673
1-4 Family residential:						
Owner occupied	450	4	269	723	93,926	94,649
Non-owner occupied	_	_	_	_	20,223	20,223
Construction	4	_	_	4	6,121	6,125
Consumer	261	49	_	310	23,953	24,263
Total	\$ 715	\$ 53	\$ 898	\$ 1,666	\$ 557,198	\$ 558,864

The above table of past due loans includes the recorded investment in non-accrual loans of \$898 in the 90 days or greater category and \$1,042 in the loans not past du category.

The following table presents the aging of the recorded investment in past due loans as of June 30, 2020 by class of loans:

	Days Past Du	e		
30 - 59	60 - 89	90 Days or	Total	Loans Not

	Day	/S	Days		Greater	Past Due		Past Due	Total
Commercial	\$		\$ -		\$ 21	\$ 21	\$	157,307	\$ 157,328
Commercial real estate:									
Construction		_	-	_	_	_	-	16,241	16,241
Other		_		2	628	630	)	229,086	229,716
1-4 Family residential:									
Owner occupied		_	-		172	172		91,102	91,274
Non-owner occupied		_	-	_	_	_	-	19,410	19,410
Construction		_	-	_	_	_	-	9,435	9,435
Consumer		127	4	49	12	188	;	21,205	21,393
Total	\$	127	\$ :	51	\$ 833	\$ 1,011	\$	543,786	\$ 544,797

The above table of past due loans includes the recorded investment in non-accrual loans of \$2 in the 60-89 days, \$792 in the 90 days or greater category and \$391 in th loans not past due category.

#### Troubled Debt Restructurings (TDR):

The Corporation has certain loans that have been modified in order to maximize collection of loan balances that are classified as TDRs. A modified loan is usually classified as a TDR if, for economic reasons, management grants a concession to the original terms and conditions of the loan to a borrower who is experiencing financial difficultie that it would not have otherwise considered. In response to COVID-19, on March 22, 2020 the Corporation adopted a loan modification program to assist borrower impacted by the virus. The program is available to most borrowers whose loan was not past due on March 22, 2020, the date this loan modification program was adopted. The program offers principal and interest payment deferrals for up to 90 days or interest only payments for up to 90 days. Borrowers are eligible for an additional 90 day of payment deferrals if situations warrant a need for an extension. Interest will be deferred but will continue to accrue during the deferment period and the maturity date or amortizing loans will be extended by the number of months the payment was deferred. Consistent with issued regulatory guidance, modifications made under this program in response to COVID-19 will not be classified as TDRs. As of December 31, 2020, 18 borrowers with an aggregate outstanding balance of \$667 are in payment deferrastatus under this loan modification program.

As of December 31, 2020 and June 30, 2020, the Corporation had \$934 and \$974, respectively, of loans classified as TDRs which are included in impaired loans above As of December 31, 2020 and June 30, 2020, the Corporation had not committed to lend any additional funds to customers with outstanding loans that were classified a troubled debt restructurings. As of December 31, 2020 and June 30, 2020, the Corporation had \$3 and \$12, respectively, of specific reserve allocated to these loans.

During the three- and six-month periods ended December 31, 2020 and 2019, there were no loan modifications completed that were classified as troubled deb restructurings. There were no charge offs from troubled debt restructurings that were completed during the three- and six-month periods ended December 31, 2020 and 2019.

There were no loans classified as troubled debt restructurings for which there was a payment default within 12 months following the modification during the three- and six-month periods ended December 31, 2020 and 2019. A loan is considered in payment default once it is 90 days contractually past due under the modified terms.

#### Credit Quality Indicators:

The Corporation categorizes loans into risk categories based on relevant information about the ability of borrowers to service their debt such as: current financial information, historical payment experience, credit documentation, public information, current economic trends and other relevant information. The Corporation analyze loans individually by classifying the loans as to credit risk. This analysis includes loans with a total outstanding loan relationship greater than \$100 and non-homogeneou loans, such as commercial and commercial real estate loans. Management monitors the loans on an ongoing basis for any changes in the borrower's ability to service their debt and affirms the risk ratings for the loans and leases in their respective portfolio on an annual basis. The Corporation uses the following definitions for risk ratings:

Special Mention. Loans classified as special mention have a potential weakness that deserves management's close attention. If left uncorrected, these potential weaknesses may result in deterioration of the repayment prospects for the loan or of the institution's credit position at some future date.

**Substandard.** Loans classified as substandard are inadequately protected by the current net worth and paying capacity of the obligor or of the collateral pledged, in any. Loans so classified have a well-defined weakness or weaknesses that jeopardize the liquidation of the debt. They are characterized by the distinct possibility that the institution will sustain some loss if the deficiencies are not corrected.

**Doubtful.** Loans classified as doubtful have all the weaknesses inherent in those classified as substandard, with the added characteristic that the weaknesses mak collection or liquidation in full, on the basis of currently existing facts, conditions, and values, highly questionable and improbable.

Loans not meeting the criteria above that are analyzed individually as part of the above described process are considered pass rated loans. Loans listed as not rated are either less than \$100 or are included in groups of homogeneous loans. Generally, 1-4 Family Residential and Consumer loans are not risk rated, except when collateral is used for a business purpose. These loans are evaluated based on delinquency status, which are disclosed in the previous table within this footnote. Based on the most recer analysis performed, the recorded investment by risk category of loans by class of loans was as follows:

			S	pecial					Not
		Pass	M	lention	Substanda	ard	Do	ubtful	Rated
Commercial	\$	158,054	\$	453	\$ 3	359	\$		\$ 274
Commercial real estate:									
Construction		3,791		_		—		_	_
Other		241,702		1,933	5,0	)75		902	1,061
1-4 Family residential real estate:									
Owner occupied		1,988		_		18		603	92,040
Non-owner occupied		19,265		173	2	210		217	358
Construction		1,306		_		_		_	4,819
Consumer		687		_		_		_	23,576
Total	\$	426,793	\$	2,559	\$ 5,6	662	\$	1,722	\$ 122,128

As of June 30, 2020, and based on the most recent analysis performed, the recorded investment by risk category of loans by class of loans is as follows:

	As of June 30, 2020										
	 Special										
	Pass	Mention		Substandard	Do	ubtful		Rated			
Commercial	\$ 152,911	\$ 14	13	\$ 3,979	\$	21	\$	274			
Commercial real estate:											
Construction	16,241	-	_	_		_		_			
Other	220,311	1,46	9	5,378		785		1,773			
1-4 Family residential real estate:											
Owner occupied	2,419	-	_	334		_		88,521			

Non-owner occupied	18,435	186	223	236	330
Construction	3,234	_	_	_	6,201
Consumer	153	_	_	_	21,240
Total	\$ 413,704	\$ 1,798	\$ 9,914	\$ 1,042	\$ 118,339

### Note 5 - Fair Value

6 Months Ended Dec. 31, 2020

Notes to Financial
Statements
Fair Value Disclosures [Text Block]

### Note 5 - Fair Value

Fair value is the exchange price that would be received for an asset or paid to transfer a liability (an exit price) in the principal or most advantageous market for the asset or liability in an orderly transaction between market participants on the measurement date. There are three levels of inputs that may be used to measure fair values:

Level 1: Quoted prices (unadjusted) for identical assets or liabilities in active markets that the entity has the ability to access as of the measurement date.

Level 2: Significant other observable inputs other than Level 1 prices such as quoted prices for similar assets or liabilities; quoted prices in markets that are not active; or other inputs that are observable or can be corroborated by observable market data.

Level 3: Significant unobservable inputs that reflect a company's own assumptions about the assumptions that market participants would use in pricing an asset or liability.

Financial assets and financial liabilities measured at fair value on a recurring basis include the following:

Securities available-for-sale: When available, the fair values of available-for-sale securities are determined by obtaining quoted prices on nationally recognized securities exchanges (Level 1 inputs). For securities where quoted market prices are not available, fair values are calculated based on market prices of similar securities (Level 2 inputs). For securities where quoted prices or market prices of similar securities are not available, fair values are calculated using discounted cash flows or other unobservable inputs (Level 3 inputs).

Assets and liabilities measured at fair value on a recurring basis are summarized below, segregated by the level of the valuation inputs within the fair value hierarchy utilized to measure fair value:

F : Y/1 36

						e Measuremonder 31, 202		sat
	Balance at December 31, 2020			Level 1		Level 2		Level 3
Assets:								
U.S. Treasury	\$	501	\$	_	\$	501	\$	_
Obligations of U.S. government-sponsored entities and agencies		8,109		_		8,109		_
Obligations of states and political subdivisions		67,034		_		67,034		_
U.S. Government-sponsored mortgage-backed securities – residential		58,248		_		58,248		_
U.S. Government-sponsored mortgage-backed securities – commercial		7,078		_		7,078		_
U.S. Government-sponsored collateralized mortgage obligations - residential		4,695		_		4,695		_
Equity securities		400		_		400		_
				Fair V	alue	e Measureme	ents	s at

			June 30, 2020								
	Balance at June 30, 2020		Level 1		Level 2			Level 3			
Assets:											
Securities available-for-sale:											
Obligations of U.S. Treasury	\$	1,256	\$	_	\$	1,256	\$	_			
Obligations of government-sponsored entities		10,532		_		10,532		_			
Obligations of states and political subdivisions		63,492		_		63,492		_			
U.S. Government-sponsored mortgage-backed securities - residential		50,156		_		50,156		_			
U.S. Government-sponsored mortgage-backed securities – commercial		8,497		_		8,497		_			
U.S. Government-sponsored collateralized mortgage obligations		9,985		_		9,985		_			

There were no transfers between Level 1 and Level 2 during the three-month and six-month periods ended December 31, 2020 or 2019.

Certain assets and liabilities are measured at fair value on a non-recurring basis; that is, the instruments are not measured at fair value on an ongoing basis but are subject to fair value adjustments in certain circumstances. Assets and liabilities measured at fair value on a non-recurring basis include the following:

Impaired Loans: At the time a loan is considered impaired, it is valued at the lower of cost or fair value. Impaired loans carried at fair value generally receive specific allocations of the allowance for loan losses or are charged down to their fair value. For collateral dependent loans, fair value is commonly based on recent real estate appraisals. These appraisals may utilize a single valuation approach or a combination of approaches including comparable sales and the income approach. Adjustments are routinely made in the appraisal process by the appraisers to adjust for differences between the comparable sales and income data available. Such adjustments are usually significant and typically result in a Level 3 classification of the inputs for determining fair value. There were no impaired loans measured at fair value on a non-recurring basis at December 31, 2020 or June 30, 2020 and there was no impact to the provision for loan losses for the three- or six-month periods ended December 31, 2020 or 2019.

Other Real Estate and Repossessed Assets Owned: Assets acquired through or instead of loan foreclosure are initially recorded at fair value less costs to sell when acquired, establishing a new cost basis. These assets are subsequently accounted for at lower of cost or fair value less estimated costs to sell. Real estate owned properties and other repossessed assets, which are primarily vehicles, are evaluated on a quarterly basis for additional impairment and adjusted accordingly. There was no other real estate owned or other repossessed assets being carried at fair value as of December 31, 2020 or June 30, 2020.

The following table shows the estimated fair values of financial instruments that are reported at amortized cost in the Corporation's consolidated balance sheets, segregated by the level of the valuation inputs within the fair value hierarchy utilized to measure fair value:

	Decemb	er :	31, 2020		June	2020	
	Carrying Amount		Estimated Fair Value	Carrying Amount			Estimated Fair Value
Financial Assets:							
Level 1 inputs:							
Cash and cash equivalents	\$ 9,045	\$	9,045	\$	9,659	\$	9,659
Level 2 inputs:							
Certificates of deposit in other financial institutions	8,599		8,819		11,635		11,889
Loans held for sale	4,770		4,907		3,507		3,566
Accrued interest receivable	2,327		2,327		2,646		2,646
Level 3 inputs:							
Securities held-to-maturity	3,336		3,605		3,541		3,868
Loans, net	551,345		565,397		537,183		548,247
Financial Liabilities:							
Level 2 inputs:							
Demand and savings deposits	540,811		540,811		517,973		517,973
Time deposits	107,551		108,158		115,382		116,238
Short-term borrowings	13,275		13,275		6,943		6,943
Federal Home Loan Bank advances	18,083		18,423		31,161		31,571
Accrued interest payable	88		88		107		107

### **Note 6 - Earnings Per Share**

6 Months Ended Dec. 31, 2020

Notes to Financial
Statements
Earnings Per Share [Text
Block]

### Note 6 - Earnings Per Share

Basic earnings per share is the amount of earnings available to each share of common stock outstanding during the reporting period and is equal to net income divided by the weighted average number of shares outstanding during the period. Diluted earnings per share is the amount of earnings available to each share of common stock outstanding during the reporting period adjusted to include the effect of potentially dilutive common shares that may be issued upon the vesting of restricted stock awards. There were 569 shares of restricted stock that were anti-dilutive for the six-month period ended December 31, 2020. There were 1,645 and 2,204 shares of restricted stock that were anti-dilutive for the three- and six-month periods ended December 30, 2019. The following table details the calculation of basic and diluted earnings per share:

	For the Three Months Ended December 31,									Months Ende mber 31,			
	_	2020	2019			2020				2019	)		
Basic:													
Net income available to common shareholders	9	5 2,	507	\$	1,	440	\$	4,9	800	\$	2	2,943	
Weighted average common shares outstanding		3,017,	268		2,737,	800	3	,015,7	741		2,737	7,755	
Basic income per share	5	6 (	0.83	\$	(	).53	\$	1	.63	\$		1.07	
Diluted:													
Net income available to common shareholders	\$	2,507	\$		1,440	\$	4,9	806	\$	2	,943		
Weighted average common shares outstanding	3	,017,268		2,73	7,800		3,015,	741	2	,737	,755		
Dilutive effect of restricted stock		21			_			—			_		
Total common shares and dilutive potential common shares	3.	,017,289		2,73	7,800		3,015,	741	2	,737	,755		
Dilutive income per share	\$	0.83	\$		0.53	\$	1	.63	\$		1.07		

## Note 7 - Accumulated Other Comprehensive Income (Loss)

6 Months Ended **Dec. 31, 2020** 

Notes to Financial Statements

<u>Comprehensive Income (Loss)</u> <u>Note 7 – Accumulated Other Comprehensive Income (Loss)</u>

Note [Text Block]

The components of other comprehensive income related to unrealized gains and losses on available-for-sale securities for the three and six-month periods ended December 31, 2020 and 2019, were as follows:

	Pretax		Tax Effect		After-tax	Affected Line Item in Consolidated Statements of Income
Balance as of September 30, 2020	\$	5,463	\$	(1,147)	\$ 4,316	
Unrealized holding loss on available-for-sale securities arising during the period		278		(59)	219	
Amounts reclassified from accumulated other comprehensive income		(8)		2	 (6)	(a)(b)
Net current period other comprehensive loss		270		(57)	213	
Balance as of December 31, 2020	\$	5,733	\$	(1,204)	\$ 4,529	
					,	
Balance as of September 30, 2019	\$	2,694	\$	(566)	\$ 2,128	
Unrealized holding gain on available-for-sale securities arising during the period		(28)		5	(23)	
Amounts reclassified from accumulated other comprehensive income		(4)		1	(3)	(a)(b)
Net current period other comprehensive income		(32)		6	(26)	
Balance after reclassification as of December 31, 2019	\$	2,662	\$	(560)	\$ 2,102	

<sup>(</sup>a) Securities (gains) losses, net

<sup>(</sup>b) Income tax expense

Pretax		Tax Effect		After-tax	Affected Line Item in Consolidated Statements of Income
\$ 5,393	\$	(1,133)	\$	4,260	
348		(73)		275	
(8)		2		(6)	(a)(b)
340		(71)		269	
\$ 5,733	\$	(1,204)	\$	4,529	
\$ 1,982	\$	(416)	\$	1,566	
790		(166)		624	
(110)		22		(88)	(a)(b)
680		(144)		536	
\$ 2,662	\$	(560)	\$	2,102	
\$	\$ 5,393 348 (8) 340 \$ 5,733 \$ 1,982 790 (110) 680	\$ 5,393 \$ 348	\$ 5,393 \$ (1,133) 348 (73) (8) 2 340 (71) \$ 5,733 \$ (1,204) \$ 1,982 \$ (416) 790 (166) (110) 22 680 (144)	\$ 5,393 \$ (1,133) \$ 348 (73) \$ (8) 2 \$ (71) \$ \$ 5,733 \$ (1,204) \$ \$ \$ \$ 790 (166) \$ (110) 22 \$ 680 (144)	\$ 5,393 \$ (1,133) \$ 4,260 348 (73) 275 (8) 2 (6) 340 (71) 269 \$ 5,733 \$ (1,204) \$ 4,529 \$ 1,982 \$ (416) \$ 1,566 790 (166) 624 (110) 22 (88) 680 (144) 536

<sup>(</sup>a) Securities (gains) losses, net

<sup>(</sup>b) Income tax expense

### Note 8 - COVID-19

6 Months Ended **Dec. 31, 2020** 

COVID 19 [Member]

Notes to Financial
Statements
Unusual or Infrequent Items,
or Both, Disclosure [Text
Block]

### Note 8 – COVID-19

In December 2019, a novel strain of coronavirus surfaced in Wuhan, China, and has spread around the world, resulting in business and social disruption. The coronavirus was declared a Pandemic by the World Health Organization on March 11, 2020. The operations and business results of the Corporation could be materially adversely affected. The extent to which the coronavirus may impact business activity or investment results will depend on future developments, which are highly uncertain and cannot be predicted, including new information which may emerge concerning the severity of the coronavirus and the actions required to contain the coronavirus or treat its impact, among others. As a result of the economic shutdown engineered to slow down the spread of COVID-19, the ability of our customers to make payments on loans could be adversely impacted, resulting in elevated loan losses and an increase in the Corporation's allowance for loan losses. Additionally, it is reasonably possible future evaluations of the carrying amount of goodwill could result in a conclusion that goodwill is impaired.

# **Significant Accounting Policies (Policies)**

# Accounting Policies [Abstract]

Nature of Operations [Policy Text Block]

Basis of Accounting, Policy
[Policy Text Block]

Segment Reporting, Policy [Policy Text Block]

Reclassification,
Comparability Adjustment
[Policy Text Block]
New Accounting
Pronouncements, Policy
[Policy Text Block]

# 6 Months Ended Dec. 31, 2020

**Nature of Operations:** Consumers Bancorp, Inc. (the Corporation) is a bank holding company headquartered in Minerva, Ohio that provides, through its banking subsidiary, Consumers National Bank (the Bank), a broad array of products and services throughout its primary market area of Carroll, Columbiana, Jefferson, Stark, Summit, Wayne and contiguous counties in Ohio. The Bank's business involves attracting deposits from businesses and individual customers and using such deposits to originate commercial, mortgage and consumer loans in its primary market area.

Basis of Presentation: The consolidated financial statements for interim periods are unaudited and reflect all adjustments (consisting of only normal recurring adjustments), which, in the opinion of management, are necessary to present fairly the financial position and results of operations and cash flows for the periods presented. The unaudited financial statements are presented in accordance with the requirements of Form 10-Q and do not include all disclosures normally required by accounting principles generally accepted in the United States of America. The financial statements should be read in conjunction with the consolidated financial statements and notes thereto included in the Corporation's Form 10-K for the year ended June 30, 2020. The results of operations for the interim period disclosed herein are not necessarily indicative of the results that may be expected for a full year.

The consolidated financial statements include the accounts of the Corporation and the Bank. All significant inter-company transactions and accounts have been eliminated in consolidation.

**Segment Information:** The Corporation is a bank holding company engaged in the business of commercial and retail banking, which accounts for substantially all the revenues, operating income, and assets. Accordingly, all of the Corporation's operations are recorded in one segment, banking.

**Reclassifications:** Certain items in prior financial statements have been reclassified to conform to the current presentation. Any reclassifications had no impact on prior year net income or shareholders' equity.

Recently Issued Accounting Pronouncements Not Yet Effective: In June 2016, the Financial Accounting Standards Board (FASB) issued ASU 2016-13, Financial Instruments—Credit Losses (Topic 326): Measurement of Credit Losses on Financial Instruments. This ASU adds a new Topic 326 to the codification and removes the thresholds that companies apply to measure credit losses on financial instruments measured at amortized cost, such as loans, receivables, and held-to-maturity debt securities. Under current U.S. generally accepted accounting principles, companies generally recognize credit losses when it is probable that the loss has been incurred. The revised guidance will remove all current loss recognition thresholds and will require companies to recognize an allowance for credit losses for the difference between the amortized cost basis of a financial instrument and the amount of amortized cost that the corporation expects to collect over the instrument's contractual life. ASU 2016-13 also amends the credit loss measurement guidance for available-for-sale debt securities and beneficial interests in securitized financial assets. The guidance in ASU 2016-13 is effective for "public business entities," as defined in the guidance, that are SEC filers for fiscal years and for interim periods within those fiscal years beginning after December 15, 2019. Early adoption of the guidance is permitted for fiscal years beginning after December 15, 2018, including interim periods within those fiscal years. However, during July 2019, FASB unanimously voted for a proposal to delay this ASU to January 2023 for smaller reporting companies. On October 16, 2019, FASB approved a final ASU delaying the effective date. The new guidance is effective for annual and interim periods beginning after December 15, 2022 for certain entities, including smaller reporting companies. The Corporation is a smaller reporting company.

In March 2020, the FASB issued ASU 2020-04, "Facilitation of the Effects of Reference Rate Reform on Financial Reporting". The ASU is intended to provide relief for companies preparing for discontinuation of interest rates based on LIBOR, or other reference rates that may be

discontinued, and provides optional expedients and exceptions for applying GAAP to contract modifications and hedging relationships, subject to meeting certain criteria. The ASU also provides for a one-time sale and/or transfer to AFS or trading to be made for HTM debt securities that both reference an eligible reference rate and were classified as HTM before January 1, 2020. ASU 2020-04 is effective March 12, 2020 through December 31, 2022. The guidance requires companies to apply the guidance prospectively to contract modifications and hedging relationships while the one-time election to sell and/or transfer debt securities classified as HTM may be made any time after March 12, 2020. The Corporation does not expect ASU 2020-04 to have a material impact on its financial statements and disclosures.

## Note 2 - Acquisition (Tables)

## 6 Months Ended Dec. 31, 2020

## **Notes Tables**

Schedule of Recognized
Identified Assets Acquired and
Liabilities Assumed [Table
Text Block]

	Consideration Paid	\$	10,405
d	Net assets acquired:		
<u></u>	Cash and cash equivalents	\$ 833	
	Certificates of deposit in other financial institutions	11,839	
	Securities, available-for-sale	4,051	
	Federal bank and other restricted stocks, at cost	154	
	Loans, net	55,320	
	Premises and equipment	818	
	Core deposit intangible	270	
	Accrued interest receivable and other assets	140	
	Noninterest-bearing deposits	(11,979)	
	Interest-bearing deposits	(48,872)	
	Federal funds purchased	(2,348)	
	Federal Home Loan Bank advances	(491)	
	Other liabilities	(166)	
	Total net assets acquired	_	9,569
	Goodwill	\$	836

### **Note 3 - Securities (Tables)**

### 6 Months Ended Dec. 31, 2020

# Notes Tables Marketable Securities [Table Text Block]

Available –for-Sale	A	mortize Cost	d	Gross Unrealized Gains	l Unre	ross ealized osses		Fair ⁄alue
December 31, 2020								
U.S. Treasury	\$		199	\$ 31	2 \$		\$	501 8,109
Obligations of U.S. government-sponsored entities and agencies Obligations of state and political subdivisions		63,3	793 859	3,67				67,034
U.S. Government-sponsored mortgage-backed						(5)		
securities-residential		56,7	157	1,49	6	(5)		58,248
U.S. Government-sponsored mortgage-backed securities—commercial		7,0	004	7	5	(1)		7,078
U.S. Government-sponsored collateralized mortgage obligations- residential	- 	4,5	520	17	5			4,695
Total available-for-sale securities	\$	139,9	932	\$ 5,73		(6)	\$	145,665
Held-to-Maturity		rtized ost	Uı	Gross nrecognized Gains	Unreco	oss ognized sses		Fair Value
December 31, 2020	Φ.	2.226	Φ.	2.00			Φ.	2.60.5
Obligations of state and political subdivisions	\$ \$	3,336	\$ \$	269 269	\$ \$		\$ \$	3,605
Total held-to-maturity securities	Þ	3,330	<b></b>		<del></del>		<b>3</b>	3,605
Available-for-Sale	A	mortize Cost	d	Gross Unrealized Gains	Unre	ross ealized osses		Fair ⁄alue
June 30, 2020	_							
U.S. Treasury	\$	,	248		8 \$	_	\$	1,256
Obligations of U.S. government-sponsored entities and agencies Obligations of state and political subdivisions		10,1 60,3		39 3,14		_		10,532 63,492
U.S. government-sponsored mortgage-backed securities - residential		48,6		1,51		(4)		50,156
U.S. government-sponsored mortgage-backed securities - commercial		8,4	144	5	5	(2)		8,497
U.S. government-sponsored collateralized mortgage obligations - residential		9,7	712	28	5	(12)		9,985
Total available-for-sale securities	\$	138,5	525	\$ 5,41	1 \$	(18)	\$	143,918
Held-to-Maturity		rtized ost	Uı	Gross nrecognized Gains	Unreco	oss ognized sses		Fair Value
June 30, 2020			_	Gams		3303		
Obligations of state and political subdivisions	\$	3,541	\$	327	\$		\$	3,868
Total held-to-maturity securities	\$	3,541	\$	327	\$	_	\$	3,868
		nree Mon Decem 020				Months En		
Proceeds from sales and calls		020	_	2017				10
	\$	2,733	\$	1.650	2.7	733	6.1	
Gross realized gains	\$	2,733	\$	1,650 4	2,7	733	6,1 1	10
Gross realized gams Gross realized losses	\$		\$		2,7	31 23	1	10
Gross realized losses  Available-for-Sale	\$	31	\$	4 — Amor	tized est	31 23 Estimat Va	1 ted Fa lue	10 — air
Gross realized losses  Available-for-Sale  Due in one year or less	\$	31	\$	4 — Amor	rtized est 4,930	31 23 Estimat Va	ted Fa	10 — air
Available-for-Sale Due in one year or less Due after one year through five years	\$	31	\$	4 — Amor	tized ost 4,930 14,015	31 23 Estimat Va	1 ted Fa lue 5,0 14,5	10 — nir 006 445
Available-for-Sale Due in one year or less Due after one year through five years Due after five years through ten years	\$	31	\$	4 — Amor	tized est 4,930 14,015 13,700	31 23 Estimat Va	1 ted Fa lue 5,0 14,5 14,3	10 — nir 006 445 227
Available-for-Sale Due in one year or less Due after one year through five years	\$	31	\$	4 — Amor	tized ost 4,930 14,015	31 23 Estimat Va	1 ted Fa lue 5,0 14,5	10 — nir 006 445 227
Available-for-Sale Due in one year or less Due after one year through five years Due after five years through ten years Due after ten years		31	\$	4 ————————————————————————————————————	4,930 14,015 13,700 39,006 71,651	31 23 Estimat Va \$	1 ted Fa lue 5,0 14,5 14,3 41,7	10 
Gross realized losses  Available-for-Sale  Due in one year or less  Due after one year through five years  Due after five years through ten years  Due after ten years  Total		31	\$	4 ————————————————————————————————————	tized 98t 4,930 14,015 13,700 39,006 71,651	31 23 Estimat Va \$	1 ted Falue 5,0 14,5 14,3 41,7 75,6	10 ————————————————————————————————————
Gross realized losses  Available-for-Sale  Due in one year or less  Due after one year through five years  Due after five years through ten years  Due after ten years  Total  U.S. Government-sponsored mortgage-backed and related securit  Total available-for-sale securities  Held-to-Maturity		31	\$	Amor Cc \$	4,930 14,015 13,700 39,006 71,651 69,281 139,932	31 23 Estimat Va \$	1 ted Falue 5,0 14,5 14,3 41,7 75,6 70,0 145,6	10 — nir 1006 1445 127 1666 1444 1221 1665
Gross realized losses  Available-for-Sale  Due in one year or less  Due after one year through five years  Due after five years through ten years  Due after ten years  Total  U.S. Government-sponsored mortgage-backed and related securit  Total available-for-sale securities  Held-to-Maturity  Due after one year through five years		31	\$	4 ————————————————————————————————————	4,930 14,015 13,700 39,006 71,651 69,281 139,932	31 23 Estimat Va \$	1 ted Falue 5,0 14,5 14,3 41,7 75,6 70,0 145,6 3	10 — nir 1006 1445 127 1666 1444 121 1665
Gross realized losses  Available-for-Sale  Due in one year or less  Due after one year through five years  Due after five years through ten years  Due after ten years  Total  U.S. Government-sponsored mortgage-backed and related securit  Total available-for-sale securities  Held-to-Maturity  Due after one year through five years  Due after five years through ten years		31	\$	Amor Cc \$	4,930 14,015 13,700 39,006 71,651 69,281 139,932	31 23 Estimat Va \$	1 ted Falue 5,0 14,5 14,3 41,7 75,6 70,0 145,6	10 — nir 1006 1445 127 1666 1444 121 1665 156 135
Gross realized losses  Available-for-Sale  Due in one year or less  Due after one year through five years  Due after five years through ten years  Due after ten years  Total  U.S. Government-sponsored mortgage-backed and related securit  Total available-for-sale securities  Held-to-Maturity  Due after one year through five years  Due after five years through ten years  Due after ten years		31	\$	4 ————————————————————————————————————	4,930 14,015 13,700 39,006 71,651 69,281 139,932 334 667 2,335	31 23 Estimat Va \$	1 ted Falue 5,0 14,5 14,3 41,7 75,6 70,0 145,6 3	10 — nir 1006 1445 227 1666 1444 121 1665 156 135 114
Gross realized losses  Available-for-Sale  Due in one year or less  Due after one year through five years  Due after five years through ten years  Due after ten years  Total  U.S. Government-sponsored mortgage-backed and related securit  Total available-for-sale securities  Held-to-Maturity  Due after one year through five years  Due after five years through ten years		31 23		Amor Cc \$	4,930 14,015 13,700 39,006 71,651 69,281 139,932	31 23 Estimat Va \$	11 ted Fa lue 5,00 14,5 14,3 41,7 75,6 70,0 145,6 3,7 2,5 3,6	10 — nir 1006 1445 127 1666 144 144 1565 156 135 114

Fair

Value

Unrealized

Loss

Unrealized

Loss

Fair

Value

Unrealized

Loss

Schedule of Unrealized Loss on Investments [Table Text Block]

Available-for-sale

Schedule of Realized Gain (Loss) [Table Text Block]

Investments Classified by Contractual Maturity Date [Table Text Block]

Fair

Value

December 31, 2020									
U.S. Government-sponsored mortgage-backed securities – residential	1,198	(5)	_	_	1,198	(5)			
U.S. Government-sponsored mortgage-backed securities- commercial	1,422	(1)			1,422	(1)			
Total temporarily impaired	\$ 2,620	\$ (6)	\$ —	\$ —	\$ 2,620	\$ (6)			
	Less than	12 Months	12 Month	s or more	Total				
Available-for-sale	Fair Value	Unrealized Loss	Fair Value	Unrealized Loss	Fair Value	Unrealized Loss			
June 30, 2020									
U.S. Government-sponsored mortgage-backed securities – residential	_	_	625	(4)	625	(4)			
U.S. Government-sponsored mortgage-backed securities –	1,806	(2)			1,806	(2)			
commercial	1,000	(2)		_	1,000	(2)			
U.S. Government-sponsored collateralized mortgage	1,700	(12)	_	_	1,700	(12)			
U.S. Government-sponsored	,		\$ 625	<u>\$ (4)</u>	,	· ·			

### 6 Months Ended Dec. 31, 2020

### Note 4 - Loans (Tables)

### **Notes Tables**

Schedule of Accounts, Notes, Loans and Financing
Receivable [Table Text Block]

	Dec	ember 31, 2020	•	June 30, 2020
Commercial	\$	158,911	\$	158,667
Commercial real estate:				
Construction		3,802		16,235
Other		250,529		229,029
1 – 4 Family residential real estate:				
Owner occupied		93,726		90,494
Non-owner occupied		20,210		19,370
Construction		6,026		9,344
Consumer		24,230		21,334
Subtotal		557,434		544,473
Net Deferred loan fees and costs		(177)		(1,612)
Allowance for loan losses		(5,912)		(5,678)
Net Loans	\$	551,345	\$	537,183

Financing Receivable, Allowance for Credit Loss [Table Text Block]

						93	,726		90,49	94		
							,210		19,3			
						6	,026		9,3	44		
						24	,230		21,33	34		
						557	,434		544,4	73		
										_		
					\$	551	,345 \$		537,13	83		
Comr	nercial		Real	Resi	idential Real	Con	sumer		Total	_		
\$		\$		\$		\$		\$				
	(82)		122									
	_				_		. ,					
•	959	0		6	1.020	•		¢.				
3	838	Con	nmercial	1-4 Resi	Family idential	3	209	<u>\$</u>	3,9	<u>12</u>		
Comn	nercial					Con	sumer	_	Total	_		
\$		\$	3,623	\$	989	\$	119	\$				
			192		39							
	(22)		_				. ,					
		_						_				
\$	858	Con	nmercial	1-4 Resi	Family idential	<u>\$</u>	209	<u>\$</u>	5,9	<u>12</u>		
Comn	nercial					Con	sumer	_	Total	_		
\$		\$		\$		\$		\$				
	117		6		64							
			1									
•	766	•		•		•		•	4.00			
3	700	Con	nmercial	1-4 Resi	Family idential	3	62	<u> </u>	4,0	93		
Comn	nercial	I	Estate	E	state	Con	sumer		Total	_		
\$	660	\$	2,575	\$	494	\$	59	\$	3,78	88		
	106		75		120		14		3	15		
	_		_		_		(23)		(2	23)		
	_		2		1		12			15		
\$	766	\$	2,652	\$	615	\$	62	\$	4,09	95		
				Co	mmercial		-4 Family esidentia	_	-	=		
		Con	nmercial		Estate		Estate		Cons	umer		Total
oans:												
74115.		\$	3	\$	_	- \$		6	\$	_	\$	9
oans:					101			84		_		185
	irment		_									
			855		3,716	,	9.	38		209		5,718
for impa		\$	855 858	\$	3,716 3,817		9. 1,0	_	\$	209	\$	5,718 5,912
for impa		\$		\$				_	\$		\$	
for impa d for imp			858		3,817	\$	1,0	28				5,912
for impa d for imp	airment	<u>\$</u>	148	<u>\$</u>	3,817 1,151	\$ \$	1,0	28 44	\$	209	\$	2,343
for impa d for imp irment for impa	airment		148 661		3,817 1,151 7,824	\$	1,0 1,0 23,2	28 44 20		209 — 8,935		2,343 40,640
for impa d for imp	airment		148		3,817 1,151	\$	1,0	44 20 33		209		2,343
	\$	\$ 858  Commercial  \$ 947 (67) (22)  \$ 858  Commercial  \$ 649 117 \$ 766  Commercial	S   940   S   (82)	\$ 940 \$ 3,694   (82)	Commercial Real Real Real Restate         <	Commercial         Real Estate         Real Estate           \$ 940         \$ 3,694         \$ 997           (82)         122         31           —         —         —           —         —         —           —         —         —           —         —         —           —         —         —           S 858         \$ 3,817         \$ 1.4 Family           Residential Real Estate         Estate           S 947         \$ 3,623         \$ 989           (67)         192         39           (22)         —         —           —         —         —           S 858         \$ 3,817         \$ 1,028           I-4 Family         Residential Real Estate           Commercial         Estate         Estate           \$ 649         \$ 2,645         \$ 550           117         6         64           —         —         —           —         —         —           S 766         \$ 2,652         \$ 615           I-4 Family         Residential Real Estate           Commercial         Residential Real Estate	Commercial Real Estate	Commercial   Real   Estate   Consumercial   Real   Real   Estate   Consumercial   Real   Real   Real   Estate   Consumercial   Real   Real   Real   Real   Estate   Consumercial   Real   Real   Real   Estate   Consumercial   Real   Real   Estate   Consumercial   Real   Real   Estate   Consumercial   Real   Real   Estate   Consumercial   Real   Real   Estate   Consumercial   Real   Real   Estate   Consumercial   Real   R	Commercial   Commercial   Real   Estate   Consumer   Consumer   Commercial   Real   Estate   Consumer   Consumer   Commercial   Real   Estate   Consumer   Commercial   Commercial   Real   Estate   Consumer   Commercial   Commercial   Real   Estate   Consumer   Commercial   Real   Consumer   Commercial   Real   Consumer   Commercial   Consumer   Commercial   Consumer   Commercial   Consumer   Commercial   Consumer   Consumer   Commercial   Consumer   Consumer	Commercial Real   Commercial Real   Estate   Consumer   Commercial Real Real   Estate   Consumer   Commercial Real Real Real Real Real Real Real Re	Commercial   Commercial   Real   Estate   Consumer   Commercial   Real   Estate   Consumer   Total   Commercial   Real   Estate   Consumer   Total   Commercial   Real   Estate   Consumer   Total   Commercial   Commercial   Real   Estate   Consumer   Total   Commercial   Comme	Commercial Real   Estate   Consumer   Total

						Real		Real				
			C	ommercial		Estate		Estate		Consumer		Total
Allowance for loan losses:												
Ending allowance balance attributable												
Individually evaluated for impairmen			\$	28	\$	6	\$	_	\$	_	\$	34
Acquired loans collectively evaluate						103		94				197
Originated loans collectively evaluat	ed for imp	airmen		919		3,514		895	_	119	_	5,447
Total ending allowance balance			\$	947	\$	3,623	\$	989	\$	119	\$	5,678
Recorded investment in loans:												
Loans individually evaluated for imp			\$	179	\$	1,045	\$	699	\$	_	\$	1,923
Acquired loans collectively evaluate				1,095		8,072		27,252		12,550		48,969
Originated loans collectively evaluat	ed for imp	_	156,054		236,840		92,168		8,843		493,905	
Total ending loans balance			\$	157,328	\$	245,957	\$	120,119	\$	21,393	\$	544,797
		Aso	of De	cember 31, 2	020			Six Mont	hs ei	nded Decembe	er 31	. 2020
	Unpai	d		Í	A 11a	owance for		Average		Interest		Cash Basis
	Princip		г	Recorded		an Losses		ecorded		Income		Interest
	Balanc		_	vestment		Ilocated	-	vestment	1	Recognized	т	Recognized
With no related allowance recorded:	Dalanc			ivesilient		inocated		vestilient		Recognized		Recognized
Commercial real estate:												
Other	\$ 1	,286	\$	1,151	\$	_	\$	867	\$	4	\$	4
1-4 Family residential real estate:												
Owner occupied		947		797				629		11		11
Non-owner occupied		274		217		_		225		_		_
With an allowance recorded:												
Commercial		147		148		3		160		4		4
Commercial real estate:												
Other				_		_		205		6		6
								-00		O		O
1-4 Family residential real estate:												
1-4 Family residential real estate: Owner occupied		30		30		6		5				

2,091 \$ Cash Basis Total Interest Average Recorded Interest Income Investment Recognized Recognized With no related allowance recorded: Commercial real estate: Other 907 \$ 3 \$ 1-4 Family residential real estate: Owner occupied 720 5 5 Non-owner occupied 220 With an allowance recorded: 2 Commercial 150 2 Commercial real estate: Other 204 3 3 1-4 Family residential real estate: Owner occupied 10 Total 2,211 13 13

		Α	s of Ju	ine 30, 202	20			19				
	Pri	Unpaid Principal Recorded Balance Investment		Lo	Allowance for Loan Losses Allocated		Average lecorded vestment	Interest Income Recognized		Int	Basis erest gnized	
With no related allowance recorded:												
Commercial real estate:												
Other	\$	922	\$	836	\$	_	\$	303	\$	87	\$	87
1-4 Family residential real estate:												
Owner occupied		604		463		_		25		7		7
Non-owner occupied		284		236		_		254		_		_
With an allowance recorded:												
Commercial		176		179		28		167		5		5
Commercial real estate:												
Other		209		209		6		219		6		6
Total	\$	2,195	\$	1,923	\$	34	\$	968	\$	105	\$	105
			Av	erage		Interes	t Cash Basis					
			Re	corded		Incom	e		Interest			

	Rec	Recorded		icome	Interest	
	Inve	Investment		ognized	Recognized	
With no related allowance recorded:				_		
Commercial real estate:						
Other	\$	244	\$	1	\$	1
1-4 Family residential real estate:						
Owner occupied		10		_		_
Non-owner occupied		250		_		_
With an allowance recorded:						
Commercial		165		2		2
Commercial real estate:						
Other		218		3		3
Total	\$	887	\$	6	\$	6
				December	31, 2020	)

Financing Receivable, Past Due [Table Text Block]

Impaired Financing Receivables [Table Text

Block]

June 30, 2020 Loans Past Due Loans Past Due Over 90 Days Over 90 Days Still

Still

			Non-accrual	Accruing	Non-accrual	Accruing
Commercial			\$ —	\$ —	\$ 21	\$ —
Commercial real estate:						
Other			902	_	785	_
1 – 4 Family residential:						
Owner occupied			821	_	143	29
Non-owner occupied			217	_	236	_
Consumer			_	_	_	12
Total			\$ 1,940	\$ —	\$ 1,185	\$ 41
	Days P	ast Due				
	30 - 59 60	- 89 90 Days or	Total Lo	oans Not		
	Days Days	ays Greater	Past Due P	ast Due Tot	al	
Commercial	<u>\$</u> \$	_ \$	\$ — <b>\$</b>	159,140 \$ 15	9,140	
Commercial real estate:						
Construction	_		_	3,791	3,791	
Other	_	— 629	629	250,044 25	0,673	

269

723

4

93,926 20,223 6,121 94,649 20,223 6,125

Construction	4	_	_	4	0,121	0,123
Consumer	261	49		310	23,953	24,263
Total	\$ 715	\$ 53	\$ 898	\$ 1,666	\$ 557,198	\$ 558,864
		Days Past Due				 
	30 - 59	30 - 59 60 - 89 90 Days or		Total	Loans Not	
	Days	Days	Greater	Past Due	Past Due	Total
Commercial	\$ —	\$ —	\$ 21	\$ 21	\$ 157,307	\$ 157,328
Commercial real estate:						
Construction	_	_	_	_	16,241	16,241
Other	_	2	628	630	229,086	229,716
1-4 Family residential:						
Owner occupied	_	_	172	172	91,102	91,274
Non-owner occupied	_	_	_	_	19,410	19,410
Construction	_	_	_	_	9,435	9,435
Consumer	127	49	12	188	21,205	21,393
Total	\$ 127	\$ 51	\$ 833	\$ 1,011	\$ 543,786	\$ 544,797

4

450

4

Financing Receivable Credit Quality Indicators [Table Text Block]

1-4 Family residential:

Owner occupied
Non-owner occupied
Construction

Total	\$ 127	\$	51	\$	833	\$	1,011	\$	543,786	\$	544,797
				As of December 3				2020	0		
		_			Special						Not
			Pass		Mention	Subst	tandard	D	oubtful		Rated
Commercial		\$	158,054	\$	453	\$	359	\$		\$	274
Commercial real estate:											
Construction			3,791		_		_		_		_
Other			241,702	1,933 5,075 90				902		1,061	
1-4 Family residential real estate:											
Owner occupied			1,988		_		18		603		92,040
Non-owner occupied			19,265		173		210		217		358
Construction			1,306		_		_		_		4,819
Consumer			687		_		_		_		23,576
Total		\$	426,793	\$	2,559	\$	5,662	\$	1,722	\$	122,128
		_		_	As	of Jur	ne 30, 20	20		_	
					Special						Not
			Pass		Mention	Subst	tandard	D	oubtful		Rated
Commercial		\$	152,911	\$	143	\$	3,979	\$	21	\$	274
Commercial real estate:											
Construction			16,241		_		_		_		_
Other			220,311		1,469		5,378		785		1,773
1-4 Family residential real estate:											
Owner occupied			2,419		_		334		_		88,521
Non-owner occupied			18,435		186		223		236		330
Construction			3,234		_		_		_		6,201
Consumer			153								21,240
Total		\$	413,704	\$	1,798	\$	9,914	\$	1,042	\$	118,339

## **Note 5 - Fair Value (Tables)**

## 6 Months Ended Dec. 31, 2020

Fair Value Measurements at

### **Notes Tables**

Schedule of Fair Value, Assets and Liabilities Measured on Recurring Basis [Table Text Block]

			December 31, 2020						
	Balance at December 31, 2020	L	evel 1	Level 2	Level 3	3			
Assets:									
U.S. Treasury	\$ 501	\$	_	\$ 501	\$	_			
Obligations of U.S. government-sponsored entities and agencies	8,109		_	8,109		—			
Obligations of states and political subdivisions	67,034		_	67,034		_			
U.S. Government-sponsored mortgage-backed securities – residential	58,248		_	58,248		_			
U.S. Government-sponsored mortgage-backed securities – commercial	7,078		_	7,078		_			
U.S. Government-sponsored collateralized mortgage obligations -	4,695			4,695					
residential	4,093		_	4,093		_			
Equity securities	400		_	400		_			
			Fair V	Value Measurem June 30, 2020	ents at				
	Balance at June 30, 2020	L	evel 1	Level 2	Level 3	3			
Assets:									
Securities available-for-sale:									
Obligations of U.S. Treasury	\$ 1,256	\$	_	\$ 1,256	\$	—			
Obligations of government-sponsored entities	10,532		_	10,532		—			
Obligations of states and political subdivisions	63,492		_	63,492		—			
U.S. Government-sponsored mortgage-backed securities - residential	50,156		_	50,156		_			
U.S. Government-sponsored mortgage-backed securities – commercial	8,497		_	8,497		_			
U.S. Government-sponsored collateralized mortgage obligations	9,985		_	9,985		_			

Fair Value, by Balance Sheet Grouping [Table Text Block]

	December 31, 2020				June 30, 2020			
	Carrying Amount		Estimated Fair Value		Carrying Amount		Estimated Fair Value	
Financial Assets:								
Level 1 inputs:								
Cash and cash equivalents	\$ 9,045	\$	9,045	\$	9,659	\$	9,659	
Level 2 inputs:								
Certificates of deposit in other financial institutions	8,599		8,819		11,635		11,889	
Loans held for sale	4,770		4,907		3,507		3,566	
Accrued interest receivable	2,327		2,327		2,646		2,646	
Level 3 inputs:								
Securities held-to-maturity	3,336		3,605		3,541		3,868	
Loans, net	551,345		565,397		537,183		548,247	
Financial Liabilities:								
Level 2 inputs:								
Demand and savings deposits	540,811		540,811		517,973		517,973	
Time deposits	107,551		108,158		115,382		116,238	
Short-term borrowings	13,275		13,275		6,943		6,943	
Federal Home Loan Bank advances	18,083		18,423		31,161		31,571	
Accrued interest payable	88		88		107		107	

## Note 6 - Earnings Per Share (Tables)

**Notes Tables** 

Schedule of Earnings Per Share, Basic and Diluted [Table Text Block]

## 6 Months Ended Dec. 31, 2020

	For the Three Months Ended December 31,						Months Ende mber 31,				
			2020		2019		2	020		201	9
Basic:											
Net income available to common shareholders		\$	2,5	07	\$ 1.	,440	\$	4,90	8 \$		2,943
Weighted average common shares outstanding			3,017,2	68	2,737	,800	3,	015,74	1	2,73	7,755
Basic income per share		\$	0.	.83	\$	0.53	\$	1.6	3 \$		1.07
Diluted:											
Net income available to common shareholders	\$		2,507	\$	1,440	\$	4,9	08 \$		2,943	
Weighted average common shares outstanding		3,0	17,268		2,737,800		3,015,7	41	2,73	7,755	
Dilutive effect of restricted stock			21		_			_		_	
Total common shares and dilutive potential common shares		3,0	17,289	_	2,737,800	_	3,015,7	41	2,73	7,755	
Dilutive income per share	\$		0.83	\$	0.53	\$	1.	.63 \$	,	1.07	

### Note 7 - Accumulated Other Comprehensive Income (Loss) (Tables)

## **Notes Tables**

Schedule of Accumulated
Other Comprehensive Income
(Loss) [Table Text Block]

# 6 Months Ended Dec. 31, 2020

	Pretax		Tax Effect		After-tax	Affected Line Item in Consolidated Statements of Income
Balance as of September 30, 2020	\$ 5,463	\$	(1,147)	\$	4,316	
Unrealized holding loss on available-for-sale securities arising during the period	278		(59)		219	
Amounts reclassified from accumulated other comprehensive income	(8)		2		(6)	(a)(b)
Net current period other comprehensive loss	270		(57)		213	
Balance as of December 31, 2020	\$ 5,733	\$	(1,204)	\$	4,529	
Balance as of September 30, 2019	\$ 2,694	\$	(566)	\$	2,128	
Unrealized holding gain on available-for-sale securities arising during the period	(28)		5		(23)	
Amounts reclassified from accumulated other comprehensive income	(4)		1		(3)	(a)(b)
Net current period other comprehensive income	(32)		6		(26)	
Balance after reclassification as of December 31, 2019	\$ 2,662	\$	(560)	\$	2,102	
						A CC 4 1 T 1
	Pretax		Tax Effect		After-tax	Affected Line Item in Consolidated Statements of Income
Balance as of June 30, 2020	\$ Pretax 5,393	\$	Tax Effect (1,133)	\$	After-tax 4,260	Item in Consolidated Statements of
Balance as of June 30, 2020 Unrealized holding gain on available-for-sale securities arising during the period	\$	\$		\$		Item in Consolidated Statements of
Unrealized holding gain on available-for-sale	\$ 5,393	\$	(1,133)	\$	4,260	Item in Consolidated Statements of
Unrealized holding gain on available-for-sale securities arising during the period  Amounts reclassified from accumulated other	\$ 5,393 348	\$	(1,133) (73) 2 (71)	_	4,260	Item in Consolidated Statements of Income
Unrealized holding gain on available-for-sale securities arising during the period  Amounts reclassified from accumulated other comprehensive income	\$ 5,393 348 (8)	\$	(1,133) (73) 2	_	4,260 275 (6)	Item in Consolidated Statements of Income
Unrealized holding gain on available-for-sale securities arising during the period  Amounts reclassified from accumulated other comprehensive income  Net current period other comprehensive income	 5,393 348 (8) 340	_	(1,133) (73) 2 (71)	\$	4,260 275 (6) 269	Item in Consolidated Statements of Income
Unrealized holding gain on available-for-sale securities arising during the period  Amounts reclassified from accumulated other comprehensive income  Net current period other comprehensive income  Balance as of December 31, 2020  Balance as of June 30, 2019  Unrealized holding gain on available-for-sale securities arising during the period	\$ 5,393 348 (8) 340 5,733	\$	(1,133) (73) 2 (71) (1,204)	\$	4,260 275 (6) 269 4,529	Item in Consolidated Statements of Income
Unrealized holding gain on available-for-sale securities arising during the period Amounts reclassified from accumulated other comprehensive income Net current period other comprehensive income Balance as of December 31, 2020  Balance as of June 30, 2019 Unrealized holding gain on available-for-sale	\$ 5,393 348 (8) 340 5,733	\$	(1,133) (73) 2 (71) (1,204) (416)	\$	4,260 275 (6) 269 4,529	Item in Consolidated Statements of Income
Unrealized holding gain on available-for-sale securities arising during the period  Amounts reclassified from accumulated other comprehensive income  Net current period other comprehensive income  Balance as of December 31, 2020  Balance as of June 30, 2019  Unrealized holding gain on available-for-sale securities arising during the period  Amounts reclassified from accumulated other	\$ 5,393 348 (8) 340 5,733 1,982 790	\$	(1,133) (73) 2 (71) (1,204) (416) (166)	\$	4,260 275 (6) 269 4,529 1,566 624	Item in Consolidated Statements of Income  (a)(b)
Unrealized holding gain on available-for-sale securities arising during the period  Amounts reclassified from accumulated other comprehensive income  Net current period other comprehensive income  Balance as of December 31, 2020  Balance as of June 30, 2019  Unrealized holding gain on available-for-sale securities arising during the period  Amounts reclassified from accumulated other comprehensive income	\$ 5,393 348 (8) 340 5,733 1,982 790 (110)	\$	(1,133) (73) 2 (71) (1,204) (416) (166) 22	\$ \$	4,260 275 (6) 269 4,529 1,566 624 (88)	Item in Consolidated Statements of Income  (a)(b)

Note 1 - Summary of Significant Accounting Policies (Details Textual) 6 Months Ended Dec. 31, 2020

Number of Reportable Segments 1

Note 2 - Acquisition (Details Textual) - USD (\$) \$ in Thousands	Dec. 29, 2020	Jan. 01, 2020	6 Months Ended Dec. 31, 2019	12 Months Ended Jun. 30, 2020
Core Deposits [Member]				
Finite-Lived Intangible Asset, Useful Life (Year)			10 years	
CFBank Branches [Member]				
Business Combination, Recognized Identifiable Assets Acquired and	\$			
Liabilities Assumed, Certificates of Deposit	100,000	)		
Business Combination, Recognized Identifiable Assets Acquired and	15,000			
Liabilities Assumed, Current Assets, Marketable Securities	,			
Business Combination, Recognized Identifiable Assets Acquired and Liabilities Assumed, Loans Receivable, Net	\$ 3,100			
Business Combination, Consideration Transferred, Deposit Premium,				
Percent	1.75%			
CFBank Branches [Member]   Single Family Residential Mortgage Loans				
and Home Equity Lines of Credit [Member]   Maximum [Member]				
Business Combination, Recognized Identifiable Assets Acquired and	\$			
Liabilities Assumed, Loans Receivable, Net	13,500			
CFBank Branches [Member]   Commercial and Commercial Real Estate				
Loans [Member]   Minimum [Member]				
Business Combination, Recognized Identifiable Assets Acquired and	\$			
Liabilities Assumed, Loans Receivable, Net	15,000			
Peoples Bancorp of Mt. Pleasant, Inc. [Member]				
Business Combination, Recognized Identifiable Assets Acquired and		\$		
Liabilities Assumed, Certificates of Deposit		11,839		
Business Combination, Recognized Identifiable Assets Acquired and		4,051		
<u>Liabilities Assumed, Current Assets, Marketable Securities</u> Business Combination, Recognized Identifiable Assets Acquired and				
Liabilities Assumed, Loans Receivable, Net		55,320		
Business Combination, Consideration Transferred, Total		\$		
Business Comomation, Consideration Transferred, Total		10,405		
Business Acquisition, Equity Interest Issued or Issuable, Number of Shares		269,920		
(in shares)		,		
Payments to Acquire Businesses, Gross		\$ 5,128		
Business Combination, Recognized Identifiable Assets Acquired and Liabilities Assumed, Assets, Total			\$ 72,016	
Business Combination, Recognized Identifiable Assets Acquired and				
Liabilities Assumed, Loans			55,273	
Business Combination, Recognized Identifiable Assets Acquired and				
Liabilities Assumed, Deposits			\$ 60,826	
Business Combination, Recognized Identifiable Assets Acquired and Liabilities Assumed, Loans, Credit-related Adjustment		(890)		

Business Combination, Recognized Identifiable Assets Acquired and	\$ 937	
Liabilities Assumed, Loans, Non-credit-related Adjustments	\$ 937	
Business Combination, Acquisition Related Costs		\$ 827
Business Combination, Acquisition Related Costs, Net of Tax		\$ 680

Note 2 - Acquisition - Assets Acquired and Liabilities Assumed (Details) - USD (\$) \$ in Thousands	Jan. 01, Dec. 31, Jun. 30, 2020 2020 2020
Goodwill	\$ 836 \$ 836
Peoples Bancorp of Mt. Pleasant, Inc. [Member]	
Business Combination, Consideration Transferred, Total	\$ 10,405
Cash and cash equivalents	833
Business Combination, Recognized Identifiable Assets Acquired and Liabilities	11,839
Assumed, Certificates of Deposit	11,037
Business Combination, Recognized Identifiable Assets Acquired and Liabilities	4,051
Assumed, Current Assets, Marketable Securities	•
Federal bank and other restricted stocks, at cost	154
Business Combination, Recognized Identifiable Assets Acquired and Liabilities	55,320
Assumed, Loans Receivable, Net	•
Premises and equipment	818
Core deposit intangible	270
Accrued interest receivable and other assets	140
Noninterest-bearing deposits	(11,979)
<u>Interest-bearing deposits</u>	(48,872)
Federal funds purchased	(2,348)
Federal Home Loan Bank advances	(491)
Other liabilities	(166)
Total net assets acquired	9,569
Goodwill	\$ 836

Note 3 - Securities (Details	3 Mont	hs Ended	6 Months Ende		
Textual) - USD (\$) \$ in Thousands	Dec. 31, 2020	Dec. 31, 2019	Dec. 31, 2020	Dec. 31, 2019	
Available-for-sale Securities Income Tax Provision on Gross Realized Gains	\$ 2	\$ 1	\$ 2	\$ 22	

## Note 3 - Securities -Amortized Cost and Fair Value of Securities Availablefor-sale and Securities Heldto-maturity (Details) - USD (\$)

**\$** in Thousands

Dec. 31, 2020 Jun. 30, 2020

Available-for-sale securities, amortized cost	\$ 139,932	\$ 138,525
Available-for-sale securities, gross unrealized gains	5,739	5,411
Available-for-sale securities, gross unrealized losses	(6)	(18)
Available-for-sale securities, fair value	145,665	143,918
	· ·	*
Held-to-maturity, amortized cost	3,336	3,541
Held-to-maturity, gross unrecognized gains	269	327
Held-to-maturity, gross unrecognized losses	2.605	2.060
Held-to-maturity, fair value	3,605	3,868
US Treasury Securities [Member]		
Available-for-sale securities, amortized cost	499	1,248
Available-for-sale securities, gross unrealized gains	2	8
Available-for-sale securities, gross unrealized losses		
Available-for-sale securities, fair value	501	1,256
US States and Political Subdivisions Debt Securities [Member]		
Available-for-sale securities, amortized cost	63,359	60,343
Available-for-sale securities, gross unrealized gains	3,675	3,149
Available-for-sale securities, gross unrealized losses		
Available-for-sale securities, fair value	67,034	63,492
Held-to-maturity, amortized cost	3,336	3,541
Held-to-maturity, gross unrecognized gains	269	327
Held-to-maturity, gross unrecognized losses		
Held-to-maturity, fair value	3,605	3,868
US Government-sponsored Enterprises Debt Securities [Member	1	
Available-for-sale securities, amortized cost	7,793	10,133
Available-for-sale securities, gross unrealized gains	316	399
Available-for-sale securities, gross unrealized losses		
Available-for-sale securities, fair value	8,109	10,532
Residential Mortgage Backed Securities [Member]	,	,
Available-for-sale securities, amortized cost	56,757	48,645
Available-for-sale securities, gross unrealized gains	1,496	1,515
Available-for-sale securities, gross unrealized losses	(5)	(4)
Available-for-sale securities, fair value	58,248	50,156
Commercial Mortgage Backed Securities [Member]	50,210	30,130
Available-for-sale securities, amortized cost	7,004	8,444
Available-for-sale securities, gross unrealized gains	75	55
Available-for-sale securities, gross unrealized losses		
	(1)	(2)
Available-for-sale securities, fair value	7,078	8,497

Collateralized Mortgage Obligations [Member]		
Available-for-sale securities, amortized cost	4,520	9,712
Available-for-sale securities, gross unrealized gains	175	285
Available-for-sale securities, gross unrealized losses		(12)
Available-for-sale securities, fair value	\$ 4,695	\$ 9,985

Note 3 - Securities - Proceeds	s 3 Mo	nths Ended	6 Mo	onths Ended
From Sales and Calls of Available-for-sale Securities (Details) - USD (\$)	Dec. 31, 20	020 Dec. 31, 20	)19 Dec. 31, 20	020 Dec. 31, 2019
\$ in Thousands				
Proceeds from sales and calls	\$ 2,733	\$ 1,650	\$ 2,733	\$ 6,110
Gross realized gains	31	4	31	110
Gross realized losses	\$ 23		\$ 23	

## Note 3 - Securities -**Amortized Cost and Fair**

Values of Debt Securities by  Contractual Maturity  (Details) - USD (\$)  \$ in Thousands	Dec. 31, 2020	Jun. 30, 2020
Available-for-sale Securities, Debt Maturities, Next Twelve Months, Amortized Cost	\$ 4,930	
<u>Basis</u>	\$ 4,930	
Available-for-sale Securities, Debt Maturities, Next Twelve Months, Fair Value	5,006	
Available-for-sale Securities, Debt Maturities, Year Two Through Five, Amortized Cos Basis	<u>t</u> 14,015	
Available-for-sale Securities, Debt Maturities, Year Two Through Five, Fair Value	14,545	
Available-for-sale Securities, Debt Maturities, Year Six Through Ten, Amortized Cost Basis	13,700	
Available-for-sale Securities, Debt Maturities, Year Six Through Ten, Fair Value	14,327	
Available-for-sale Securities, Debt Maturities, after Ten Years, Amortized Cost Basis	39,006	
Available-for-sale Securities, Debt Maturities, after Ten Years, Fair Value	41,766	
<u>Available-for-sale Securities, Debt Maturities, Single Maturity Date, Amortized Cost Basis</u>	71,651	
Available-for-sale Securities, Debt Maturities, Single Maturity Date	75,644	
Available-for-sale securities, Amortized Cost Basis	139,932	\$ 138,525
Total available-for-sale securities	145,665	143,918
Due after one year through five years, Net Carrying Amount	334	173,710
Due after one year through five years, Fair Value	356	
Held-to-maturity Securities, Debt Maturities, after Five Through Ten Years, Net		
Carrying Amount	667	
Held-to-maturity Securities, Debt Maturities, Year Five Through Ten, Fair Value	735	
Held-to-maturity Securities, Debt Maturities, after Ten Years, Net Carrying Amount	2,335	
Held-to-maturity Securities, Debt Maturities, after Ten Years, Fair Value	2,514	
Total held-to-maturity securities	3,336	3,541
Held-to-maturity Securities, Fair Value	3,605	\$ 3,868
U.S. Government-sponsored Mortgage-backed and Related Securities [Member]		
Available-for-sale Securities, Debt Maturities, without Single Maturity Date, Amortized Cost Basis	69,281	
Available-for-sale Securities, Debt Maturities, without Single Maturity Date, Fair Value	\$ 70,021	

Note 3 - Securities - Securities With Unrealized and Unrecognized Losses (Details) - USD (\$) \$ in Thousands	Dec. 31, 2020	Jun. 30, 2020
Available-for-sale Securities, Continuous Unrealized Loss Position, Less than Twelve	\$ 2,620	\$ 3,506
Months, Fair Value		,
Available-for-sale Securities, Continuous Unrealized Loss Position, Less than 12  Months, Aggregate Loss	(6)	(14)
Available-for-sale Securities, Continuous Unrealized Loss Position, Twelve Months or Longer, Fair Value		625
Available-for-sale Securities, Continuous Unrealized Loss Position, 12 Months or Longer, Aggregate Loss		(4)
Available-for-sale Securities, Continuous Unrealized Loss Position, Fair Value	2,620	4,131
Available-for-sale Securities, Continuous Unrealized Loss Position, Aggregate Loss Residential Mortgage Backed Securities [Member]	(6)	(18)
Available-for-sale Securities, Continuous Unrealized Loss Position, Less than Twelve		
Months, Fair Value	1,198	
Available-for-sale Securities, Continuous Unrealized Loss Position, Less than 12  Months, Aggregate Loss	(5)	
Available-for-sale Securities, Continuous Unrealized Loss Position, Twelve Months or Longer, Fair Value		625
Available-for-sale Securities, Continuous Unrealized Loss Position, 12 Months or Longer, Aggregate Loss		(4)
Available-for-sale Securities, Continuous Unrealized Loss Position, Fair Value	1,198	625
Available-for-sale Securities, Continuous Unrealized Loss Position, Aggregate Loss	(5)	(4)
Commercial Mortgage Backed Securities [Member]	(3)	(4)
Available-for-sale Securities, Continuous Unrealized Loss Position, Less than Twelve		
Months, Fair Value	1,422	1,806
Available-for-sale Securities, Continuous Unrealized Loss Position, Less than 12  Months, Aggregate Loss	(1)	(2)
Available-for-sale Securities, Continuous Unrealized Loss Position, Twelve Months or		
Longer, Fair Value		
Available-for-sale Securities, Continuous Unrealized Loss Position, 12 Months or		
Longer, Aggregate Loss		
Available-for-sale Securities, Continuous Unrealized Loss Position, Fair Value	1,422	1,806
Available-for-sale Securities, Continuous Unrealized Loss Position, Aggregate Loss	\$ (1)	(2)
Collateralized Mortgage Obligations [Member]		
Available-for-sale Securities, Continuous Unrealized Loss Position, Less than Twelve Months, Fair Value		1,700
Available-for-sale Securities, Continuous Unrealized Loss Position, Less than 12		(1.5)
Months, Aggregate Loss		(12)
Available-for-sale Securities, Continuous Unrealized Loss Position, Twelve Months or		
Longer, Fair Value		

Available-for-sale Securities, Continuous Unrealized Loss Position, 12 Months or	
Longer, Aggregate Loss	
Available-for-sale Securities, Continuous Unrealized Loss Position, Fair Value	1,700
Available-for-sale Securities, Continuous Unrealized Loss Position, Aggregate Loss	\$ (12)

Note 4 - Loans (Details	3 Mont	hs Ended	6 Montl	hs Ended	
Textual)		, Dec. 31,			
\$ in Thousands	2020	2019	2020	2019	2020
	-	) USD (\$)		USD (\$)	
Loans and Leases Receivable, Gross, Total	\$ 557,434		\$ 557,434		\$ 544,473
Interest Receivable	1,607		1,607		1,936
Financing Receivable, Nonaccrual	\$ 1,940		\$ 1,940		1,185
Financing Receivable, Number of Loans Deferred Under Loan	•		ŕ		1,103
Modification Program	18		18		
Financing Receivable, Deferred Payments	\$ 667		\$ 667		
Financing Receivable, Troubled Debt Restructuring	934		934		974
Financing Receivable, Troubled Debt Restructuring, Commitmen	t o		0		0
to Lend	-0		0		0
Troubled Debt Restructuring, Debtor, Subsequent Periods,	\$ 3		\$ 3		12
Contingent Payments, Amount, Total	\$ 3		<b>Ф</b> 3		12
Financing Receivable, Modifications, Number of Contracts	0	0	0	0	
Financing Receivables, Impaired, Troubled Debt Restructuring,	\$ 0	\$ 0	\$ 0	\$ 0	
Write-down	Ψ	Ψ 0	Ψ 0	Ψ	
Financing Receivable, Troubled Debt Restructuring, Subsequent	0	0	0	0	
Default, Number of Contracts					
Threshold Amount of Loans Outstanding to Perform Credit	\$ 100		\$ 100		
Analysis Financial Asset, Equal to or Greater than 90 Days Past Due					
[Member]					
Financing Receivable, Nonaccrual	898		898		792
Financial Asset, 60 to 89 Days Past Due [Member]	070		0,0		,,,_
Financing Receivable, Nonaccrual					2
Commercial Portfolio Segment [Member]					
Loans and Leases Receivable, Gross, Total	158,911		158,911		158,667
Financing Receivable, Nonaccrual			,		21
Commercial Portfolio Segment [Member]   SBA CARES Act					
Paycheck Protection Program [Member]					
Loans and Leases Receivable, Gross, Total	52,539		52,539		66,606
Commercial Portfolio Segment [Member]   Mortgage Loan					
Warehouse Line of Credit [Member]					
Loans and Leases Receivable, Gross, Total	47,268		47,268		32,869
Non-accrual Loans [Member]					
Financing Receivable, Nonaccrual	\$ 1,042		\$ 1,042		\$ 391

Note 4 - Loans - Major Classifications (Details) - USD (\$) \$ in Thousands	Dec. 31, 2020	Sep. 30, 2020	Jun. 30, 2020	Dec. 31, 2019	Sep. 30, 2019	Jun. 30, 2019
Loans and Leases Receivable, Gross, Total	\$		\$			
	557,434	•	544,473			
Net Deferred loan fees and costs	(177)		(1,612)			
Allowance for loan losses	(5,912)	\$ (5,767)	(5,678)	\$ (4,095)	\$ (3,909)	\$ (3,788)
Net loans	551,345		537,183			,
Commercial Portfolio Segment [Member]						
Loans and Leases Receivable, Gross, Total	158,911		158,667	,		
Allowance for loan losses	(858)	(940)	(947)	(766)	(649)	(660)
Commercial Real Estate Portfolio Segment [Member]						
Allowance for loan losses	(3,817)	(3,694)	(3,623)	(2,652)	(2,645)	(2,575)
Commercial Real Estate Portfolio Segment [Member]						
Construction Loans [Member]						
Loans and Leases Receivable, Gross, Total	3,802		16,235			
Commercial Real Estate Portfolio Segment [Member]   Other	•					
Commercial Real Estate Loans [Member]						
Loans and Leases Receivable, Gross, Total	250,529	)	229,029	)		
Residential Portfolio Segment [Member]						
Allowance for loan losses	(1,028)	(997)	(989)	(615)	(550)	(494)
Residential Portfolio Segment [Member]   Construction						
Loans [Member]						
Loans and Leases Receivable, Gross, Total	6,026		9,344			
Residential Portfolio Segment [Member]   Residential Real						
Estate Owner Occupied Loans [Member]						
Loans and Leases Receivable, Gross, Total	93,726		90,494			
Residential Portfolio Segment [Member]   Residential Real						
Estate Non-owner Occupied Loans [Member]						
Loans and Leases Receivable, Gross, Total	20,210		19,370			
Consumer Portfolio Segment [Member]						
Loans and Leases Receivable, Gross, Total	24,230		21,334			
Allowance for loan losses	\$ (209)	\$ (136)	\$ (119)	\$ (62)	\$ (65)	\$ (59)

Note 4 - Loans - Allowance	3 Months Ended 6 Months Ended					
for Loan Losses by Portfolio Segment (Details) - USD (\$) \$ in Thousands	Dec. 31, 2020	Dec. 31, 2019	Dec. 31, 2020	Dec. 31, 2019	Dec. 31, 2020	Jun. 30, 2020
Beginning balance	\$ 5,767	\$ 3,909	\$ 5,678	\$ 3,788		
Provision for loan losses	130	185	260	315		
Loans charged-off	(12)	(7)	(78)	(23)		
Recoveries	27	8	52	15		
Total ending allowance balance	5,912	4,095	5,912	4,095		
Individually evaluated for impairment	,	,	,	,	\$ 9	\$ 34
Total ending allowance balance	5,767	4,095	5,678	3,788	5,912	5,678
Loans individually evaluated for impairment	,	,	,	,	2,343	1,923
Total ending loans balance					-	544,797
Financial Asset Acquired [Member]					,	,
Collectively evaluated for impairment					185	197
Loans collectively evaluated for impairment					40,640	48,969
Financial Asset Originated [Member]					,	,
Collectively evaluated for impairment					5,718	5,447
Loans collectively evaluated for impairment					· ·	493,905
Commercial Portfolio Segment [Member]					,	,
Beginning balance	940	649	947	660		
Provision for loan losses	(82)	117	(67)	106		
Loans charged-off	(- )		(22)			
Recoveries			( )			
Total ending allowance balance	858	766	858	766		
Individually evaluated for impairment		,		, , ,	3	28
Total ending allowance balance	858	766	947	660	858	947
Loans individually evaluated for impairment		,			148	179
Total ending loans balance						157,328
Commercial Portfolio Segment [Member]   Financial					10,7110	107,020
Asset Acquired [Member]						
Collectively evaluated for impairment						
Loans collectively evaluated for impairment					661	1,095
Commercial Portfolio Segment [Member]   Financial						,
Asset Originated [Member]						
Collectively evaluated for impairment					855	919
Loans collectively evaluated for impairment					158,331	156,054
Commercial Real Estate Portfolio Segment [Member]					•	
Beginning balance	3,694	2,645	3,623	2,575		
Provision for loan losses	122	6	192	75		
Loans charged-off						
Recoveries	1	1	2	2		
Total ending allowance balance	3,817	2,652		2,652		
Individually evaluated for impairment	-	•	-	-		6

	2.017	2.652	2 (22	2.575	2.017	2.622
Total ending allowance balance	3,817	2,652	3,623	2,575	3,817	3,623
Loans individually evaluated for impairment					1,151	1,045
Total ending loans balance					254,464	245,957
Commercial Real Estate Portfolio Segment [Member]   Financial Asset Acquired [Member]						
Collectively evaluated for impairment					101	103
Loans collectively evaluated for impairment					7,824	8,072
Commercial Real Estate Portfolio Segment [Member]					7,024	0,072
Financial Asset Originated [Member]						
Collectively evaluated for impairment					3,716	3,514
Loans collectively evaluated for impairment						236,840
Residential Portfolio Segment [Member]					243,407	230,040
Beginning balance	997	550	989	494		
Provision for loan losses	31	64	39	120		
Loans charged-off	<i>J</i> 1	04	37	120		
Recoveries		1		1		
Total ending allowance balance	1,028	615	1,028	615		
Individually evaluated for impairment	1,020	013	1,020	013	6	
Total ending allowance balance	1,028	615	989	494	1,028	989
Loans individually evaluated for impairment	1,020	013	707	777	1,044	699
Total ending loans balance						120,119
Residential Portfolio Segment [Member]   Financial Asse	t				120,777	120,117
Acquired [Member]	<u>.</u>					
Collectively evaluated for impairment					84	94
Loans collectively evaluated for impairment					23,220	27,252
Residential Portfolio Segment [Member]   Financial Asse	t					_
Originated [Member]	<u> </u>					
Collectively evaluated for impairment					938	895
Loans collectively evaluated for impairment					96,733	92,168
Consumer Portfolio Segment [Member]					,	,
Beginning balance	136	65	119	59		
Provision for loan losses	59	(2)	96	14		
Loans charged-off	(12)	(7)	(56)	(23)		
Recoveries	26	6	50	12		
Total ending allowance balance	209	62	209	62		
Individually evaluated for impairment						
Total ending allowance balance	\$ 209	\$ 62	\$ 119	\$ 59	209	119
Loans individually evaluated for impairment						
Total ending loans balance					24,263	21,393
Consumer Portfolio Segment [Member]   Financial Asset					,	,
Acquired [Member]						
Collectively evaluated for impairment						
Loans collectively evaluated for impairment					8,935	12,550
Consumer Portfolio Segment [Member]   Financial Asset						
Originated [Member]						

Collectively evaluated for impairment
Loans collectively evaluated for impairment

209 119 \$ 15,328 \$ 8,843

Note 4 - Loans - Impaired Loans (Details) - USD (\$) \$ in Thousands				hs Ended , Dec. 31, 2019	Jun. 30, 2020
Impaired Financing Receivable, Related Allowance	\$ 9		\$9		\$ 34
Impaired Financing Receivable, Unpaid Principal Balance	2,684		2,684		2,195
Impaired Financing Receivable, Recorded Investment	2,343		2,343		1,923
Impaired Financing Receivable, Average Recorded Investment	2,211	\$ 887	2,091	\$ 968	
Impaired Financing Receivable, Interest Income, Accrual Method	13	6	25	105	
Impaired Financing Receivable, Interest Income, Cash Basis	13	6	25	105	
Method	13	U	23	103	
Commercial Real Estate Portfolio Segment [Member]   Other Commercial Real Estate Loans [Member]					
Impaired Financing Receivable, with No Related Allowance,	1,286		1,286		922
<u>Unpaid Principal Balance</u>	1,200		1,200		722
Impaired Financing Receivable, with No Related Allowance,	1,151		1,151		836
Recorded Investment	1,131		1,151		050
Impaired Financing Receivable, with No Related Allowance,	907	244	867	303	
Average Recorded Investment					
Impaired Financing Receivable, with No Related Allowance,	3	1	4	87	
Interest Income, Accrual Method					
Impaired Financing Receivable, with No Related Allowance, Interest Income, Cash Basis Method	3	1	4	87	
Impaired Financing Receivable, with Related Allowance, Unpaid					
Principal Balance					209
Impaired Financing Receivable, with Related Allowance,					
Recorded Investment					209
Impaired Financing Receivable, Related Allowance					6
Impaired Financing Receivable, with Related Allowance, Average	204	210	205	210	
Recorded Investment	204	218	205	219	
Impaired Financing Receivable, with Related Allowance, Interest	3	3	6	6	
Income, Accrual Method	3	3	O	6	
Impaired Financing Receivable, with Related Allowance, Interest	3	3	6	6	
Income, Cash Basis Method	3	3	O	O	
Residential Portfolio Segment [Member]   Residential Real Estate					
Owner Occupied Loans [Member]					
Impaired Financing Receivable, with No Related Allowance,	947		947		604
Unpaid Principal Balance					
Impaired Financing Receivable, with No Related Allowance,	797		797		463
Recorded Investment					
Impaired Financing Receivable, with No Related Allowance, Average Recorded Investment	720	10	629	25	
Impaired Financing Receivable, with No Related Allowance,					
Interest Income, Accrual Method	5		11	7	
Impaired Financing Receivable, with No Related Allowance,					
Interest Income, Cash Basis Method	5		11	7	

Impaired Financing Receivable, with Related Allowance, Unpaid Principal Balance	30		30		
Impaired Financing Receivable, with Related Allowance,	2.0		2.0		
Recorded Investment	30		30		
Impaired Financing Receivable, Related Allowance	6		6		
Impaired Financing Receivable, with Related Allowance, Average	10		5		
Recorded Investment	10		3		
Impaired Financing Receivable, with Related Allowance, Interest					
Income, Accrual Method					
Impaired Financing Receivable, with Related Allowance, Interest					
Income, Cash Basis Method					
Residential Portfolio Segment [Member]   Residential Real Estate					
Non-owner Occupied Loans [Member]					
Impaired Financing Receivable, with No Related Allowance,	274		274		284
<u>Unpaid Principal Balance</u>	2/4		2/4		20 <del>4</del>
Impaired Financing Receivable, with No Related Allowance,	217		217		236
Recorded Investment	217		21/		230
Impaired Financing Receivable, with No Related Allowance,	220	250	225	254	
Average Recorded Investment	220	230	223	234	
Impaired Financing Receivable, with No Related Allowance,					
Interest Income, Accrual Method					
Impaired Financing Receivable, with No Related Allowance,					
Interest Income, Cash Basis Method					
Commercial Portfolio Segment [Member]					
Impaired Financing Receivable, with No Related Allowance,				5	
Interest Income, Cash Basis Method				3	
Impaired Financing Receivable, with Related Allowance, Unpaid	147		147		176
Principal Balance	14/		14/		1/0
Impaired Financing Receivable, with Related Allowance,	148		148		179
Recorded Investment	148		148		1/9
Impaired Financing Receivable, Related Allowance	3		3		\$ 28
Impaired Financing Receivable, with Related Allowance, Average	150	165	160	167	
Recorded Investment	150	165	160	167	
Impaired Financing Receivable, with Related Allowance, Interest	2	2	1	\$ 5	
Income, Accrual Method	۷	۷	4	<b>\$</b> 2	
Impaired Financing Receivable, with Related Allowance, Interest	\$ 2	\$ 2	\$ 4		
Income, Cash Basis Method	ΦΔ	φ∠	<b>Φ</b> +		

Note 4 - Loans - Loans Past Due (Details) - USD (\$) \$ in Thousands	Dec. 31, 2020	Jun. 30, 2020
Financing Receivable, Recorded Investment, Nonaccrual Status	\$ 1,940	\$ 1,185
Financing Receivable, Recorded Investment, 90 Days Past Due and Still Accruing		41
Financing Receivable, Recorded Investment, Past Due	1,666	1,011
Financing Receivable, Recorded Investment, Not Past Due	557,198	8 543,786
Financing Receivable, Recorded Investment, Total	558,864	4544,797
Financial Asset, 30 to 59 Days Past Due [Member]		
Financing Receivable, Recorded Investment, Past Due	715	127
Financial Asset, 60 to 89 Days Past Due [Member]		
Financing Receivable, Recorded Investment, Nonaccrual Status		2
Financing Receivable, Recorded Investment, Past Due	53	51
Financial Asset, Equal to or Greater than 90 Days Past Due [Member]		
Financing Receivable, Recorded Investment, Nonaccrual Status	898	792
Financing Receivable, Recorded Investment, Past Due	898	833
Commercial Portfolio Segment [Member]		
Financing Receivable, Recorded Investment, Nonaccrual Status		21
Financing Receivable, Recorded Investment, 90 Days Past Due and Still Accruing		
Financing Receivable, Recorded Investment, Past Due		21
Financing Receivable, Recorded Investment, Not Past Due	159,140	0157,307
Financing Receivable, Recorded Investment, Total	159,140	0157,328
Commercial Portfolio Segment [Member]   Financial Asset, 30 to 59 Days Past Due [Member]		
Financing Receivable, Recorded Investment, Past Due		
Commercial Portfolio Segment [Member]   Financial Asset, 60 to 89 Days Past Due [Member]		
Financing Receivable, Recorded Investment, Past Due		
Commercial Portfolio Segment [Member]   Financial Asset, Equal to or Greater than 90 Days		
Past Due [Member]		
Financing Receivable, Recorded Investment, Past Due		21
Commercial Real Estate Portfolio Segment [Member]		
Financing Receivable, Recorded Investment, Total	254,464	4245,957
Commercial Real Estate Portfolio Segment [Member]   Other Commercial Real Estate Loans		
[Member]		
Financing Receivable, Recorded Investment, Nonaccrual Status	902	785
Financing Receivable, Recorded Investment, 90 Days Past Due and Still Accruing		
Financing Receivable, Recorded Investment, Past Due	629	630
Financing Receivable, Recorded Investment, Not Past Due	250,044	4229,086
Financing Receivable, Recorded Investment, Total	250,673	3 229,716
Commercial Real Estate Portfolio Segment [Member]   Construction Loans [Member]		
Financing Receivable, Recorded Investment, Past Due		
Financing Receivable, Recorded Investment, Not Past Due	3,791	16,241
Financing Receivable, Recorded Investment, Total	3,791	16,241
Commercial Real Estate Portfolio Segment [Member]   Financial Asset, 30 to 59 Days Past Due [Member]   Other Commercial Real Estate Loans [Member]	<u>L</u>	

Financing Receivable, Recorded Investment, Past Due		
Commercial Real Estate Portfolio Segment [Member]   Financial Asset, 30 to 59 Days Past Due	2	
[Member]   Construction Loans [Member]		
Financing Receivable, Recorded Investment, Past Due		
Commercial Real Estate Portfolio Segment [Member]   Financial Asset, 60 to 89 Days Past Due	<u> </u>	
[Member]   Other Commercial Real Estate Loans [Member]		
Financing Receivable, Recorded Investment, Past Due		2
Commercial Real Estate Portfolio Segment [Member]   Financial Asset, 60 to 89 Days Past Due	<u> </u>	
[Member]   Construction Loans [Member]		
Financing Receivable, Recorded Investment, Past Due		
Commercial Real Estate Portfolio Segment [Member]   Financial Asset, Equal to or Greater		
than 90 Days Past Due [Member]   Other Commercial Real Estate Loans [Member]		
Financing Receivable, Recorded Investment, Past Due	629	628
Commercial Real Estate Portfolio Segment [Member]   Financial Asset, Equal to or Greater		
than 90 Days Past Due [Member]   Construction Loans [Member]		
Financing Receivable, Recorded Investment, Past Due		
Residential Portfolio Segment [Member]		
Financing Receivable, Recorded Investment, Total	120,997	7 120,119
Residential Portfolio Segment [Member]   Construction Loans [Member]		
Financing Receivable, Recorded Investment, Past Due	4	
Financing Receivable, Recorded Investment, Not Past Due	6,121	9,435
Financing Receivable, Recorded Investment, Total	6,125	9,435
Residential Portfolio Segment [Member]   Residential Real Estate Owner Occupied Loans		
[Member]		
Financing Receivable, Recorded Investment, Nonaccrual Status	821	143
Financing Receivable, Recorded Investment, 90 Days Past Due and Still Accruing		29
Financing Receivable, Recorded Investment, Past Due	723	172
Financing Receivable, Recorded Investment, Not Past Due	93,926	91,102
Financing Receivable, Recorded Investment, Total	94,649	91,274
Residential Portfolio Segment [Member]   Residential Real Estate Non-owner Occupied Loans		
[Member]		
Financing Receivable, Recorded Investment, Nonaccrual Status	217	236
Financing Receivable, Recorded Investment, 90 Days Past Due and Still Accruing		
Financing Receivable, Recorded Investment, Past Due		
Financing Receivable, Recorded Investment, Not Past Due	20,223	19,410
Financing Receivable, Recorded Investment, Total	20,223	19,410
Residential Portfolio Segment [Member]   Financial Asset, 30 to 59 Days Past Due [Member]	ŕ	
Construction Loans [Member]		
Financing Receivable, Recorded Investment, Past Due	4	
Residential Portfolio Segment [Member]   Financial Asset, 30 to 59 Days Past Due [Member]		
Residential Real Estate Owner Occupied Loans [Member]		
Financing Receivable, Recorded Investment, Past Due	450	
Residential Portfolio Segment [Member]   Financial Asset, 30 to 59 Days Past Due [Member]		
Residential Real Estate Non-owner Occupied Loans [Member]		
Financing Receivable, Recorded Investment, Past Due		

Residential Portfolio Segment [Member]   Financial Asset, 60 to 89 Days Past Due [Member]		
Construction Loans [Member]		
Financing Receivable, Recorded Investment, Past Due		
Residential Portfolio Segment [Member]   Financial Asset, 60 to 89 Days Past Due [Member]		
Residential Real Estate Owner Occupied Loans [Member]		
Financing Receivable, Recorded Investment, Past Due	4	
Residential Portfolio Segment [Member]   Financial Asset, 60 to 89 Days Past Due [Member]   Residential Real Estate Non-owner Occupied Loans [Member]		
Financing Receivable, Recorded Investment, Past Due		
Residential Portfolio Segment [Member]   Financial Asset, Equal to or Greater than 90 Days		
Past Due [Member]   Construction Loans [Member]		
Financing Receivable, Recorded Investment, Past Due		
Residential Portfolio Segment [Member]   Financial Asset, Equal to or Greater than 90 Days		
Past Due [Member]   Residential Real Estate Owner Occupied Loans [Member]		
Financing Receivable, Recorded Investment, Past Due	269	172
Residential Portfolio Segment [Member]   Financial Asset, Equal to or Greater than 90 Days		
Past Due [Member]   Residential Real Estate Non-owner Occupied Loans [Member]		
Financing Receivable, Recorded Investment, Past Due		
Consumer Portfolio Segment [Member]		
Financing Receivable, Recorded Investment, Nonaccrual Status		
Financing Receivable, Recorded Investment, 90 Days Past Due and Still Accruing		12
Financing Receivable, Recorded Investment, Past Due	310	188
Financing Receivable, Recorded Investment, Not Past Due	23,953	21,205
Financing Receivable, Recorded Investment, Total	24,263	21,393
Consumer Portfolio Segment [Member]   Financial Asset, 30 to 59 Days Past Due [Member]		
Financing Receivable, Recorded Investment, Past Due	261	127
Consumer Portfolio Segment [Member]   Financial Asset, 60 to 89 Days Past Due [Member]		
Financing Receivable, Recorded Investment, Past Due	49	49
Consumer Portfolio Segment [Member]   Financial Asset, Equal to or Greater than 90 Days Pas	<u>t</u>	
Due [Member]		
Financing Receivable, Recorded Investment, Past Due		\$ 12

Note 4 - Loans - Recorded Investment by Risk Category (Details) - USD (\$) \$ in Thousands	Dec. 31, 2020	Jun. 30, 2020
Pass [Member]		
Financing Receivable, Net	\$ 426,793	\$ 413,704
Special Mention [Member]		
Financing Receivable, Net	2,559	1,798
Substandard [Member]		
Financing Receivable, Net	5,662	9,914
Doubtful [Member]		
Financing Receivable, Net	1,722	1,042
Not Rated [Member]		
Financing Receivable, Net	122,128	118,339
Commercial Portfolio Segment [Member]   Pass [Member]		
Financing Receivable, Net	158,054	152,911
Commercial Portfolio Segment [Member]   Special Mention [Member]		
Financing Receivable, Net	453	143
Commercial Portfolio Segment [Member]   Substandard [Member]		
Financing Receivable, Net	359	3,979
Commercial Portfolio Segment [Member]   Doubtful [Member]		
Financing Receivable, Net		21
Commercial Portfolio Segment [Member]   Not Rated [Member]		
Financing Receivable, Net	274	274
Commercial Real Estate Portfolio Segment [Member]   Pass [Member]   Construction Loans		
[Member]		
Financing Receivable, Net	3,791	16,241
Commercial Real Estate Portfolio Segment [Member]   Pass [Member]   Other Commercial		
Real Estate Loans [Member]		
Financing Receivable, Net	241,702	220,311
Commercial Real Estate Portfolio Segment [Member]   Special Mention [Member]		
Construction Loans [Member]		
Financing Receivable, Net		
Commercial Real Estate Portfolio Segment [Member]   Special Mention [Member]   Other		
Commercial Real Estate Loans [Member]		
Financing Receivable, Net	1,933	1,469
Commercial Real Estate Portfolio Segment [Member]   Substandard [Member]		
Construction Loans [Member]		
Financing Receivable, Net		
Commercial Real Estate Portfolio Segment [Member]   Substandard [Member]   Other		
Commercial Real Estate Loans [Member]		
Financing Receivable, Net	5,075	5,378

Commercial Real Estate Portfolio Segment [Member]   Doubtful [Member]   Construction Loans [Member]		
Financing Receivable, Net		
Commercial Real Estate Portfolio Segment [Member]   Doubtful [Member]   Other		
Commercial Real Estate Loans [Member]		
Financing Receivable, Net	902	785
Commercial Real Estate Portfolio Segment [Member]   Not Rated [Member]   Construction	> <b>-</b>	, 30
Loans [Member]		
Financing Receivable, Net		
Commercial Real Estate Portfolio Segment [Member]   Not Rated [Member]   Other		
Commercial Real Estate Loans [Member]		
Financing Receivable, Net	1,061	1,773
Residential Portfolio Segment [Member]   Pass [Member]   Construction Loans [Member]	,	,
Financing Receivable, Net	1,306	3,234
Residential Portfolio Segment [Member]   Pass [Member]   Residential Real Estate Owner	1,000	2,22.
Occupied Loans [Member]		
Financing Receivable, Net	1,988	2,419
Residential Portfolio Segment [Member]   Pass [Member]   Residential Real Estate Non-	1,500	_, ,
owner Occupied Loans [Member]		
Financing Receivable, Net	19,265	18,435
Residential Portfolio Segment [Member]   Special Mention [Member]   Construction Loans	.,	-,
[Member]		
Financing Receivable, Net		
Residential Portfolio Segment [Member]   Special Mention [Member]   Residential Real		
Estate Owner Occupied Loans [Member]		
Financing Receivable, Net		
Residential Portfolio Segment [Member]   Special Mention [Member]   Residential Real		
Estate Non-owner Occupied Loans [Member]		
Financing Receivable, Net	173	186
Residential Portfolio Segment [Member]   Substandard [Member]   Construction Loans		
[Member]		
Financing Receivable, Net		
Residential Portfolio Segment [Member]   Substandard [Member]   Residential Real Estate		
Owner Occupied Loans [Member]		
Financing Receivable, Net	18	334
Residential Portfolio Segment [Member]   Substandard [Member]   Residential Real Estate		
Non-owner Occupied Loans [Member]		
Financing Receivable, Net	210	223
Residential Portfolio Segment [Member]   Doubtful [Member]   Construction Loans		
[Member]		
Financing Receivable, Net		
Residential Portfolio Segment [Member]   Doubtful [Member]   Residential Real Estate		
Owner Occupied Loans [Member]		
Financing Receivable, Net	603	
Residential Portfolio Segment [Member]   Doubtful [Member]   Residential Real Estate Non-	=	
owner Occupied Loans [Member]		

Financing Receivable, Net	217	236
Residential Portfolio Segment [Member]   Not Rated [Member]   Construction Loans		
[Member]		
Financing Receivable, Net	4,819	6,201
Residential Portfolio Segment [Member]   Not Rated [Member]   Residential Real Estate		
Owner Occupied Loans [Member]		
Financing Receivable, Net	92,040	88,521
Residential Portfolio Segment [Member]   Not Rated [Member]   Residential Real Estate		
Non-owner Occupied Loans [Member]		
Financing Receivable, Net	358	330
Consumer Portfolio Segment [Member]   Pass [Member]		
Financing Receivable, Net	687	153
Consumer Portfolio Segment [Member]   Special Mention [Member]		
Financing Receivable, Net		
Consumer Portfolio Segment [Member]   Substandard [Member]		
Financing Receivable, Net		
Consumer Portfolio Segment [Member]   Doubtful [Member]		
Financing Receivable, Net		
Consumer Portfolio Segment [Member]   Not Rated [Member]		
Financing Receivable, Net	\$ 23,576	\$ 21,240

<b>Note 5 - Fair Value (Details</b>	3 Months Ended		ails 3 Months Ended 6 Months Ended		hs Ended	
Textual) - USD (\$) \$ in Thousands	Dec. 31, 2020	Dec. 31, 2019	Dec. 31, 2020	Dec. 31, 2019	Jun. 30, 2020	
Impaired Financing Receivable, Related Allowance	\$ 9		\$ 9		\$ 34	
Other Real Estate Owned, Fair Value Disclosure	0		0		0	
Fair Value, Nonrecurring [Member]   Impaired Loans [Member]						
Impaired Financing Receivable, Related Allowance	0		0		\$ 0	
Allowance for Loan and Lease Losses, Period Increase (Decrease), Total	\$ 0	\$ 0	\$ 0	\$ 0		

Note 5 - Fair Value - Assets and Liabilities Measured at Fair Value on a Recurring Basis (Details) - USD (\$) \$ in Thousands	Dec. 31, 2020	Jun. 30, 2020
Securities, available-for-sale  Equity securities	\$ 145,665 400	\$ 143,918
US Treasury Securities [Member] Securities, available-for-sale	501	1,256
US Government-sponsored Enterprises Debt Securities [Member] Securities, available-for-sale	8,109	10,532
US States and Political Subdivisions Debt Securities [Member]  Securities, available-for-sale	67,034	63,492
Residential Mortgage Backed Securities [Member]  Securities, available-for-sale	58,248	50,156
Commercial Mortgage Backed Securities [Member]  Securities, available-for-sale  Call to a line of Marchael	7,078	8,497
Collateralized Mortgage Obligations [Member] Securities, available-for-sale Fair Value, Requiring [Member]	4,695	9,985
Fair Value, Recurring [Member]  Equity securities  Fair Value, Recurring [Member]   Fair Value, Inputs, Level 1 [Member]	400	
Equity securities  Fair Value, Recurring [Member]   Fair Value, Inputs, Level 2 [Member]		
Equity securities  Fair Value, Recurring [Member]   Fair Value, Inputs, Level 3 [Member]	400	
Equity securities Fair Value, Recurring [Member]   US Treasury Securities [Member]		
Securities, available-for-sale  Fair Value, Recurring [Member]   US Treasury Securities [Member]   Fair Value, Inputs,	501	1,256
Level 1 [Member]  Securities, available-for-sale  Fair Value, Recurring [Member]   US Treasury Securities [Member]   Fair Value, Inputs,  Level 2 [Member]  Securities, available-for-sale  Fair Value, Recurring [Member]   US Treasury Securities [Member]   Fair Value, Inputs,	501	1,256
Level 3 [Member]  Securities, available-for-sale  Fair Value, Recurring [Member]   US Government-sponsored Enterprises Debt Securities  [Member]  Securities, available-for-sale	8,109	10,532
Fair Value, Recurring [Member]   US Government-sponsored Enterprises Debt Securities [Member]   Fair Value, Inputs, Level 1 [Member]	2,207	-

Securities, available-for-sale		
Fair Value, Recurring [Member]   US Government-sponsored Enterprises Debt Securities		
[Member]   Fair Value, Inputs, Level 2 [Member]		
Securities, available-for-sale	8,109	10,532
Fair Value, Recurring [Member]   US Government-sponsored Enterprises Debt Securities		
[Member]   Fair Value, Inputs, Level 3 [Member]		
Securities, available-for-sale		
Fair Value, Recurring [Member]   US States and Political Subdivisions Debt Securities		
[Member]		
Securities, available-for-sale	67,034	63,492
Fair Value, Recurring [Member]   US States and Political Subdivisions Debt Securities		
[Member]   Fair Value, Inputs, Level 1 [Member]		
Securities, available-for-sale		
Fair Value, Recurring [Member]   US States and Political Subdivisions Debt Securities		
[Member]   Fair Value, Inputs, Level 2 [Member]		
Securities, available-for-sale	67,034	63,492
Fair Value, Recurring [Member]   US States and Political Subdivisions Debt Securities		
[Member]   Fair Value, Inputs, Level 3 [Member]		
Securities, available-for-sale		
Fair Value, Recurring [Member]   Residential Mortgage Backed Securities [Member]		
Securities, available-for-sale	58,248	50,156
Fair Value, Recurring [Member]   Residential Mortgage Backed Securities [Member]   Fair		
Value, Inputs, Level 1 [Member]		
Securities, available-for-sale		
Fair Value, Recurring [Member]   Residential Mortgage Backed Securities [Member]   Fair		
Value, Inputs, Level 2 [Member]		
Securities, available-for-sale	58,248	50,156
Fair Value, Recurring [Member]   Residential Mortgage Backed Securities [Member]   Fair		
Value, Inputs, Level 3 [Member]		
Securities, available-for-sale		
Fair Value, Recurring [Member]   Commercial Mortgage Backed Securities [Member]		
Securities, available-for-sale	7,078	8,497
Fair Value, Recurring [Member]   Commercial Mortgage Backed Securities [Member]   Fair		
Value, Inputs, Level 1 [Member]		
Securities, available-for-sale		
Fair Value, Recurring [Member]   Commercial Mortgage Backed Securities [Member]   Fair		
Value, Inputs, Level 2 [Member]		
Securities, available-for-sale	7,078	8,497
Fair Value, Recurring [Member]   Commercial Mortgage Backed Securities [Member]   Fair		
Value, Inputs, Level 3 [Member]		
Securities, available-for-sale		
Fair Value, Recurring [Member]   Collateralized Mortgage Obligations [Member]		
Securities, available-for-sale	4,695	
Fair Value, Recurring [Member]   Collateralized Mortgage Obligations [Member]   Fair		
Value, Inputs, Level 1 [Member]		

Securities, available-for-sale		
Fair Value, Recurring [Member]   Collateralized Mortgage Obligations [Member]   Fair Value, Inputs, Level 2 [Member]		
Securities, available-for-sale	4,695	
Fair Value, Recurring [Member]   Collateralized Mortgage Obligations [Member]   Fair		
Value, Inputs, Level 3 [Member]		
Securities, available-for-sale		
Fair Value, Recurring [Member]   Collateralized Debt Obligations [Member]		
Securities, available-for-sale		9,985
Fair Value, Recurring [Member]   Collateralized Debt Obligations [Member]   Fair Value,		
Inputs, Level 1 [Member]		
Securities, available-for-sale		
Fair Value, Recurring [Member]   Collateralized Debt Obligations [Member]   Fair Value,		
Inputs, Level 2 [Member]		
Securities, available-for-sale		9,985
Fair Value, Recurring [Member]   Collateralized Debt Obligations [Member]   Fair Value,		
Inputs, Level 3 [Member]		

Securities, available-for-sale

Note 5 - Fair Value - Estimated Fair Values of Financial Instruments (Details) - USD (\$) \$ in Thousands	Dec. 31, 2020	Jun. 30, 2020
Held-to-maturity, fair value	\$ 3,605	\$ 3,868
Fair Value, Inputs, Level 1 [Member]   Reported Value Measurement [Member]		
Cash and cash equivalents	9,045	9,659
Fair Value, Inputs, Level 1 [Member]   Estimate of Fair Value Measurement [Member]		
Cash and cash equivalents	9,045	9,659
Fair Value, Inputs, Level 2 [Member]   Reported Value Measurement [Member]		
Certificates of deposit in other financial institutions	8,599	11,635
Loans held for sale	4,770	3,507
Accrued interest receivable	2,327	2,646
Short-term borrowings	13,275	6,943
Federal Home Loan Bank advances	18,083	31,161
Accrued interest payable	88	107
Fair Value, Inputs, Level 2 [Member]   Reported Value Measurement [Member]   Demand		
and Savings Deposits [Member]		
Demand and savings deposits	540,811	517,973
Fair Value, Inputs, Level 2 [Member]   Reported Value Measurement [Member]   Time		
Deposits [Member]		
Demand and savings deposits	107,551	115,382
Fair Value, Inputs, Level 2 [Member]   Estimate of Fair Value Measurement [Member]		
Certificates of deposit in other financial institutions	8,819	11,889
Loans held for sale	4,907	3,566
Accrued interest receivable	2,327	2,646
Short-term borrowings	13,275	6,943
Federal Home Loan Bank advances	18,423	31,571
Accrued interest payable	88	107
Fair Value, Inputs, Level 2 [Member]   Estimate of Fair Value Measurement [Member]		
Demand and Savings Deposits [Member]		
Demand and savings deposits	540,811	517,973
Fair Value, Inputs, Level 2 [Member]   Estimate of Fair Value Measurement [Member]		
<u>Time Deposits [Member]</u>		
Demand and savings deposits	108,158	116,238
Fair Value, Inputs, Level 3 [Member]   Reported Value Measurement [Member]		
Held-to-maturity, fair value	3,336	3,541
Loans, net	551,345	537,183
Fair Value, Inputs, Level 3 [Member]   Estimate of Fair Value Measurement [Member]		
Held-to-maturity, fair value	3,605	3,868
<u>Loans, net</u>	\$ 565,397	7 \$ 548,247

Note 6 - Earnings Per Share (Details Textual) - shares	3 Months Ended	6 Months Ended	
	Dec. 31, 2019	Dec. 31, 2020	Dec. 31, 2019
Restricted Stock [Member]			
Antidilutive Securities Excluded from Computation of Earnings Per Share, Amount (in shares)	1,645	569	2,204

Note 6 - Earnings Per Share - Schedule of Earnings Per	3 Mont	hs Ended	6 Months Ended	
Schedule of Earlings Fer Share, Basic and Diluted (Details) - USD (\$) \$ / shares in Units, \$ in Thousands	Dec. 31, 2020	Dec. 31, 2019	Dec. 31, 2020	Dec. 31, 2019
Net income available to common shareholders	\$ 2,507	\$ 1,440	\$ 4,908	\$ 2,943
Weighted average common shares outstanding (in shares)	3,017,268	2,737,800	3,015,741	2,737,755
Basic income per share (in dollars per share)	\$ 0.83	\$ 0.53	\$ 1.63	\$ 1.07
Dilutive effect of restricted stock (in shares)	21			
Total common shares and dilutive potential common shares (in shares)	3,017,289	2,737,800	3,015,741	2,737,755
Dilutive income per share (in dollars per share)	\$ 0.83	\$ 0.53	\$ 1.63	\$ 1.07

		3 Mont	Months Ended		6 Months Ended	
<b>Comprehensive Income</b>						
(Loss) - Components of						
Other Comprehensive		Dec. 31,	Dec. 31,	Dec. 31,	Dec. 31,	
Income (Loss) (Details) -		2020	2019	2020	2019	
USD (\$)						
\$ in Thousands						
Accumulated other comprehensive income (loss), before tax		\$ 5,463	\$ 2,694	\$ 5,393	\$ 1,982	
Accumulated other comprehensive income (loss), tax		(1,147)	(566)	(1,133)	(416)	
Accumulated other comprehensive income (loss), net		4,316	2,128	4,260	1,566	
Unrealized holding loss on available-for-sale securities arising		278	(28)	348	790	
during the period, before tax		270	(20)	540	170	
Unrealized holding loss on available-for-sale securities arising		(59)	5	(73)	(166)	
during the period, tax		(3))	5	(13)	(100)	
Unrealized holding loss on available-for-sale securities arising		219	(23)	275	624	
during the period, net			(=0)	_,,	σ <b>-</b> .	
Amounts reclassified from accumulated other comprehensive	[1],[2	$^{[2]}(8)$	(4)	(8)	(110)	
income, before tax		(0)	(.)	(0)	(110)	
Amounts reclassified from accumulated other comprehensive	[1],[2	2]2	1	2	22	
income, tax						
Amounts reclassified from accumulated other comprehensive	[1],[2	$^{[2]}(6)$	(3)	(6)	(88)	
income, net		250		240		
Net current period other comprehensive loss, before tax		270	(32)	340	680	
Net current period other comprehensive loss, tax		(57)	6	(71)	(144)	
Other comprehensive income (loss)		213	(26)	269	536	
Accumulated other comprehensive income (loss), before tax		5,733	2,662	5,733	2,662	
Accumulated other comprehensive income (loss), tax		(1,204)	(560)	(1,204)	(560)	
Accumulated other comprehensive income (loss), net		\$ 4,529	\$ 2,102	\$ 4,529	\$ 2,102	

<sup>[1]</sup> Income tax expense

<sup>[2]</sup> Securities (gains) losses, net